Applications of Lime and Enzymes for Oil Sands Tailings Management: Dewatering and Mitigation of Methane Emissions

by

Nesma Mohamed Elhosseny Eltoukhy Allam

A thesis submitted in partial fulfillment of the requirements for the degree of

Doctor of Philosophy in Environmental Engineering

Department of Civil and Environmental Engineering University of Alberta

© Nesma Mohamed Elhosseny Eltoukhy Allam, 2023

ABSTRACT

The remediation of oil sands tailings is a crucial environmental aspect of the Canadian oil sands industry. Oil sands tailings management involves the endless storage of massive volumes of fluid fine tailings (FFT). The slow gravity settling of tailings, the release of toxic compounds, and the release of greenhouse gases (GHGs) emissions are all synergistic reasons for the comprehensive tailings environmental concerns and challenges. These mandates finding a cost-effective tailings reclamation strategy. This thesis focused on the engineered non-natural novel strategies to improve the consolidation of tailings and the inhibition of GHGs emissions. This thesis was divided into three research components.

First, the effect of lime treatment on FFT dewatering and improving cap water quality under the simulated End pit lake (EPL) scenario was investigated. EPLs are being examined as one potential approach to reduce FFT inventories for the oil sands industry. Experiments were conducted with various doses of lime (650 to 4000 ppm). The results illustrated that a high lime dose of 3500 ppm achieved the highest FFT water recovery% (WR), decreased the cap water alkalinity after 90 days, and increased the possibility of cation exchange (at day 0). In contrast, the degradation of petroleum hydrocarbons was slightly enhanced at a low lime dose of 650 ppm. In addition, the 650 ppm dose resulted in minimal change in the microbial cell counts at day 90, compared with high lime doses that significantly reduced the cell counts. FFT pore water exhibited higher toxic effects for lime dosages >1600 ppm. Nevertheless, at all lime doses, low cap water toxicity (i.e., <1.0 Toxicity Unit) at day 90 was attained. The low water

toxicity for the cap water can be ascribed to the reduction of cap water pH over time due to the dissolution of atmospheric carbon dioxide into cap water.

Second, the feasibility of enzymatic treatment (cellulase, protease, and lysozyme) was investigated for the first time to accelerate the dewatering of FFT. The findings illustrated that lysozyme (0.5% and 1%) significantly improved FFT dewatering by increasing the WR up to 20% compared with the other enzymes (up to 12%) or the control (2%). Moreover, lysozyme treatment resulted in the highest increase in ionic strength (0.038 to 0.1 mol/L), decrease in diffuse double layer (DDL) thickness $(1.54 \times 10^{-7} \text{ to } 9.40 \times 10^{-8} \text{ cm})$, and increase in zeta potential (-34.7 to -14.8 mV). Increased methane production was observed for cellulase, lysozyme, and protease (0.5%). The enhanced dewatering could be linked to the ebullition of methane gas resulting from the methanogenic activity, which created pathways for the trapped water release. In addition, the dissolution of carbonate minerals during the release of methane gas increased ionic strength and decreased the DDL of the FFT. Lysozyme 1% treatment was also the most effective in reducing naphthenic acid fractions (1934.6 to 243 ng/mL); however, the released water had high toxicity toward Vibrio fischeri and had a slight decrease in microbial populations.

The final investigation focused on the engineering strategies to reduce GHGs emissions through methanogenesis inhibition in tailings (methane inhibition) by chemical treatment (lime) and biological treatment using enzymes (lysozyme and protease). Overall, treatment with protease 3%, lysozyme 3%, and lime 5000 ppm inhibited CH₄ production (by 52%, 28%, and 25%, respectively) and were weakly associated with the archaeal abundance. Enzyme treatment resulted in a higher

reduction in CH₄ production compared with lime treatment. A 3% protease suppressed CH₄ production throughout the experiment (the change in methane was 0.78 mM), which could be attributed to the pH reduction to pH 4.9 at week 23 resulting from the formation of volatile fatty acids. The toxicity effect was greater with protease 3% and lysozyme 3% treatment than with lime treatment. Lime treatment resulted in the highest reduction in 16S rRNA gene copies.

In summary, the significant implications for the use of lime to improve water quality in EPLs and benefit the long-term success of FFT remediation within EPLs were highlighted in the first part of the thesis. The second and third parts of the study introduced a novel technique based on the enzymatic treatment that was applied for the first time in oil sand tailings. These parts provide fundamental insights into the dewatering and inhibition of GHGs emissions by biological treatment using enzymes.

PREFACE

All research completed on this thesis is an original work in which I, Nesma Allam, proposed, designed, and conducted all the sets experiments as well as the interpretation, the analysis of the data and the preparation of the manuscripts, under full revision and the supervision of Dr. Ania Ulrich and Dr. Bipro Dhar. Some researchers and lab mates also contributed to chemical preparation, or sample analysis, manuscript edits, and some of them were co-authors of the manuscripts submitted for publication. Other analyses were done in other departments of the University of Alberta, as specified below.

Chapter 3: A version of this chapter has been published as Allam, N.E.; Romaniuk, N.; Tate, M.; Meshref, M.N.A.; Dhar, B.R.; Ulrich, A. (2021): "Impact of lime treatment on tailings dewatering and cap water quality under an oil sands end pit lake scenario", Science of the Total Environment Journal, 781, 146699. I conducted all the experimental plan, experimental set-up, sampling, data collection, writing the manuscript and analysis including anions, alkalinity, DOC, toxicity, DNA extraction and qPCR analysis at Dr. Ulrich's research lab (Environmental Engineering at University of Alberta). Mr. Nikolas Romaniuk and Mr. Mike Tate, all from Graymont Western Canada provided the tailings samples and contributed with ideas for the experimental set-up. Dr. Mohamed Meshref in Dr. Dhar's research group contributed to the statistical analysis and manuscript edits. Dr. Ania Ulrich and Dr. Bipro Dhar were responsible for the review, editing, and supervising this work.

Chapter 4: A version of this chapter has been published as "Allam, N.E.; Anwar, M. N.; Kuznetsov, P.V.; Ulrich, A. C.; Dhar, B. R. (2022): "Enzyme-assisted dewatering of oil sands tailings: Significance of water chemistry and biological activity" Chemical Engineering Journal, 437, 135162. I was responsible for all the experimental plan, experimental set-up, data collection, analysis and writing the manuscript. Dr. Mian Nabeel Anwar in Dr. Ulrich's research group contributed to microbial community analysis and manuscript edits, while the data processing was done by me. Dr. Petr Kuznetsov in Dr. Ulrich's research group contributed to the experimental procedure and the manuscript edits. Dr. Ania Ulrich and Dr. Bipro Dhar were responsible for the review, editing, and supervising this work.

Chapter 5: A version of this chapter has been published as: Allam, N.E.; Zakaria, B.S.; Kuznetsov, P.V.; Dhar, B. R., & Ulrich, A. C. (2023): "Mitigating Methane Emission from Oil Sands Tailings Using Enzymatic and Lime Treatments" Chemosphere, 313, 137455. I was responsible for all the experimental set-up, sampling, analysis and writing the manuscript. Dr. Basem Zakaria in Dr. Dhar's research group contributed to microbial community analysis, and he helped in the data processing. Dr. Petr Kuznetsov in Dr. Ulrich's research group contributed in the help of the setup of the experiments and manuscript edits. Dr. Ania Ulrich and Dr. Bipro Dhar were responsible for the review, editing, and supervising this work.

DEDICATION

This work is dedicated:

To my beloved parents who have been my source of inspiration and gave me strength, support and encouragement. Words can never tell how much I am grateful to you for your care, love and support.

To my beloved husband Mohamed who I am truly blessed to have him in my life. To my superheroes Ahmed and Salma who surrounded me with all love and care during my PhD journey. You are an endless joy, support, and inspiration to me.

To my lovely sister Noha and my dear brother Ahmed who are far away but their encouragement, love, endless support and prayers are always with me. I am so grateful to have you all in my life.

ACKNOWLEDGEMENTS

In the name of Allah, the most Gracious, the most Merciful. First, I would like to thank Allah Almighty for guiding me to my accomplishments.

Then first and foremost, I would like to extend my most gratitude to my supervisors Dr. Ania Ulrich and Dr. Bipro Dhar for their intellectual support, encouragement and their motivating guidance during my PhD. I benefited from their depth, their breadth of knowledge and their professional manner during our supervisor-student relationship.

Special thanks and appreciation go to my supervisory committee member Dr. Yang Liu for her scholarly contributions, thoughtful insights and valuable feedback.

I would also like to thank my examination committee members for their valuable comments during the candidacy and final oral examination.

I would like to thank Dr. Petr Kuznetsov and Dr. Main Anwar in Dr. Ulrich's research group for their sincere efforts and support. My appreciation also goes to Dr. Mohamed Meshref, Dr. Long Lin, and Dr. Basem Reda. in Dr. Dhar's research group.

My sincere gratitude also goes to my respected professor Dr. Ian Buchanan to be honored and have him as a candidacy exam member and to assist him in teaching few environmental engineering undergraduate courses which I was benefited from his considerate and professional manner. I highly value the great support and experience gained from my remarkable professors Dr. Selma E. Guigard and Dr. Mohamed Gamal El-Din. My appreciation also goes to Dr. Pamela Chelme-Ayala and the graduate advisors in the Civil and Environmental Department particularly Mrs. Trina Cattral and Mrs. Arlene Figley. I also thank the technicians in Civil and Environmental Department: Mrs. Chen Liang, and Mr. Yupeng Zhao. I would like to extend my appreciation to my friends and colleagues, and staff in the Department of Civil and Environmental Engineering and former and current group members for all their great support during my Ph.D. journey. I would especially like to thank my co-workers in Dr. Ulrich and Dr. Dhar's labs for their great support: Dr. Stanley Poon, Dr. Sarah Miles, Dr. Xiaoxuan Yu, Dr. Evelyn Asiedu, Dr. Korris Lee, Dr. Chelsea Benally, Amy-lynne Balaberda, Heidi Cossey, Kai Wei, Anya Batycky, Erika Strytveen and others. All appreciation goes to Mr. Nikolas Romaniuk and Mr. Mike Tate, from Graymont Western Canada for their valuable feedbacks and advice.

I am very grateful for the financial support provided by the research grants from Natural Sciences and Engineering Research Council of Canada (NSERC) and to the substantial support and generous financial fund for my doctoral research projects received by various scholarships, awards and prizes. I would like to acknowledge: Alexander Graham Bell Canada Graduate Scholarship-Doctoral (CGS D), Queen Elizabeth Scholarship, President's Doctoral Prize of Distinction, Dr. Donald R Stanley Graduate Scholarship in Environmental (Civil) Engineering, Petro-Canada Graduate Scholarship, Alberta Graduate Excellence Scholarship, The Jacob H Masliyah Graduate Award in Oil Sands Engineering and University of Alberta Doctoral Recruitment Scholarship.

The continuous and unlimited support of my friend Sarah Nouh is highly appreciated. Finally, I express my deep gratitude to my family for their patience, support, endless encouragement and love. Thanks for making me see this adventure through the end.

TABLE OF CONTENTS

1 GENERAL INTRODUCTION AND RESEARCH OBJECTIVES	1
1.1 Background and Motivation	
1.1.1 Oil sands tailings management	1
1.2 Problem Statement	
1.3 Research Scope and Objectives	
1.4 Thesis Organization	6
1.5 References	
2 LITERATURE REVIEW	12
2.1 Water and tailings management in oil sands operation	
2.1.1 FFT characteristics	
2.1.2 OSPW characteristics	
2.1.3 Inorganic compounds	
2.1.4 Organic compounds	
2.2 Toxicity of OSPW/FFT	
2.3 MFT remediation/dewatering technologies	
2.3.1 Physicochemical, natural, and mechanical technologies	
2.3.2 Biological processes	
2.3.3 Application of enzymes for dewatering	
2.3.4 End pit lakes	
2.4 Tailings Emission	
2.4.1 Greenhouse Gases	
2.5 Key research gaps	
2.6 References	
3 IMPACT OF LIME TREATMENT ON TAILINGS DEWATERIN	G AND
CAP WATER QUALITY UNDER AN OIL SANDS END PIT LAKE	1
SCENARIO	
3.1 Introduction	
3.2 Materials and methods	
3.2.1 Materials	
3.2.2 Experiment	

3.2.3 Chemical analysis	
3.2.4 Toxicity Bioassay	
3.2.5 DNA Extraction and qPCR Analysis	
3.3. Results and discussion	
3.3.1 Influence of lime treatment on FFT dewatering	
3.3.2 Variations in water chemistry of FFT pore water and cap water .	
3.3.2.1 Alkalinity	
3.3.2.2 Cations	
3.3.3 Variations in bacterial cell counts	
3.3.4 Impact of lime treatment on organic fractions	
3.3.5 Toxicity and implications of pH influence	
3.4 Conclusions	
3.5 References	
4 ENZYME-ASSISTED DEWATERING OF OIL SANDS TAIL	INGS:
SIGNIFICANCE OF WATER CHEMISTRY AND BIOLOGICA	AL ACTIVITY
	118
4.1 Introduction	
4.1 Introduction4.2 Materials and methods	
4.1 Introduction4.2 Materials and methods4.2.1 Materials	
 4.1 Introduction 4.2 Materials and methods 4.2.1 Materials 4.2.2 Experiment 	
 4.1 Introduction 4.2 Materials and methods 4.2.1 Materials 4.2.2 Experiment 4.2.3 Analysis and calculations 	
 4.1 Introduction	
 4.1 Introduction 4.2 Materials and methods 4.2.1 Materials 4.2.2 Experiment 4.2.3 Analysis and calculations 4.2.3.1 Water recovery and solids content 4.2.3.2 Chemical analysis 4.2.3.3 Ionic strength (I) and Diffuse Double Layer (DDL) 4.2.3.4 Zeta potential 4.2.3.5 Toxicity Bioassay 	
 4.1 Introduction 4.2 Materials and methods 4.2.1 Materials 4.2.2 Experiment 4.2.3 Analysis and calculations 4.2.3.1 Water recovery and solids content 4.2.3.2 Chemical analysis 4.2.3.3 Ionic strength (I) and Diffuse Double Layer (DDL) 4.2.3.4 Zeta potential 4.2.3.5 Toxicity Bioassay 4.2.3.6 Microbial community and diversity 	
 4.1 Introduction	
 4.1 Introduction	118 122 122 122 123 124 124 125 126 127 127 129 129
 4.1 Introduction	118 122 122 122 123 124 124 125 126 127 127 127 129 133
 4.1 Introduction 4.2 Materials and methods 4.2.1 Materials 4.2.2 Experiment 4.2.3 Analysis and calculations 4.2.3.1 Water recovery and solids content 4.2.3.2 Chemical analysis 4.2.3.3 Ionic strength (I) and Diffuse Double Layer (DDL) 4.2.3.4 Zeta potential 4.2.3.5 Toxicity Bioassay 4.2.3.6 Microbial community and diversity 4.3 Results and discussion 4.3.1 Effects of enzymes on FFT dewatering 4.3.2 Effects on biogenic methane production 4.3.3 Variations in water chemistry 	118 122 122 122 123 124 124 125 126 127 129 133 135
 4.1 Introduction	118 122 122 122 123 124 124 125 126 127 127 127 127 127 127 127 127 127 127 127 125 125 126 127 127 127 127 127 127 127 127 127 127 127 127 127 129 133 135 135

4.3.4 Naphthenic acid fractions	
4.3.5 Toxicity	
4.3.6 Microbial communities	
4.3.7 Dewatering mechanisms	
4.4 Conclusions	
4.5 References	
5 SIGNIFICANCE OF ENZYME AND LIME TREATMENT IN T	HE
MITIGATION OF METHANE EMISSIONS FROM OIL SANDS 7	FAILINGS
	177
5.1 Introduction	
5.2 Materials and methods	
5.2.1 Materials	
5.2.2 Experiment	
5.2.3 Analysis	
5.2.3.1 Chemical analysis	
5.2.3.2 Toxicity bioassay	
5.2.3.3 Microbial community and diversity	
5.3 Results and discussion	
5.3.1 Impact of enzymes and lime on biogenic methane production	
5.3.2 Organics and hydrocarbon degradation	
5.3.3 The effect of different treatment conditions on toxicity	
5.3.4 Microbial quantity	
5.3.4.1 Microbial community	
5.3.5 Principal Component Analysis (PCA)	
5.4 Conclusions	
5.5 References	
6 CONCLUSIONS AND RECOMMENDATIONS	223
6.1 Conclusions and significance of results	223
6.2 Recommendations	225
BIBLIOGRAPHY	228
APPENDIX A. STANDARD CURVES	
AFFENDIX B. SUPPLEMENIAKY INFORMATION	Z'/8

LIST OF TABLES

TABLE 2.1. APPLICATION OF VARIOUS ENZYMES FOR DEWATERING OF MUNICIPAL, PULP
AND PAPER MILL SLUDGE
TABLE 2.2. SUMMARY OF PREVIOUS STUDIES THAT USED LIME FOR OIL SANDS TAILINGS
PROCESSING AND TREATMENT
TABLE 3.1. THE FFT PORE WATER RECOVERY $\%$ (WR%) at the END of the Experiment
(day 90) as well as FFT solids content at day 90. The solids content for
All columns at day 0 was 50%. All columns consisted of 500 ml lime
TREATED OR UNTREATED FFT COVERED WITH 500 ML RAW SYNTHETIC OSPW. ALL
columns were open to the atmosphere. The untreated FFT represents a 0
PPM TREATMENT AND ACTS AS THE CONTROL EXPERIMENT. RESULTS ILLUSTRATED
Are an average (N = 2 for 0 PPM and N = 3 for 650, 1600, and 3500 PPM) \pm one
STANDARD DEVIATION
TABLE 3.2. Concentrations of cations Ca^{2+} and Na^+ for treated and untreated
FFT pore water and cap water at day 0 and the end of the experiment (day
90) measured by ICP-OES. Results illustrated are an average (n = 2 for 0
PPM and $n = 3$ for 650, 1600, and 3500 PPM) \pm one standard deviation. All
columns consisted of 500 ml treated or untreated FFT covered with 500
ML RAW SYNTHETIC OSPW. ALL THE COLUMNS WERE OPEN TO THE ATMOSPHERE.
The untreated FFT represents a 0 ppm treatment and acts as the control
EXPERIMENT
TABLE 3.3. TOXICITY OF FFT PORE WATER PRESENTED IN TOXICITY UNITS (TU) AT DAY
0 and the end of the experiment (day 90) and cap water toxicity at day 14
and the end of the experiment (day 90) assessed using Microtox. TU ${<}1.0$
INDICATES COMPLETE DETOXIFICATION (SCOTT ET AL., 2008). ALL COLUMNS
consisted of 500 ml treated or untreated FFT covered with 500 ml Raw
SYNTHETIC OSPW. ALL THE COLUMNS WERE OPEN TO THE ATMOSPHERE EXCEPT
The 4000 ppm treatment, which was covered with parafilm. The untreated

FFT REPRESENTS A 0 PPM TREATMENT AND ACTS AS THE CONTROL EXPERIMENT. 103

- TABLE 4.1. THE BACTERIAL DIVERSITY INDEX FOR THE CONTROL CONDITIONS AT WEEK 0AND WEEK 12 FOR ALL ENZYMES TREATED CONDITIONS. THE UNTREATED FFTREPRESENTS A 0 PPM TREATMENT AND ACTS AS THE CONTROL EXPERIMENT.RESULTS ILLUSTRATED ARE AN AVERAGE (N = 3 FOR ALL TREATMENT CONDITIONS) \pm ONE STANDARD DEVIATION.156TABLE 4.2. THE ARCHAEAL DIVERSITY INDEX FOR THE CONTROL CONDITIONS AT WEEK 0AND WEEK 12 FOR ALL ENZYME-TREATED CONDITIONS. THE UNTREATED FFTREPRESENTS A 0 PPM TREATMENT AND ACTS AS THE CONTROL EXPERIMENT.RESULTS ILLUSTRATED ARE AN AVERAGE (N = 3 FOR ALL TREATMENT CONDITIONS) \pm ONE STANDARD DEVIATION.157

LIST OF FIGURES

FIGURE 2.1. SCHEMATIC DIAGRAM FOR THE OIL SANDS EXTRACTION PROCESS AND
TRADITIONAL TAILINGS MANAGEMENT ADAPTED FROM (BEIER AND SEGO 2008)
FIGURE 2.2. A SCHEMATIC DIAGRAM FOR THE STERN MODELS (LEFT) OF AN INTERACTION
BETWEEN NEGATIVELY CHARGED CLAY SURFACE AND ANIONS AND CATIONS IN THE
BULK SOLUTION AS WELL AS DIFFUSE DOUBLE LAYER SURROUNDING A CLAY
PARTICLE (RIGHT). ADAPTED FROM (BRANDON 2016; HUNTER 2001; LIU ET AL.
2013)
FIGURE 2.3. THE PROPOSED PATHWAY OF THE ANAEROBIC HYDROCARBON
DEGRADATION IN OIL SANDS TAILINGS (ADAPTED FROM SIDDIQUE ET AL. (2011)).
FIGURE 2.4. EPL CONTENT AND PERFORMANCE OVER TIME STARTING FROM THE EARLY
STAGE (ON THE LEFT), INTERMEDIATE STAGE AND FUTURE STAGE (ON THE RIGHT).
A DARTED FROM (WESTCOTT AND WATSON 2005) 22

- Figure 3.2. (a) FFT Pore-water PH at day 0 and at the end of the experiment (day 90). (b) FFT PH at day 0 and at the end of the experiment (day 90). All the columns consisted of 500 ml of treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere except the 4000 ppm treatment, which has two conditions: uncovered and covered with parafilm. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one

- FIGURE 3.6. VARIATION OF CATION (CA^{2+} and Na^+) concentrations in CAP water at day 0 and at the end of the experiment (day 90) measured by ICP-OES. Illustrated results are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 mL treated or untreated FFT covered with 500 mL Raw synthetic OSPW (Raw SPW). All the columns were open to the atmosphere. The untreated

FFT REPRESENTS A 0 PPM TREATMENT AND ACTS AS THE CONTROL EXPERIMENT.

- Figure 3.7. Variation of cation K^+ concentrations in CAP water at day 0 and the end of the experiment (day 90) measured by ICP-OES. Illustrated results are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW (Raw SPW). All the columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.95
- Figure 3.8. Bacterial cell counts for treated and untreated FFT at day 0 and at the end of the experiment (day 90) measured by qPCR of the rpoB gene. Results illustrated an average (n = 2 for control and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere except the 4000 ppm which has two conditions: uncovered and covered with parafilm. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.
- - CONTROL, PC 1 (99.9%) AND PC 2 (0.1%), (B) BIPLOT AND LOADING PLOT OF PC1

FIGURE 4.4. CHANGE IN (A) HCO₃⁻, (B) MG²⁺, (C) CA²⁺, CONCENTRATIONS AND (D) PH THROUGHOUT THE EXPERIMENT FOR ALL ENZYME-TREATED FFT COMPARED WITH THE CONTROL/ UNTREATED FFT. ALL COLUMNS CONSISTED OF 800 ML ENZYME-TREATED OR UNTREATED FFT. ALL COLUMNS WERE SEALED WITH PARAFILM AND

- FIGURE 5.1. PHOTOGRAPHS OF EXPERIMENTAL SETUP (A) 158 ML MICROCOSM CONSISTS OF 65 ML FFT AND 35 ML OF ANAEROBIC MEDIA CONTROL, TREATED WITH ENZYME, LIME OR BES, (B) THE MICROCOSMS INCUBATED IN DARK CONDITIONS AT 30 °C.183
- FIGURE 5.2. TEMPORAL CHANGES IN METHANE PRODUCTION THROUGHOUT THE EXPERIMENT FOR ALL TREATMENT CONDITIONS, CONTROL/UNTREATED FFT, AND

- FIGURE 5.7. THE ALPHA DIVERSITY METRICS OF THE MICROBIAL COMMUNITY COMPUTED USING QIIME 2 FOR ALL TREATMENT CONDITIONS, THE CONTROL/UNTREATED FFT,

FIGURE A1. METHANE DETERMINATION CALIBRATION CURVE USING STANDARDS....275 FIGURE A2. TOLUENE DETERMINATION CALIBRATION CURVE USING STANDARDS.....276 FIGURE A3. HEXANE DETERMINATION CALIBRATION CURVE USING STANDARDS......277

NOMENCLATURE

OSPW	Oil sands process-affected water
FFT	Fluid fine tailings
MFT	Mature fine tailings
EPL	End pit lake
BML	Base Mine Lake
GHGs	Greenhouse gases
PHCs	Petroleum hydrocarbons
COPCs	Constituents of potential concern
Ca[OH] ₂	Hydrated lime
Na^+	Sodium
Ca ²⁺	Calcium
HCO_3^-	Bicarbonate
CH ₄	Methane
TU	Toxicity unit
DDL	Diffuse Double Layer
Ι	Ionic strength
NAs	Naphthenic acids
NAFCs	Naphthenic acid fractions compounds
СТ	Consolidated tailings
AEF	Acid-extractable organic fractions
O ₂ -NAs	Classical naphthenic acids
O _x -NAs	Sum of classical NAs and oxidized NAs
PCA	Principal component analysis
DOC	Dissolved organic carbon
sCOD	Soluble chemical oxygen demand
DLVO	Derjaguin-Landau-Verwey-Overbeek

1 GENERAL INTRODUCTION AND RESEARCH OBJECTIVES

1.1 Background and Motivation

1.1.1 Oil sands tailings management

Alberta's oil sands are considered the third largest oil deposits in the world and contribute to Canada's economic growth. However, mitigating the adverse environmental impact of oil sands operations has been challenging. In surface mining operations and during the bitumen extraction using the Clark hot water extraction process, large volumes of fluid fine tailings (FFT) and oil sands processaffected water (OSPW) are continuously generated. To comply with the zerodischarge practice and the provincial environmental legislation, FFT and OSPW need to be stored in engineering ponds called tailings ponds (Allen 2008).

Oil sands tailings management and treatment warrant further research to deal with such endless storage of massive volumes of FFT. Due to the poor settleability of tailings, other non-natural strategies are warranted to solve the issue of the suspended particle and to accelerate the settling successfully. Oil sands operators have used several techniques such as composite tailings, freeze and thaw, centrifugation and atmospheric drying; however, these applications have been hindered due to multiple drawbacks. Thus, more economical and effective solutions are being investigated. Of these potential strategies, demonstrated end pit lakes (EPLs) have been hypothesized and emerged as one of the options to reduce FFT inventories within tailings ponds. EPLs approach, previously introduced in 2012 by Syncrude, is watercapped tailings technologies in which FFT is incorporated into oil sands mine closure landscapes. These potential attempts aim to reduce FFT inventories in tailings ponds for subsequent reclamation (Kabwe et al. 2018; Zubot 2010). EPL is created in old mined out pits by pumping fresh water and OSPW to cap the FFT. This water cap will facilitate FFT integration into mine closure landscapes (Dompierre and Barbour 2016). In Base Mine Lake (BML-the first field-scale demonstration of pit lakes by Syncrude), FFT was settled naturally without any treatment through gravity settling (i.e., self-weight consolidation) (Dompierre 2016).

The environmental development of EPLs encounters many key concerns and potential challenges (COSIA 2012). One of these concerns is the released pore water contributes OSPW- derived constituents to the EPL cap water (i.e., chemical flux of constituents of potential concern (COPCs)) (Dompierre 2016) during the slow settling of FFT. The presence of COPCs possesses high toxicity to the aquatic microorganisms (Dompierre 2016; McQueen et al. 2017; Samadi 2019; White and Liber 2018).

The release of greenhouse gases (GHGs) from tailings ponds is another key environmental concern facing the oil sands industries. The gaseous emissions from Mildred Lake Settling Basin (MLSB) are currently uncontrolled and are input into atmospheric GHGs (Siddique and Kuznetsova 2020). In 2000, the estimated methane production from MLSB was up to 43 million L/day (Holowenko et al. 2000). The release of CH₄ from tailings ponds is over 21 times that of CO₂ (Small et al. 2015). The methanogenic activity can be attributed to the organic components in tailings ponds such as residual bitumen, organic diluents, unrecovered naphtha during the extraction process and citrate. The organic components are metabolized by the indigenous microorganisms in tailings ponds and then converted into GHGs under methanogenic conditions (Siddique and Kuznetsova 2020; Small et al. 2015; Small et al. 2012). The solvents usage during oil sands operation (i.e., bitumen extraction) contributes to the organic components such as naphtha (a mixture of *n*alkanes and monoaromatics BTEX) and paraffinic solvents (Siddique and Kuznetsova 2020). Previous studies reported the methanogenic biodegradation of short-chain *n*-alkanes (C_6-C_{10}) in MLSB that include a significant amount of unrecovered naphtha (Siddique et al. 2006). The presence of different microorganisms in tailings and their activity depends on multiple factors such as tailings types (FFT, MFT, CT, TT), ponds age, depth, and the additives used during the bitumen extraction (Penner and Foght 2010).

1.2 Problem Statement

The problem statement of the thesis can be summarized as follows:

- I. Slow consolidation of fine tailings solid.
 - The storage of oil sands tailings faces many challenges due to the very slow consolidation of fine tailings solids which segregate from the sand component after deposition in tailings ponds. These fine tailings need decades to settle down (Chalaturnyk et al. 2002).
- II. Presence of inorganic and organic contaminants

- The elevated concentrations of inorganic compounds in both FFT and OSPW contribute to the toxicity (Allen 2008).
- The presence of organic contaminants such as petroleum hydrocarbons (PAHs) and naphthenic acids (NAs) in OSPW contribute to toxicity (Leishman et al. 2013; Sabyasachi et al. 2010). NAs are considered the major contributor to the toxicity of OSPW (Headley and McMartin 2004).
- III. The movement of COPCs in EPL
 - Although the development of EPLs would decrease the volumes of FFT stored in tailings, the practicability of establishing an aquatic ecosystem in the lake has not been fully evaluated yet due to the movement of COPCs from the FFT into the overlying water cap. The main constituents of the FFT pore water are high concentrations of dissolved COPCs, NAs, petroleum hydrocarbons, and unrecovered bitumen (Allen, 2008). The control of the transport of these constituents is still a key role in developing EPLs.
- IV. Uncontrolled GHGs emissions
 - The methane emission from tailings ponds is considered as input to atmospheric GHGs and it is currently uncontrolled (Siddique and Kuznetsova 2020).
 - The remediation of oil sands tailings faces environmental problems with the increase of the GHGs emissions from the ponds, such as CO₂ and CH₄, due to the methanogenesis activity in the tailings.

 A goal to reduce methane emissions from the oil and gas sector by 40– 45 % by 2025 was established by the Government of Canada with having regulations in place to assist in the mitigation of methane emissions from the oil and gas sector (Government of Canada 2018).

1.3 Research Scope and Objectives

This thesis focuses on the reclamation of oil sands tailings by enhancing the dewatering of FFT using chemical treatment (i.e., lime) and biological treatment using enzymes. Additionally, it focuses on mitigating GHGs emitted from tailings ponds which can be translated into new treatment technologies for oil sands tailings remediation. The specific objectives for the thesis can be summarized as follows:

I. To study the impact of adding different dosages of lime on the quality of cap water and FFT porewater in EPLs and dewatering of FFT;

- Evaluate the influence of adding different dosages of lime on the quality of cap water and FFT porewater in EPLs.
- Elucidate the role of cation exchange on the reduction and removal of organic fractions in FFT (DOC and PHCs).
- Examine the variation of water chemistry (alkalinity, pH, and cations), bacterial cell counts, and toxicity toward *Vibro fischeri*.

II. To evaluate Enzyme-assisted treatment of oil sands tailings for dewatering;

 Examine the technical feasibility of enzymatic treatment to enhance FFT dewatering.

- Investigate the corresponding mechanism for the dewatering after Enzymatic treatment.
- Study the influence of enzymatic treatment on reducing or removing organic fractions such as naphthenic acid fractions (NAFCs).
- Determine the shifts in microbial community diversity and toxicity toward V. fischeri.

III. To assess the impact of enzyme and lime treatment on the mitigation of methane emissions from oil sands tailings ponds:

- Investigate the feasibility of enzymatic and lime treatment to inhibit the methane emission from tailings ponds.
- Study the effect of enzymatic and lime treatment on the reduction of hydrocarbons.
- Determine the changes in microbial community diversity and toxicity toward *V. fischeri*.

1.4 Thesis Organization

This thesis consists of six chapters. Chapter 1 presents a general introduction to the research background, objectives, and significance. Specifically, it encompasses a brief review of the oil sands operation, tailings management, and the motivation for the current research, objectives, and thesis organization. The detailed experimental procedures, methodologies, results, and discussions are presented separately in each chapter (Chapters 3-5).

Chapter 2 provides background information about tailings management and challenges faced by oil sand operators, a problem statement, and a comprehensive overview of the related research conducted in this area.

Chapter 3 explores the outlook of adding hydrated lime (Ca[OH]₂) in the FFT dewatering and improving cap water quality under the simulated EPL scenario. The viability of lime treatment was investigated by evaluating the FFT water recovery and the EPL simulation cap water performance in terms of water chemistry and toxicity. The efficacy of the EPL simulation in accelerating the FFT dewatering and improving the cap water quality was tested under a range of lime dosages to optimize the optimum dose.

Chapter 4 introduces the investigation of enzymatic treatment (using cellulase, protease, and lysozyme) to accelerate the dewatering of fluid fine tailings (FFT). This study assessed the visibility of using 3 different enzymes to accelerate the dewatering of FFT and identifying the different proposed mechanisms for the dewatering. Also, it examined the changes in the microbial community and toxicity after enzymatic treatment.

Chapter 5 examines the inhibition of methane in tailings ponds by using chemical treatment (i.e., lime) and biological treatment using two different enzymes (lysozyme and protease). This project identifies the significance of inhibition potential of the methane emissions. This study assessed the treatment effectiveness on the inhibition of methane emissions, the degradation of hydrocarbons, the changes in the microbial community, and toxicity after treatment. Chapter 6 illustrates the major conclusion of the research presented in Chapters 2-5. Additionally, future recommendations for further research are encompassed in this last chapter. Finally, the Appendix section presents some experimental methodologies, with supplementary tables and figures in the main chapters.

1.5 References

- Allen EW. 2008. Process water treatment in canada's oil sands industry: I. Target pollutants and treatment objectives. Journal of Environmental Engineering and Science. 7(2):123-138.
- Chalaturnyk RJ, Don Scott J, Özüm B. 2002. Management of oil sands tailings. Petroleum Science and Technology. 20(9-10):1025-1046.
- COSIA. 2012. Technical guide for fluid fine tailings management Edmonton, Alberta, Canada: Canadian Oil Sands Innovation Alliance, Edmonton, AB. August 2012.
- Dompierre KA, Barbour SL. 2016. Characterization of physical mass transport through oil sands fluid fine tailings in an end pit lake: A multi-tracer study. Journal of Contaminant Hydrology. 189:12-26.
- Dompierre KAH. 2016. Controls on mass and thermal loading to an oil sands end pit lake from underlying fluid fine tailings.
- Government of Canada. 2018. Technical backgrounder: Federal methane regulations for the upstream oil and gas sector - canada.Ca, https://www. Canada.Ca/en/environment-climate-change/news/2018/04/federalmethane-

regulations-for-the-upstream-oil-and-gas-sector.Html, (accessed jan 28, 2022).

- Headley JV, McMartin DW. 2004. A review of the occurrence and fate of naphthenic acids in aquatic environments. Journal of Environmental Science and Health, Part A. 39(8):1989-2010.
- Holowenko FM, MacKinnon MD, Fedorak PM. 2000. Methanogens and sulfatereducing bacteria in oil sands fine tailings waste. Canadian Journal of Microbiology. 46(10):927-937.
- Kabwe LK, Scott JD, Beier NA, Wilson GW, Jeeravipoolvarn S. 2018. Environmental implications of end pit lakes at oil sands mine. Environmental Geotechnics.1-25.
- Leishman C, Widdup EE, Quesnel DM, Chua G, Gieg LM, Samuel MA, Muench DG. 2013. The effect of oil sands process-affected water and naphthenic acids on the germination and development of arabidopsis. Chemosphere. 93(2):380-387.
- McQueen AD, Kinley CM, Hendrikse M, Gaspari DP, Calomeni AJ, Iwinski KJ, Castle JW, Haakensen MC, Peru KM, Headley JV. 2017. A risk-based approach for identifying constituents of concern in oil sands processaffected water from the athabasca oil sands region. Chemosphere. 173:340-350.

- Penner TJ, Foght JM. 2010. Mature fine tailings from oil sands processing harbour diverse methanogenic communities. Canadian journal of microbiology. 56(6):459-470.
- Sabyasachi M, Venkatesh M, Ajay K D, Dena W M, John V H, Kerry M P. 2010. Photocatalysis of naphthenic acids in water. Journal of Water Resource and Protection. 2010.
- Samadi N. 2019. Partitioning of inorganic contaminants between fluid fine tailings and cap water under end pit lake scenario: Biological, chemical and mineralogical processes.
- Siddique T, Fedorak PM, Foght JM. 2006. Biodegradation of short-chain n-alkanes in oil sands tailings under methanogenic conditions. Environmental science & technology. 40(17):5459-5464.
- Siddique T, Kuznetsova A. 2020. Linking hydrocarbon biodegradation to greenhouse gas emissions from oil sands tailings and its impact on tailings management. Canadian Journal of Soil Science.1-9.
- Small CC, Cho S, Hashisho Z, Ulrich AC. 2015. Emissions from oil sands tailings ponds: Review of tailings pond parameters and emission estimates. Journal of Petroleum Science and Engineering. 127:490-501.
- Small CC, Ulrich AC, Hashisho Z. 2012. Adsorption of acid extractable oil sands tailings organics onto raw and activated oil sands coke. Journal of Environmental Engineering. 138(8):833-840.

- White KB, Liber K. 2018. Early chemical and toxicological risk characterization of inorganic constituents in surface water from the canadian oil sands first large-scale end pit lake. Chemosphere. 211:745-757.
- Zubot W. 2010. Water use and aquatic reclamation at syncrude canada ltd. Integrated oil sands mining operation The Association of Professional Engineers, Geologists, and Geophysicists of Alberta (APEGGA).

2 LITERATURE REVIEW¹

2.1 Water and tailings management in oil sands operation

Surface-mined oil sands ore in north-eastern Alberta produces almost twothirds of the total bitumen production in Alberta (ERCB 2012; 2009). In 2017, while supplying 2.5% of the global oil consumption, 4 million barrels of bitumen was produced per day from the Alberta oil sands (Choudhary et al. 2019; Alberta 2019; Government of Alberta 2015). In 2035, there is an expected increase in the oil sands production in Canada to reach around 5,500 thousand barrels per day (Choudhary et al. 2019; Government of Alberta 2019). The Clark Hot-Water Bitumen Extraction occurs by adding hot water and caustic soda to the sand to generate a slurry and allow bitumen to separate from the sand and clay. After the sediments settle, bitumen froth is collected at the surface of the slurry (Clark and Pasternack 1932). The recovered bitumen is then subjected to the process of upgrading in order to transform it into synthetic crude oil. This process currently demands large volumes of water in which 2 to 2.5 m³ of water is required for each m³ of crude oil produced (AER 2019; Zhu et al. 2017; Zubot et al. 2012). Thus, water management challenges are considered a fundamental part of oil sands operation.

¹ A version of this chapter will be submitted to Science of the Total Environment Journal as: Allam, N.E.; Dhar, B.R.; Ulrich, A. (2023): "A critical review on oil sands dewatering and Greenhouse Gas emissions".
The surface mining operation reuses the water to a certain extent, where a large number of slurries are generated after the bitumen extraction process and stored in engineered tailings ponds. The hot water extraction process generates massive volumes of oil sands process-affected water (OSPW) as well as fluid tailings, which consists of water, dissolved salts, an aqueous suspension of clay, silt, organics and bitumen (Allen 2008a). OSPW is usually recycled back to the extraction process a few times until it reached specific water quality that can deteriorate the bitumen quality (Allen 2008b). The produced water cannot be directly discharged into the environment due to Directive 074 discharge practice following the environmental legislation (ERCB 2012; McQueen et al. 2017b). Therefore, oil sands operators store them in tailings ponds until further treatment.

To date, the tailing ponds cover 170–180 km² while the surface area is continuously increasing as new ponds are being developed (Vedoy and Soares 2015; Young et al. 2020; Zhu et al. 2017). A schematic diagram for the traditional management of oil sands tailings ponds is shown in Figure 2.1. The suspension of the fine clay and silt particles hinders the reclamation of tailings ponds due to the slow consolidation rate of the colloidal particles. The colloidal particles need decades to settle down by self-weight consolidation (Jeeravipoolvarn et al. 2009). Hence, the growing inventory compels operators to seek alternative tailings management strategies (Allen 2008b). The presence of dissolved organics, salts, and trace metals within OSPW and FFT have been identified as compounds of potential concerns (COPCs) that possess toxicity effects on the aquatic organisms in many studies (Anderson et al. 2012; Kavanagh et al. 2011; Li et al. 2017; McQueen et al. 2017a). To reduce the environmental risks of FFT and OSPW on the landscape, oil sands operators are seeking different tailings reclamation strategies to manage the accumulated volumes of tailings.



Figure 2.1. Schematic diagram for the oil sands extraction process and traditional tailings management adapted from (Beier and Sego 2008).

2.1.1 FFT characteristics

The composition of the oil sands tailings is 20–30 wt.% of solids (i.e., sand, silt, clays, dissolved salts, organics, and minerals) as well as 70–80 wt.% water, and 1–3 wt.% as residual bitumen. In the tailings ponds, the coarse sands quickly settle when deposited, while much of the fine solids and residual bitumen remain suspended, leading to the formation of mature fine tailings (MFT). The settling of tailings usually results in three layers in which the course sands occurs at the bottom, the fine tailings in the middle (i.e., fluid fine tailing (FFT) or MFT), and an aqueous layer at the top also known as oil sands process-affected water (OSPW)

(Mackinnon and Zubot 2013; Wang et al. 2015). Fine tailings can settle to 20 wt% solids within a few weeks; however, to reach 30–35 wt% of solids, several years are required. When this solids content is reached, the consolidation becomes considerably slow, and the sludge is designated as MFT (Allen 2008a).

The composition and water chemistry of the tailings make them hard to dewater. The coarse particles settle quickly; however, the fine particles remain in suspension, with no further dewatering for many decades (Gumfekar et al. 2017). The particle size (i.e., radius) plays an essential role in settling according to Stokes settling velocity equation as shown in Equation 1 (Yang et al. 2015). This equation has been derived based on the balance of buoyancy force, drag force and gravity force which governs the particle settling velocity in a uniform stagnant fluid (Yang et al. 2015). The overall particle size distribution in the typical MFT samples are approximately 57% in the range of 1–44 μ m, 33% are less than 1 μ m and 10% are greater than 44 μ m (Dompierre et al. 2016a). In particular, the clay particles that are 0.2–2 μ m (Kaminsky et al. 2009) are highly stable in suspensions and slow to settle (Siddique et al. 2014a; Zhu et al. 2011).

$$W_s = \frac{2}{9} \frac{g' r_p^2}{v} \tag{1}$$

Where, Ws = settling velocity (cm/s)

g' = reduced gravity =
$$(g(\rho_p - \rho_w)/\rho_w)$$

 ρ_p = particle density (g/cm^3)
 ρ_w = water density (g/cm^3)

 $r_p = particle radium (cm)$

v = kinematic viscosity of water (cm²/s)

Due to the surface charge of these fine particles coagulation or aggregation is difficult, so their settling is hindered. The electrical double layer (EDL) thickness surrounding the clay particle is considered the main factor that controls the flocculation mechanism (Brandon 2016; Schofield and Samson 1954; Siddique et al. 2014a). In brief, the diffuse double layer (DDL) is formed in aqueous solution due to the surrounding of the clay particle with the negatively charged ions (coions) and the excess positively charged ions (counter-ions) as shown in Figure 2.2.

To evaluate the stability of clay in aqueous media, the zeta potential is one of the electrokinetic parameters that can be used in DLVO (Derjaguin-Landau-Verwey-Overbeek) mathematical model by balancing the attractive force (Van der Waals) and repulsive force (electrostatic) (Benitez et al. 2007; Grasso et al. 2002; Missana and Adell 2000). DLVO theory was applied to understand the stability of FFT particles by understanding the distribution of forces between two particles (i.e., repulsive and attractive forces) (Derjaguin and Landau, 1993; Li et al. 2008). The shear plane serves as a boundary to separate the fixed (stern plane) and more mobile diffuse layers, as well as a region at which the electrical potential (zeta potential) can be measured (Masliyah and Bhattacharjee 2006; Sawyer et al. 1994). Therefore, the high stability of the clay particles in suspension is due to their high zeta potential (Brandon 2016).



Figure 2.2. A schematic diagram for the Stern models (left) of an interaction between negatively charged clay surface and anions and cations in the bulk solution as well as diffuse double layer surrounding a clay particle (right). Adapted from Masliyah and Bhattacharjee (2006).

2.1.2 OSPW characteristics

OSPW contains suspended solids, salts, organic compounds, inorganic ions, and trace elements (Allen 2008b; El-Din et al. 2011). After the addition of OSPW in the tailing ponds, some solids, and the rapidly settled particles are separated from the OSPW. The fine particles with ~20% (wt.) as well as a low amount of solids form a surface water layer (Allen 2008a; Pourrezaei 2013). The presence of NAs and PAHs compounds in OSPW contribute to toxicity (Leishman et al. 2013; Sabyasachi et al. 2010). NAs has the major contribution to the toxicity of OSPW (Headley and McMartin 2004). OSPW is identified to have chronic and acute toxicity to a variety of organisms such as algae, bacteria and fish (Gosselin et al. 2010). The high toxicity of OSPW possesses an extreme risk when release into the environment. Therefore, it is necessary to be stored in tailing ponds in order to comply with the zero-discharge practice (ERCB 2012; Kannel and Gan 2012). Treating OSPW is essential to reduce fresh water quantity consumption and protect the extracted bitumen quality by using a recycled treated water with good quality (Allen 2008a).

2.1.3 Inorganic compounds

The elevated concentrations of inorganic compounds in both FFT and OSPW, compared to the natural surface waters of the Athabasca region could contribute to the toxicity (Allen 2008a). OSPW is usually alkaline (~ 800–1000 mg L^{-1} HCO₃⁻) with pH in the range of 7.9–8.4 (Allen 2008a). Since tailings water contains high concentrations of total dissolved solids (TDS, 2000–2500 mg/L), it has been classified as brackish (Allen 2008a). It contains high concentrations of dissolved solids, such as sodium, sulfate, ammonia, and chloride (Choi et al. 2014; Pourrezaei et al. 2011). During the extraction process, trace metals such as aluminum, nickel, arsenic, copper, zinc, and chromium that presents in the ores are released into OSPW (Allen 2008a; Baker et al. 2012).

2.1.4 Organic compounds

OSPW comprised a mixture of a wide variety of organic compounds, such as bitumen (measured as oil and grease), asphaltenes, naphthenic acids (NAs), phthalates and polycyclic aromatic hydrocarbons (PAHs), BTEX (benzene, toluene, ethylbenzene, xylene), phenols, and other organic acids (humic and fulvic acids) (Allen 2008a; Madill et al. 2001; Strosher and Peake 1978). Bitumen concentration in OSPW may range from 9 to 92 mg/L in Suncor tailings ponds (Gulley 1992; Nix 1983). In Mildred Lake Settling Basin (MLSB; Syncrude tailings pond) the concentration of oil and grease was reported as 25 mg/L (MacKinnon and Boerger 1986; Pourrezaei 2013). The concentration of PAH combined with BTEX concentration is typically less than 0.01 mg/L (Allen 2008a; Rogers et al. 2002).

NAs is considered the major organic COPC in the OSPW. During the bitumen extraction process, NAs are solubilized from bitumen (Morandi et al. 2017), it is mainly alkyl-substituted cyclic and aliphatic carboxylic acids (Allen 2008a). NAs are considered the primary contributor of OSPW toxicity (Meshref et al. 2017; Scott et al. 2005). In tailings ponds, the concentrations of NAs differ from 8.9 to 39.2 mg/L and it can reach 130 mg/L in some cases (Grewer et al. 2010). The general formula of NAs is $C_nH_{2n}+_ZO_x$, where n is the number of carbon atoms (7 \leq $n \le 26$), Z ($0 \le -Z \le 18$) represents the hydrogen deficiency due to the formation of the ring or double bond structure, and x shows the number of oxygen atoms present in the NAs (Balaberda and Ulrich 2021). The classical NAs are represented by O₂ species at x= 2 and the oxidized NAs species are the O₃, O₄, O₅, O₆ at $(3 \le x \le 6)$ (Meshref et al. 2017). The background concentrations of NAs in surface water such as Athabasca River is typically <1 mg/L (Allen 2008a; Headley and McMartin 2004). The natural degradation processes can result in the decrease of NAs concentrations over time. For example, NAs concentrations reduced from 73 mg/L to 33 mg/L in eight years (1995–2003) in MFT pore water at depth 10–15 m (Allen 2008a; MacKinnon 2004).

2.2 Toxicity of OSPW/FFT

Continuous accumulation and storage of oil sand tailings present significant challenges, with a high risk similar to any release of OSPW or FFT into the environment (Allen, 2008a). The primary toxicological effects of OSPW are typically related to oil sands organic acid fractions (i.e., NAs in the acid extractable organic fraction - AEO) in which NAs are major contributors and the main source in oil sands byproducts (OSPW, FFT) for the chronic (Gosselin et al. 2010) and acute toxicity towards aquatic organisms (Clemente and Fedorak 2005; Grewer et al. 2010; Li et al. 2017; Scott et al. 2005; Zubot et al. 2021). The toxicity effect attributed to NAs can vary based on the NAs composition, source, and age and the relative concentrations as low concentrations of NAs can have a relatively high toxicity. For instance, at a low NAs concentration of 16 mg/L, a Microtox IC_{50} value of 32 % for MLSB water (Holowenko et al. 2002). The complex mixture of the acid extractable organic fraction in OSPW contains O_2^- NA species and other species (Balaberda and Ulrich 2021). This mixture requires a significant decrease of the absolute bulk concentrations of the O_2^-NAs and O_x^-NAs and the removal of specific NAs structures (Meshref et al. 2017). Previous studies reported the correlation between the lower molecular weight constituents and the toxicity of the NAs toward Vibrio fischeri (Frank et al. 2008). The toxicity of NAs also could be related to their classification as surfactants with hydrophobic ends that can penetrate the organisms' cells membrane wall and cause toxicity (Beltrán-Heredia and Sánchez-Martín 2009; Crittenden and Trussell 2005). Previous studies reported the toxicity of NAs to a variety of organisms such as amphibians, plants, mammals

and fish (Beltrán-Heredia and Sánchez-Martín 2009; Crittenden and Trussell 2005; Preez and S Preston 1992; Wagner et al. 1982). Furthermore, NAs had been reported to have an adverse effect on mice's immune system (Shen et al. 2008). NAs could be introduced to the fresh surface and underground water resources through leaching Fields (Mohamed et al. 2011; Seyedy Niasar 2017). Given the above-mentioned facts, the benefits of developing passive low-cost and low energy treatment technologies for both OSPW and tailings are crucial (White et al. 2018). The oil sands industry could incorporate to ensure sustainable management of treated OSPW and the protection of the environment (Brown and Ulrich 2015; Zubot et al. 2021).

2.3 MFT remediation/dewatering technologies

To date, different remediation/dewatering technologies have been explored to tackle the increased volume of tailings. The main targets of these studies were to enhance the settling of fine particles and recover the trapped water. The following subsections will discuss the fundamental and applied aspects of these approaches and their limitations and key research gaps.

2.3.1 Physicochemical, natural, and mechanical technologies

Composite/Consolidated tailings (CT) is considered one of the first technologies developed for the tailings management (Caughill et al. 1993). For CT, gypsum is added to MFT at a sand to fines ratio of 4:1 (Beier and Sego 2008; Caughill et al. 1993). After mixing, a non-segregating slurry is formed, achieving faster consolidation and water recovery when deposited in the tailings pond (Beier and Sego 2008). One of the drawbacks of CT is the production of aqueous and gaseous H_2S as a result of gypsum addition (Reid and Warren 2016). In addition, elevated ion concentrations (e.g., NA^+ , Ca^{2+}) were reported in the CT porewater, which can posture challenges to the growth of vegetation (Huc 2011; Wu et al. 2011).

Another technology is the freeze-thawing process, where CT or MFT is loaded in multiple thin layers before the winter season to freeze the layers and then thaw in the summer season. Beier et al. (2009) reported increasing the MFT solid contents from 35% to 56% after this treatment. Several studies that used the freezethawing technology reported a considerable amount of released water when the thickness of layers is 5 cm to 15cm (Dawson and Sego; 1993; Dawson et al. 1999; Johnson et al. 1993). Paste or thickened tailing technology is another reclamation technology where flocculants, such as polyacrylamides (PAM), starch derivatives as natural flocculants and other minerals (e.g., silica, bentonite, alum and ferrichydroxide), are used to promote the settling of suspended fines particles (Farkish 2013; Wells and Riley 2007; Yuan 2009). The flocculated fines allow settling quickly and forming a stable solids bed that can be deposited in tailings ponds for land reclamation. Also, the released water can be recycled back to the bitumen extraction process (Gumfekar et al. 2019). Since the paste technology can only thicken low solids effluent, it is still not considered an effective method (Gumfekar et al. 2019). Furthermore, to improve the settling rates of tailings and release water, solid-liquid separation (mechanical treatment) processes, such as filtration, centrifuge (Allen 2008b; Gumfekar et al. 2019; Mikula et al. 2008; Wang et al. 2014) or filtration combined with polymer additives (Farkish and Fall 2014) will still be required. Centrifugation requires a high initial cost and also polymer addition requires proper control for the polymer dosage concerning the variation in the contents of the clay in the FFT (Brandon 2016; Mikula et al. 2008). Overall, all the above-mentioned remediation methods may provide a partial solution but they cannot solve the challenges of FFT and OSPW in oil sands tailings alone (Brandon 2016).

Flocculants and coagulants are widely used for solid-liquid separation, they can reduce the suspension of fine particles suspended in water, reduce the time of settlement and achieve a high solid content (Jeeravipoolvarn 2010; Yuan 2009). Previous studies found that the repulsion between clay particles and coagulating colloidal suspensions was reduced more by using salts containing multivalent ions compared with salts containing monovalent ions (Hardy 1900; Schulze 1882). Many coagulation/flocculation agents have been commonly used in drinking water treatment and wastewater treatment to enhance the settling of fine particles such as ferric chloride, calcium sulfate, and aluminum sulfate (White et al. 2011). In the same sense, in the oil sands industry, multivalent cations have been used to improve the tailings dewatering (Chalaturnyk et al. 2002; Long et al. 2006; Sworska et al. 2000). Alum and gypsum are considered two of the more successful coagulant aids that have been used for oil sands tailings reclamation Fields (MacKinnon et al. 2001; Matthews et al. 2002). Alum (Al₂(SO₄)₃·14.3 H₂O) is a trivalent cation Al³⁺ that has been used as a coagulant aid in the oil sands industry to accelerate the dewatering of FFT. In the CT treatment process, FFT, sand fractions, and alum are

mixed to rapidly dewater the FFT (Matthews et al. 2002). In CT, a lower amount of alum is required compared to gypsum which can illustrate the role of the trivalent cation Al^{3+} ion versus the divalent cation Ca^{2+} (MacKinnon et al. 2000; Redfield et al. 2003).

Coagulation/flocculation by the addition of divalent cations can enhance the settling rate or densification of clay particles in an aqueous media (Chalaturnyk et al. 2002; Chen et al. 2013b; Zhu et al. 2011). Previous studies investigated using lime as a sole coagulant or in combination with gypsum or polymers to treat tailings such as non-segregating tailings and rim ditching drying (Baillie and Malmberg 1969; Beaty and Lane 1983; Caughill et al. 1993; Chalaturnyk et al. 2002; Ewin et al. 1981; Hamza et al. 1996; Lorentz et al. 2014). Caughill et al. (1993) reported that the addition of Ca²⁺ as a flocculating agent converts the tailings into a nonsegregating mix, where the fines aggregate and clasp sand particles. The formation of non-segregating tailings is a function of tailings solid fines content (17% (wt%) of <44 mm particles of the total solids) and total fines ratio (wt% of water and fines together). Using different hydrated lime Ca[OH]₂ and gypsum CaSO₄ as flocculants improve the formation of the non-segregating tailings (i.e., the lower solids concentrations) (Chalaturnyk et al. 2002). Chalaturnyk et al. (2002) investigated the effectiveness of combining Ca[OH]₂ and CO₂ followed by a suitable thickener in accelerating the FFT released water. They reported that the FFT could dewater by 70% in 50 hours compared with 60% without CO₂ addition. In addition, using 2000 ppm of CaO at 70 °C, accelerated the dewatering of FFT by half in two hours. Proskin et al. (2010) reported an increase in the hydraulic conductivity of FFT that results in faster consolidation after the addition of quicklime before freeze-thawing. Tate et al. (2016) reported that adding Ca[OH]₂ alone as a coagulant to a diluted FFT sample at a dose of 1250 ppm resulted in the settling of FFT. In general, adding Ca[OH]₂ causes an increase in pH and provides Ca²⁺ cations, which is required for improving the FFT coagulation. Cation exchange improved at pH > 11, as calcium ions become soluble and react with the negative charge at the surface of clay particles (Tate et al. 2016). Romaniuk et al. (2015) investigated the influence of CaO on the recovery of residual bitumen from FFT. They reported that using doses of 750 to 1500 mg-CaO/kg-MFT could facilitate the floatation of residual bitumen and hence improve the dewatering of MFT.

Due to the high efficiency in dewatering oil sands tailings, flocculation with polymers has been widely explored and implemented by the oil sands industries over the last two decades (Farkish and Fall 2013; Vajihinejad et al. 2021; Vedoy and Soares 2015). In brief, polymers (as flocculants) are generally employed to overcome the potential energy barrier between the negatively charged solid particles in the tailings (Lu 2016; Lu et al. 2016). The polymers can prompt bringing the particles into the effective range of attractive Van der Waals force and polymer bridging attraction to trigger the destabilization and flocculation of suspended particles (i.e., forming large flocs/or sediments) (Ji et al. 2013). Different mechanisms of destabilization could be involved in the flocculation process based on the composition of tailings samples and the properties of employed polymer flocculants. These mechanisms include charge patching, bridging, and charge neutralization (O'Shea et al. 2011). Different polymers have been used in the

industrial flocculation of oil sands tailings. For instance, Polyacrylamide (PAM) has been widely used in the industrial flocculation of oil sands tailings Fields (Amoako 2021; Haveroen et al. 2005). PAM could be found in different forms, and it is varied due to different parameters such as charge densities, molecular weights, and charges (anionic, cationic, or non-ionic) (Cossey et al. 2021; Haveroen et al. 2005). Ji et al. (2013) investigated the effect of three polymers: PAM, Al (OH)₃polyacrylamide (AL-PAM), and Magnafloc 1011 (MF), on enhancing the settling of tailings minerals in freshwater and saline solution. MF resulted in the best performance for tailings flocculation in both cases saline and freshwater. Farkish and Fall (2013) examined the superabsorbent polymer (SAP) to produce dense MFT. SAP is a hydrophilic gel that consists of a network of polymer chains, relative to their mass it is capable to adsorb and retain massive amounts of water (Staples and Chatterjee 2002). The results illustrated a significant improvement in the tailings dewatering, density and an increase in the undrained shear strength of MFT (Farkish and Fall 2013). Polymeric flocculants not only have been used alone in the oil sands industry, but also combination with other materials in other applications. In CT, polymers are combined with electrolyte coagulants (Matthews et al. 2002). During tailing reduction operation (TRO), MFT is mixed with polymer anionic PAM (Mamer 2010).

2.3.2 Biological processes

Biological treatment has also been investigated to treat or remediate oil sands tailings and densify high fine content in tailings by bacterial actions (Fedorak et al. 2003; Guo 2009). Biological treatment is a well-known bioremediation method where bacteria, archaea, fungi, and protists can biodegrade organics to intermediate products and mineralize the organics completely into water and biogenic gases (CO₂ and CH₄) and achieve high organics removal (Clemente and Fedorak 2005; Leahy and Colwell 1990). Microbial activity exhibited great potential for OSPW detoxification, turbidity removal, heavy metals, and biodegradation of the organics, such as NAs, residual bitumen, and diluent. It can also accelerate the tailings dewatering (Allen 2008a; Brown and Ulrich 2015; Mahdavi et al. 2012; Nelson et al. 1993; Siddique et al. 2014b). Previous studies have examined the role of native microbial communities in OSPW and tailings microbial remediation (Herman et al. 1994; Mahdavi et al. 2015; Nix and Martin 1992; Siddique et al. 2012).

Previous research identified the presence of harbour indigenous microbial communities in tailings ponds that can anaerobically biodegrade the organic compounds such as BTEX and hydrocarbons to biogenic gases CH₄ and CO₂ (Penner and Foght 2010; Siddique et al. 2011) (Figure 2.3). In the MLSB tailings pond, diverse aerobic and anaerobic microorganisms have been identified, including methanogens, denitrifies, iron and sulfate reducing bacteria (Holowenko et al. 2000; Penner and Foght 2010). Tailings ponds provide highly anaerobic environments due to the high depth of FFT and tailings material properties that prevent light and oxygen from entering the ponds. Thus, methanogenic archaea can be enriched in FFT. Both acetoclastic and hydrogenotrophic methanogens were identified in the oil sands tailings (Foght et al. 2017; Penner and Foght 2010; Samadi 2019; Siddique and Kuznetsova 2020; Siddique et al. 2015). However, there is a syntrophic relationship between bacteria and archaea in methanogenesis.

Bacteria can initially biodegrade complex organic substrates into simple byproducts (i.e., acetate, hydrogen, and carbon dioxide). Then, methanogens use these byproducts as precursors to produce methane by either acetoclastic methanogens (utilize acetate to produce methane) or hydrogenotrophic methanogens (consume H₂ and CO₂) to produce methane (Siddique et al. 2012; Siddique et al. 2018). Figure 2.3 shows BTEX and hydrocarbons biodegradation pathways to CH₄ and CO₂.



Figure 2.3. The proposed pathway of the anaerobic hydrocarbon degradation in oil sands tailings (adapted from Siddique et al. (2011)).

2.3.3 Application of enzymes for dewatering

Enzymes are proteins with a catalytic activity that have been previously used as conditioners for enhancing sludge dewaterability in the conventional wastewater treatment (Ayol and Dentel 2005; Dursun et al. 2006; Thomas et al. 1993). Many studies have examined the possibility of using enzymes to enhance the dewatering of wastewater sludge. Thomas et al. (1993) reported improvement of digested sludge dewaterability by using an enzymatic product consisting of carbohydrase, lipase and protease. Enzymes improved the dewatering of anaerobically digested sludge at the lab scale (27% increase in the cake solid content); however, the centrifuge pilot-scale did at the Wilmington, Delaware (US)

wastewater treatment plan not show the same improvement in the dewatering (only 20% increase in the cake solid content-field; Dursun et al. 2006). The reason for the lack of the improvement in the solid content in the pilot-scale is due to the high shear applied during the centrifuge which leads to floc deterioration. Recently, the dewaterability of deinking sludge (DS) after enzymatic treatment using cellulase enzymes was enhanced (i.e., 10–14% increase in cake solid content) (Steffen et al. 2018). In another study by Wu et al. (2016), a reduction in capillary suction time (CST) by 48.8% and enhancement in waste-activated sludge (WAS) dewatering were observed after the treatment with composite hydrolysis enzymes. Additionally, cellulase and protease were used to enhance the dewaterability of WAS from the paper mill (Wu et al. 2014). The authors reported high solid content at a dose 0.5%compared with other doses (0.25, 0.75 and 1%) for both cellulase and protease Fields (Wu et al. 2014). Lu et al. (2011) investigated the use of cellulase enzymes to increase the cake solids content and improve the dewatering of wood pulp fibers with a belt filter press simulator.

Bonilla et al. (2015) revealed the mechanisms of enzymatic treatment and their effect on the physical and chemical properties of pulp and paper mills sludge. In their study, five different enzymes (cellulase, amylase, protease from Bacillus licheniformis, protease from Bacillus sp. and lysozyme) were examined; however, only lysozyme could improve the dewaterability of the sludge at an optimum dose of 0.5% (wt.%) (i.e., increase in the dry solids content of the cake from 5.8% to 8.9%). In terms of the effectiveness of enzymatic activity (active or thermo-inactive), it has been reported that doses in the range of 0.2–0.6% (wt.%), showed

a similar enhancement of the dewaterability. The mechanism of dewatering was thought to be a physicochemical interaction between lysozyme and the sludge particles. Lysozyme was reported to act as a cationic polymer (Bonilla et al. 2015). At a pH of approximately 7.4, a lysozyme that carries a positive charge on its surface can interact with the negative charge on the sludge particles. Thus, the repulsion force will be reduced, and aggregation of particles will improve (i.e., larger particles size range). Consequently, the sludge dewaterability can be improved (Bonilla et al. 2015).

Despite the positive reports on the effectiveness of enzymes in dewatering, enzymatic treatment has not yet been thoroughly explored for accelerating the dewatering of oil sands tailings. Only one study used a proprietary biological amendment of microbes, enzymes, and organic carrier called UltraZyme, to accelerate the dewatering of oil sands tailings (Yu et al. 2018). They reported that using 1000 ppm of UltraZyme could increase the MFT solids content from 34 wt.% to 38 wt.% in 50 days. The authors reported that UltraZyme could trigger the microbially induced accelerated dewatering of MFT through the methane flux, which created channels to pressurize pore water to release. Also, they reported an increase in the concentration of divalent ions such as Ca²⁺ and Mg²⁺ due to the dissolution of minerals resulting in an increase in clay flocculation by increasing the ionic strength and decrease the DDL. To date the feasibility of using enzymatic treatment (such as: cellulase, protease and lysozyme) to accelerate the dewatering of oil sands tailings has not been investigated yet.

2.3.4 End pit lakes

Due to the slow gravity settling of tailings, other non-natural strategies are warranted to effectively accelerate the settling of suspended solids. Furthermore, the growing inventory compels oil sand mine operators to seek alternative tailings management strategies to integrate FFT into reclaimed landscapes (ERCB 2012). Of these potential strategies, demonstrated end pit lakes (EPLs) have been hypothesized and emerged as one of the options to reduce FFT inventories within tailings ponds. EPLs are created in old mined-out pits by pumping in fresh water and OSPW to cap the FFT (Figure 2.4). The water cover is initially dominated by OSPW (Dompierre et al. 2016b). The water cover quality is expected to be improved over time by the in situ biogeochemical processes and the continuous freshwater inputs (e.g., precipitation and runoff from the nearby reclaimed areas). This water cap will facilitate FFT integration into mine closure landscapes (Dompierre and Barbour 2016a).



Figure 2.4. EPL content and performance over time starting from the early stage (on the left), intermediate stage and future stage (on the right). Adapted from (Westcott and Watson 2005).

Base Mine Lake (BML) is the first full-scale Athabasca region EPL commissioned in 2013 and operated by Syncrude Canada Ltd. BML was filled with 186 million m³ FFT (45 m depth) and capped with 90% OSPW and 10% freshwater. The initial solids content of the FFT placed in BML was approximately 35% (wt/wt). The increase of FFT solids content in BML with depth can be attributed to consolidation by self-weight (Dompierre and Barbour 2016b). Although the development of EPLs would decrease the volumes of FFT stored in tailings, the practicability of establishing an aquatic ecosystem in the lake has not been fully evaluated yet due to the movement of constituents of potential concerns (COPCs) from the FFT into the overlying water cap. The main constituents of the FFT pore water are high concentrations of dissolved COPCs, NAs, petroleum hydrocarbons, and unrecovered bitumen (Allen, 2008). The control of the transport of these constituents is still a key role in the development of the EPLs (Dompierre et al. 2016b).

Previous studies investigated an improvement in FFT dewatering and settlement of fine particles due to the methanogenesis process within the FFT layer where hydrocarbons (i.e., n-alkanes and monoaromatics) are degraded and produce CH₄ and CO₂ (Fedorak et al. 2003; Siddique et al. 2006; 2007; 2014b; 2011). Siddique et al.(2014b) illustrated the associated mechanism, where the CO₂ dissolved in the MFT endorses the dissolution of carbonate minerals. Subsequently, the divalent cations such as Ca²⁺, Mg²⁺ are released to FFT pore water and exchanged with Na⁺ at the surfaces of clay minerals. Therefore, enhancement in the FFT dewatering and settlement were reported due to the reduction in the electrical double layer (EDL) thickness (Brown et al. 2013; Siddique et al. 2014b).

Furthermore, in FFT deposits Fe and microbial SO_4^{2-} reduction are considered very essential anaerobic respiration processes (Stasik et al. 2014; Stasik and Wendt-Potthoff 2014). During these processes, H₂S and Fe²⁺ are released as a result of the reduction of SO_4^{2-} and Fe³⁺ associated with the organic carbon oxidation (Siddique et al. 2014c; Stasik et al. 2014). The secondary Fe²⁺ sulfide precipitate and the dissolved Fe²⁺ promote the precipitation of siderite (FeCO₃) within the FFT (Chen et al. 2013a; Dompierre et al. 2016b; Siddique et al. 2014c). The surface charge neutralization between these secondary phases and clay particles will promote the aggregation (Siddique et al. 2014c).

2.4 Tailings Emission

The release of CH₄ from tailings ponds is over 21 times that of CO₂; thus, it represents an environmental concern as a greenhouse gas (GHG) (Small et al.

2015). Many components influence the emissions from tailings ponds, such as organic diluents, citrate, and residual bitumen. These components serve as substrates for tailings indigenous microorganisms and metabolize them into GHG under methanogenic conditions (Siddique and Kuznetsova 2020; Small et al. 2015; Small et al. 2012). Unrecovered bitumen is considered the key organics in the MFT, which is hard to biodegrade (Holowenko et al. 2000).

The bitumen represents around 2-5% w/w of the bitumen extraction process tailings (MacKinnon 1989). During the bitumen extraction process, oil sands operators use different solvents. Some of them use naphtha such as Syncrude Canada Ltd., Suncor, and Canadian Natural Resource (CNRL), and some use paraffinic solvents, such as Canadian Natural Upgrading Limited (CNUL), Imperial Oil, and Suncor at Fort Hills (Siddique and Kuznetsova 2020). Naphtha consists of a mixture of n-alkanes (i.e., heptane, octane, nonane), monoaromatics BTEX (i.e., benzene, toluene, ethylbenzene, xylenes), iso-alkanes and cycloalkanes (Siddique et al. 2007; Siddique and Kuznetsova 2020). The composition of paraffinic solvent is relatively simpler than naphtha (e.g., C_5-C_6 *n*-and iso-alkanes). A very small amount of naphtha is unrecovered during the extraction process (<1%), which escapes into tailings ponds and serves as a substrate for methanogens in FFT (Siddique et al. 2007). Siddique et al. (2006) reported the methanogenic biodegradation of short-chain *n*-alkanes (C_6-C_{10}) in MLSB that include a significant amount of unrecovered naphtha. Siddique et al. (2007) reported metabolism of only 15-23% of whole naphtha, especially *n*-alkanes (nonane >octane > heptane) and BTEX compounds (toluene > o-xylene > m-xylene).

However, no changes were observed for other naphtha constituents, such as (i.e., iso-alkanes and cycloalkanes) (Siddique et al. 2007).

Siddique et al. (2012) reported the dominance of Firmicutes (Peptococcaceae) and Proteobacteria (Syntrophus/Smithella) in Syncrude MFT as well as acetoclastic methanogens (Methanosaetaceae) during the biodegradation of short-chain *n*-alkanes (C_6-C_{10}) and monoaromatics (BTEX) compounds. The biodegradation of long-chain *n*-alkanes (C₁₄, C₁₆, and C₁₈) was reported under methanogenic conditions after a lag phase of 180 days (Siddique et al. 2011). Another report was published on the methanogenic degradation of major isoalkanes (C₇–C₉) and cycloalkanes in the MLSB MFT (Siddique et al. 2020). The report indicated biodegradation of iso-alkanes in the MFT after a lag phase of ~ 630 days, while complete degradation took 1700 days; however, the cycloalkanes remained undegraded. The time required for biodegradation and the lag phase of short-chain *n*-alkanes (C_6-C_{10}) < longer-chain *n*-alkanes ($C_{14}-C_{18}$) < iso-alkanes (C7-C9) (Siddique and Kuznetsova 2020). Siddique et al. (2015) evaluated the methanogenic biodegradation of the n-and iso-alkanes (C5-C6) for MFT collected from tailings ponds (e.g., Muskeg River Mine) operated by CNUL and uses the paraffinic solvent as a diluent. During 6 years of incubation, biodegradation happened between 900 to 1800 days. Such longer lag phases could be attributed to the young age of the CNUL MFT compared with MLSB. The differences in MFTs from different operators (Albian and CNRL) may result in differences in the diversity of bacterial communities during the biodegradation of longer chain nalkanes (C_6-C_{18}) (Shahimin et al. 2016; Shahimin and Siddique 2017). Both

acetoclastic (e.g., *Methanosaetaceae*) and hydrogenotrophic methanogens (e.g., *Candidatus* Methanoregula) were co-dominant in the CNRL MFT containing naphtha, BTEX, long-chain *n*-alkanes (C_{14} – C_{18}), and some iso-alkanes (Shahimin et al. 2016; Shahimin and Siddique 2017). During biodegradation of the two-alkane and four-alkane in the CNRL amended MFT, *Methanosaetaceae* was most dominant (~87–93%). However, further microcosm study showed the dominance of *Candidatus* Methanoregula (~74%) over the *Methanosaetaceae* (Shahimin et al. 2016).

2.4.1 Greenhouse Gases

Due to methanogenic activity, the remediation of oil sands tailings might face a potential challenge of GHGs emissions from the ponds. Many factors affect the presence and activity of different microorganisms in tailings, such as tailings types (FFT, MFT, CT, TT), depth, the tailings ponds age and the additives used during the bitumen extraction (Penner and Foght 2010; Ramos-Padrón et al. 2010). The methane production in 2000 from MLSB was estimated at up to 43 million L/day (Holowenko et al. 2000). Furthermore, toxic compounds such NAs, and H₂S gas might transfer from the FFT into the overlying water cap during the methane flux (Holowenko et al. 2000). The emission of GHGs from tailings ponds raises environmental concerns. The gaseous emissions from MLSB are currently uncontrolled and are input into atmospheric greenhouse gases (Siddique and Kuznetsova 2020). Siddique et al. (2008) investigated the kinetics of GHG emissions from tailings. They used the data obtained from the literature (Siddique et al. 2006; Siddique et al. 2007), where Syncrude MFT from MLSB was spiked with naphtha, monoaromatics, and short-chain *n*-alkanes (C₆–C₁₀) and incubated for 365 days (Siddique et al. 2008). Only *n*-alkanes (C₆–C₁₀), toluene, and isomers of xylenes were used to predict the emissions of CH₄ for the kinetic analysis. Other naphtha components such as iso-alkanes, cycloalkanes, benzene, and ethylbenzene, were excluded from the analysis due to their trivial degradation (Siddique et al. 2008). Siddique et al. (2008) reported that degradation of naphtha (0.01–1.0 wt%) would follow zero and first-order kinetic models, CH₄ production from naphtha degradation in MLSB was estimated at 8.9–400 million L day⁻¹.

Kong et al. (2019) developed a kinetic model including microbial growth and death rates, an extended range of biodegradable hydrocarbons, and nitrogen (N) as a growth-limiting nutrient. The predicted CH₄ emissions using the model were compared with the reported CH₄ emissions from tailings ponds. The model predicted 50%-55% and 77%-95% of the reported emissions from Syncrude MLSB and CNUL tailings ponds measured emissions. These results illustrated that other endogenous organic compounds (e.g., additional labile diluent hydrocarbons, recalcitrant hydrocarbons, slowly-degradable metabolites produced, and labile organic matter associated with clays in oil sands ores) exist in oil sands tailings ponds and support the CH₄ production rather than the compounds considered in the model. Moreover, several studies focused on predicting CH₄ production under longer incubation time (> 4 years) for different types of MFTs fields (Shahimin and

Siddique 2017; Siddique et al. 2020). Shahimin and Siddique (2017) illustrated the role of microbial metabolism in the degradation of carbon from hydrocarbons to CH₄. GC-FID (purge and trap) was used to quantify the mass of individual alkanes consumed during the incubation (i.e., maximum theoretical CH₄ production). Then the theoretical production was compared with the measured CH₄ in microcosms during the experiment. The results illustrated a variation of 68 and 88% of the predicted theoretical maximum CH₄ in Albian and CNRL MFT amended with paraffinic solvent during the 1600 d incubation period. Siddique et al. (2020), compared the methanogenic biodegradation of iso- and cycloalkanes by incubating them for 1700 d with MFT with the measurements of CH₄ in live amended cultures after subtracting the CH₄ productonproduction from endogenous substrates (i.e., measured in parallel live). The amended cultures (i.e., with MFT) produced around 56-73% CH₄ compared to the theoretical CH₄ at 1700 d; however, in the No-MFTamended cultures, less biodegradation of hydrocarbons to CH4 was observed at 43% of the theoretical maximum. These results illustrated the feasibility of using kinetic models for predicting in situ methane from different tailings ponds.

2.5 Key research gaps

The enzyme applications in oil sands dewatering were rarely explored. In contrast, municipal sludge, such as waste-activated sludge (WAS), was widely examined for enzymatic treatment to improve the sludge dewaterability. The summary of enzyme applications for municipal and pulp and paper mill sludge dewatering is illustrated in Table 2.1.

The first attempt of the enzymatic treatment was reported by Thomas et al. (1993), where hydrolytic enzymes were utilized to improve the dewatering of sludge. The study examined the hydrolytic enzymes in both labs and full-scale belt press. The study highlighted the benefit of combining belt-filter presses preceding the enzymatic treatment with flocculation aid to improve the dewatering. However, despite improved dewatering, the need for both mechanical dewatering and flocculant aid presents economic concerns. Using commercial Enviro-Zyme 216 (enzyme mixture of protease and lipidase and anaerobic bacteria, Aspergillus), Wu et al. (2016) showed the enhanced dewatering of WAS by composite hydrolysis enzymes. The study referred to the sensitivity of the reaction temperature in which the preferential temperature range of the culture of Aspergillus oryzae is 25–35 °C. The dewatering of WAS by Enviro-Zyme 216 could be limited at high temperatures> 35 °C, due to the negative effect on enzyme activity at high temperatures. Recently, Lin et al. (2019) reported a positive impact on the combination of lysozyme and freeze-thaw conditioning effect on sludge dewaterability. The work suggested optimum conditions for freeze-thaw conditioning following the enzymatic treatment at a freezing temperature of -26.56°C and a freezing time of 5.78 h. The cell walls of bacteria have been broken down by lysozyme conditioning and the bound water has been released consequently. To date, enzymatic treatment has not yet been thoroughly investigated on the oil sands remediation field. Yu et al. (2018) reported an enhancement in the dewatering of tailings after using a proprietary biological amendment (UltraZyme). Further research still necessary to the explore the

practicability of using different enzymes such as cellulase, protease and lysozyme to accelerate the dewatering of oil sands tailings.

In contrast to the enzymatic treatment, the application of lime or calcium hydroxide (Ca[OH]₂) for oil sands tailings dewatering was previously considered in previous studies (see Table 2.2). The initial concept of using lime in oil sands was to expedite the rate of tailings settling and provide a coagulant and flocculant aid (Chalaturnyk et al. 2002; Hamza et al. 1996; Lorentz et al. 2014; Romaniuk et al. 2015; Tate et al. 2017). For instance, Hamza et al., (1996) demonstrated the positive impact of co-flocculation of tailings with the slaked lime treatment and a low dose of a high molecular weight anionic PAM. In their study, lime as a secondary flocculant on tailings treatment followed by a polymer flocculant showed an improvement in the dewatering rate of tailings.

Previous research highlighted the benefits of lime addition and stabilization on enhancing the mineral properties in both soil (Locat et al., 1990) and oil sands tailings (Chalaturnyk et al., 2002; Farkish, 2013; Hamza et al., 1996; Lane, 1983; Tate et al., 2017). Lime was endorsed as an approach for tailings in which a reasonable improvement of soil properties and high-water separation efficiency can be attained (Chalaturnyk et al., 2002; Lane; 1983; Tate et al., 2017). Lane (1983) suggested coagulation process as a good description of the lime stabilization in soil and tailings. Lime addition can modify particle surface charge and lead to agglomeration and interparticle bonding.

A few studies also examined the ability of lime as a sole coagulant (Lorentz et al. 2014; Tate et al. 2017). Chalaturnyk et al. (2002) found that lime could not improve the water release from tailings; however, the combination of lime and CO_2 followed by thickening resulted in a reduction in the time for the water to release without any increase in the concentrations of ions. Similarly, Tate et al. (2017) investigated the lime alone for the tailings settling. The study reported that the increase of pH above 11 could increase the solubility of Ca²⁺. At this pH level cation exchange start to occur where Ca^{2+} combines with the clay particle. Therefore, an increase in the particle size of the tailings and enhancement of tailings settling were reported after lime addition. Tate et al. (2017) highlighted the effect of samples dilution as one of the limitations of settling improvement. Nevertheless, a similar impact of lime addition on FFT coagulation was reported in the literature due to the increase of pH and providing the divalent Ca²⁺ ions (Chalaturnyk et al. 2002; Tate et al. 2017). Based on literature, the max pH can be attained after lime treatment is 12.5, any addition to extra lime will not increase the pH above this range (Burnham et al. 1992; Christy et al. 1990). Burnham et al. (1992) reported that the solubility constant (Ksp) of the lime at 25°C is 1.4×10^{-6} . Additionally, Lane (1983) reported the occurrence pozzolanic reaction after lime addition at pH above 11, in which a stabilization process can occur. The pozzolanic reactions can be described as the soluble Ca²⁺ reacts with alumina and silica of the clay forming hydrated minerals of calcium aluminate and silicate. In the same sense, the study elucidated the stabilization reactions as the pozzolanic reactions and carbonation reactions (Lane, 1983).

Table 2.1. Application of various enzymes for dewatering of municipal, pulp and paper mill sludge.

Enzyme	Application	Key focus	Key findings	Research gaps/further research needs	Reference
Hydrolytic enzymes	Sludge dewatering	Improving the dewaterability of sludge in both lab and full- scale belt press tests.	Using enzymes with polymer flocculant can improve the mechanical dewatering of belt-filter presses.	Using enzymes alone was not able to improve the sludge dewatering.	(Thomas et al. 1993)
Enviro-Zyme 216	Anaerobic digested biosolids	Examining the effect of biosolids pre-treatment with enzyme on the amount of polymer conditioning.	The combination of enzyme and polymer enhanced the dewatering of biosolid samples, and increase the solid contents using CST,	The sole effect of enzyme was not assessed.	(Ayol and Dentel 2005)
Enviro-Zyme 216	Anaerobically digested sludge	Assessing the impact of enzyme pre-treatment on sludge dewaterability at both laboratory and pilot scale.	27% increase in cake solid content after enzyme pre-treatment for the lab experiment and only 20% increase in the pilot scale.	Failure to achieve high solid content for the pilot scale compared with lab scale.	(Dursun et al. 2006)

Enzyme	Application	Key focus	Key findings	Research gaps/further research needs	Reference
Cellulase	Wood pulp fibers or (cellulosic sludge)	Examining the dewatering of fibers using a belt press simulator.	An increase in cake solids (3%–6%) was achieved after enzyme treatment.	Only a little increase in the cake solids was achieved.	(Lu et al. 2011)
Lysozyme, Cellulase Protease and a-Amylase	Pulp and paper mill biosludge	Investigating the effect of enzymes treatment in the dewaterability of biosludge.	5.6% to 8.9% increase in the Dry solids content after lysozyme treatment.	Longer incubation period was required instead of CST to assess the behavior of enzyme in the long run.	(Bonilla et al. 2015)
Enviro-Zyme 216	Waste- activated sludge (WAS)	Improving the dewatering of WAS by hydrolysis enzymes using CST.	48.8 % reduction in CST was achieved with enzyme dosage of 1 mg/g DS.	The effect of enzymes treatment needs to be addressed for long time experiment.	(Wu et al. 2016)
Cellulase	Deinking sludge (DS)	Investigating the effect of four cellulase on the DS conditioning under neutral pH and without pH adjustment (the pH of DS).	14.4% increase in cake solids was achieved by cellulase. 20% reduction on the annually generated	Limited to the applications at neutral pH.	(Steffen et al. 2018)

Enzyme	Application	Key focus	Key findings	Research gaps/further research needs	Reference
			amount of DS at neutral pH.		
Alpha amylase + and protease	Activated sludge	Examining the combination of different enzymatic lysis with chemical flocculant (FeCl ₃ and PACl) on the dewatering of activated sludge.	An improvement in the cake solid content of sludge and filtration rate after $\dot{\alpha}$ -amylase treatment compared with protease.	Enzymes were used as a pre-treatment step only.	(Chen et al. 2015)
Lysozyme (LZM)	Activated sludge	Examining the combination of LZM and freeze-thaw (26.56°C and 5.78 h) conditioning effect on sludge dewaterability.	19.84% decrease in the water content was achieved by lysozyme treatment.	Freeze-thaw conditioning was implemented for the enhancement of dewatering.	(Lin et al. 2019)
UltraZyme	Oil Sands Tailings	Assessing the effect of UltraZyme on the dewatering of three tailings sources after treatment.	30% increase of the solids content in 112 days was achieved using 1.0 g/L of UltraZyme in all three tailings sources without physical mixing.	The effect of a commercial enzyme needs to be addressed instead of the proprietary biological amendment (UltraZyme).	(Yu et al. 2018)

	Table 2.2. Summar	y of previ	ous studies th	nat used lime	for oil sands	s tailings p	processing and treatment.
--	-------------------	------------	----------------	---------------	---------------	--------------	---------------------------

Key focus	Key findings	Research gaps/further research needs	Reference
Using lime as a secondary flocculant on tailings treatment followed by polymer flocculant.	Oil sands tailings can be co-flocculated after slaked lime treatment followed by a low dose of a high molecular weight anionic PAM.	The sole impact of lime treatment on the tailings dewatering was not assessed.	(Hamza et al. 1996)
Studying the impact of pH adjustment (12.0 to 11.6) by injecting CO ₂ and adding lime as a flocculant agent on the consolidation of tailings.	The combination of using lime and CO_2 and then thickening using thickener resulted in a reduction in the retention time for the recovered water to release without any increase in the concentrations of ions such as Ca^{2+} .	Using lime alone could not improve the water release from tailings samples compared with lime and CO ₂ .	(Chalaturnyk et al. 2002)
Understanding the effects of lime coagulation on the reclamation properties of FFT	At pH >11, Ca ²⁺ becomes soluble and combines with clay through cation exchange, resulting in increased particle size and enhanced settling. At pH >12, pozzolanic chemical reactions between Ca ²⁺ and clay improve FFT dewatering.	FFT samples without dilution showed marginal improvement in settling.	(Romaniuk et al. 2015)

Key focus	Key findings	Research gaps/further research needs	Reference
Application of rim ditching to dewater the oil sands tailings and additives (lime, gypsum and polymer) to produce a trafficable deposit for reclamation.	The tailings properties (solids fraction) were improved after using additives and the tailings strength have been developed to yield a non-segregating deposit for reclamation. The consolidation and dewatering of tailings have been accelerated due to the control of release water and rainfall or runoff.	Limited information found on the released water and dewatering performance.	(Lorentz et al. 2014)
Examining the effect of using lime as a coagulant to enhance the treatment of oil sands tailings	Lime addition increases pH >12, it provides divalent Ca ²⁺ for coagulation of FFT. This increases particle size and enhances settling.	In this study diluted FFT samples was to improve the settling.	(Tate et al. 2017)

2.6 References

- AER. 2019. State of fluid tailings management for mineable oil sands, 2018. Alberta Energy Regulator: Calgary, AB, Canada.
- Alberta Go. 2019. Electronic resource about oil sands. [www document], url https://www.Alberta.Ca/about-oil-sands.Aspx. Bioresource Technology Reports. 11:100547.
- Allen EW. 2008a. Process water treatment in canada's oil sands industry: I. Target pollutants and treatment objectives. Journal of Environmental Engineering and Science. 7(2):123-138.
- Allen EW. 2008b. Process water treatment in canada's oil sands industry: Ii. A review of emerging technologies. Journal of Environmental Engineering and Science. 7(5):499-524.
- Amoako KA. 2021. Geotechnical behaviour of two novel polymer treatments of oil sands fine tailings.
- Anderson J, Wiseman SB, Moustafa A, El-Din MG, Liber K, Giesy JP. 2012. Effects of exposure to oil sands process-affected water from experimental reclamation ponds on chironomus dilutus. water research. 46(6):1662-1672.
- Ayol A, Dentel SK. 2005. Enzymatic treatment effects on dewaterability of anaerobically digested biosolids-ii: Laboratory characterizations of drainability and filterability. Process Biochemistry. 40(7):2435-2442.

- Baillie RA, Malmberg EW. 1969. Removal of clay from the water streams of the hot water process by flocculation. Google Patents.
- Baker LF, Ciborowski JJH, MacKinnon MD. 2012. Petroleum coke and soft tailings sediment in constructed wetlands may contribute to the uptake of trace metals by algae and aquatic invertebrates. Science of the Total Environment. 414:177-186.
- Balaberda AL, Ulrich AC. 2021. Persulfate oxidation coupled with biodegradation by pseudomonas fluorescens enhances naphthenic acid remediation and toxicity reduction. Microorganisms. 9(7).
- Beaty NB, Lane MD. 1983. The polymerization of acetyl-coa carboxylase. Journal of Biological Chemistry. 258(21):13051-13055.

. Natural dewatering strategies for oil sands fine tailings. 2009 2009.

- Beier N, Sego D. 2008. The oil sands tailings research facility. Geotechnical news. 26(2):72-77.
- Beltrán-Heredia J, Sánchez-Martín J. 2009. Removal of sodium lauryl sulphate by coagulation/flocculation with moringa oleifera seed extract. Journal of hazardous materials. 164(2-3):713-719.
- Benitez EI, Genovese DB, Lozano JE. 2007. Effect of ph and ionic strength on apple juice turbidity: Application of the extended dlvo theory. Food Hydrocolloid. 21(1):100-109.
- Bonilla S, Tran H, Allen DG. 2015. Enhancing pulp and paper mill biosludge dewaterability using enzymes. Water research. 68:692-700.
- Brandon JT. 2016. Turbidity mitigation in an oil sands end pit lake through ph reduction and fresh water addition.
- Brown LD, Pérez-Estrada L, Wang N, El-Din MG, Martin JW, Fedorak PM, Ulrich AC. 2013. Indigenous microbes survive in situ ozonation improving biodegradation of dissolved organic matter in aged oil sands process-affected waters. Chemosphere. 93(11):2748-2755.
- Brown LD, Ulrich AC. 2015. Oil sands naphthenic acids: A review of properties, measurement, and treatment. Chemosphere. 127:276-290.
- Burnham, J.C., Hatfield, N., Bennett, G.F. and Logan, T.J., 1992. Use of kiln dust with quicklime for effective municipal sludge pasteurization and stabilization with the N-Viro Soil process. Innovations and uses for lime, pp.128-141.
- Caughill DL, Morgenstern NR, Scott JD. 1993. Geotechnics of nonsegregating oil sand tailings. Canadian Geotechnical Journal. 30(5):801-811.
- Chalaturnyk RJ, Don Scott J, Özüm B. 2002. Management of oil sands tailings. Petroleum Science and Technology. 20(9-10):1025-1046.
- Chen M, Walshe G, Chi Fru E, Ciborowski JJH, Weisener CG. 2013a. Microcosm assessment of the biogeochemical development of sulfur and oxygen in oil sands fluid fine tailings. Applied Geochemistry. 37:1-11.
- Chen M, Walshe G, Fru EC, Ciborowski JJH, Weisener CG. 2013b. Microcosm assessment of the biogeochemical development of sulfur and oxygen in oil sands fluid fine tailings. Applied geochemistry. 37:1-11.

- Chen Z, Zhang W, Wang D, Ma T, Bai R. 2015. Enhancement of activated sludge dewatering performance by combined composite enzymatic lysis and chemical re-flocculation with inorganic coagulants: Kinetics of enzymatic reaction and re-flocculation morphology. Water Research. 83:367-376.
- Choi J, Hwang G, El-Din MG, Liu Y. 2014. Effect of reactor configuration and microbial characteristics on biofilm reactors for oil sands process-affected water treatment. International Biodeterioration & Biodegradation. 89:74-81.
- Choudhary M, Ray MB, Neogi S. 2019. Evaluation of the potential application of cactus (opuntia ficus-indica) as a bio-coagulant for pre-treatment of oil sands process-affected water. Separation and Purification Technology. 209:714-724.
- Clark KA, Pasternack DS. 1932. Hot water seperation of bitumen from alberta bituminous sand. Industrial & Engineering Chemistry. 24(12):1410-1416.
- Clemente JS, Fedorak PM. 2005. A review of the occurrence, analyses, toxicity, and biodegradation of naphthenic acids. Chemosphere. 60(5):585-600.
- Cossey HL, Batycky AE, Kaminsky H, Ulrich AC. 2021. Geochemical stability of oil sands tailings in mine closure landforms. Minerals. 11(8).
- Crittenden JC, Trussell RR. 2005. Hand, dw; howe, kj: Tchobanoglous, g. Water treatment principles and design john wiley & sons. Inc. Design concepts for thin layered freeze-thaw dewatering systems. 1993.
- Christy, R.W., 1990. Sludge disposal using lime. Water Environment & Technology, 2(4), pp.56-61.

- Dawson RF, Sego DC, Pollock GW. 1999. Freeze-thaw dewatering of oil sands fine tails. Canadian geotechnical journal. 36(4):587-598.
- Derjaguin, B., Landau, L., 1993. Theory of the stability of strongly charged lyophobic sols and of the adhesion of strongly charged-particles in solutions of electrolytes. Prog. Surf. Sci. 43 (1-4), 30-59.
- Dompierre KA, Barbour SL. 2016a. Characterization of physical mass transport through oil sands fluid fine tailings in an end pit lake: A multi-tracer study. Journal of Contaminant Hydrology. 189:12-26.
- Dompierre KA, Barbour SL. 2016b. Thermal properties of oil sands fluid fine tailings: Laboratory and in situ testing methods. Canadian Geotechnical Journal. 54(3):428-440.
- Dompierre KA, Lindsay MBJ, Cruz-Hernández P, Halferdahl GM. 2016a. Initial geochemical characteristics of fluid fine tailings in an oil sands end pit lake. Science of The Total Environment. 556(Supplement C):196-206.
- Dompierre KA, Lindsay MBJ, Cruz-Hernández P, Halferdahl GM. 2016b. Initial geochemical characteristics of fluid fine tailings in an oil sands end pit lake. Science of The Total Environment. 556:196-206.
- Dursun D, Turkmen M, Abu-Orf M, Dentel SK. 2006. Enhanced sludge conditioning by enzyme pre-treatment: Comparison of laboratory and pilot scale dewatering results. Water science and technology. 54(5):33-41.
- El-Din MG, Fu H, Wang N, Chelme-Ayala P, Pérez-Estrada L, Drzewicz P, Martin JW, Zubot W, Smith DW. 2011. Naphthenic acids speciation and removal

during petroleum-coke adsorption and ozonation of oil sands process-affected water. Science of the Total Environment. 409(23):5119-5125.

- ERCB. 2012. Alberta's energy reserves 2011 and supply/demand outlook 2012-2021. Government of alberta, c., ab, canada, 2012 (ed).
- Ewin GJ, Erno BP, Hepler LG. 1981. Clay chemistry: Investigation of thermodynamics of ion exchange reactions by titration calorimetry. Canadian Journal of Chemistry. 59(20):2927-2933.
- Farkish A. 2013. Sap based rapid dewatering of oil sands mature fine tailings. Ma sc thesis. Citeseer.
- Farkish A, Fall M. 2013. Rapid dewatering of oil sand mature fine tailings using super absorbent polymer (sap). Minerals Engineering. 50-51(Supplement C):38-47.
- Farkish A, Fall M. 2014. Consolidation and hydraulic conductivity of oil sand mature fine tailings dewatered by using super absorbent polymer. Journal of Geotechnical and Geoenvironmental Engineering. 140(7):06014006.
- Fedorak PM, Coy DL, Dudas MJ, Simpson MJ, Renneberg AJ, MacKinnon MD. 2003. Microbially-mediated fugitive gas production from oil sands tailings and increased tailings densification rates. Journal of Environmental Engineering and Science. 2(3):199-211.
- Foght JM, Gieg LM, Siddique T. 2017. The microbiology of oil sands tailings: Past, present, future. Fems Microbiology Ecology. 93(5):22.

- Frank RA, Kavanagh R, Burnison BK, Arsenault G, Headley JV, Peru KM, Van Der Kraak G, Solomon KR. 2008. Toxicity assessment of collected fractions from an extracted naphthenic acid mixture. Chemosphere. 72(9):1309-1314.
- Gosselin P, Hrudey SE, Naeth MA, Plourde A, Therrien R, Van Der Kraak G, Xu Z. 2010. Environmental and health impacts of canada's oil sands industry. Royal Society of Canada, Ottawa, ON. 10.
- Government of Alberta. 2015. Lower athabasca region : Tailings management framework for the mineable athabasca oil sands.
- Government of Alberta. 2019. Electronic resource about oil sands. [www document], url https://www.Alberta.Ca/about-oil-sands.Aspx. Bioresource Technology Reports. 11:100547.
- Grasso D, Subramaniam K, Butkus M, Strevett K, Bergendahl J. 2002. A review of non-dlvo interactions in environmental colloidal systems. Reviews in Environmental Science and Biotechnology. 1(1):17-38.
- Grewer DM, Young RF, Whittal RM, Fedorak PM. 2010. Naphthenic acids and other acid-extractables in water samples from alberta: What is being measured? Science of the Total Environment. 408(23):5997-6010.

Gulley JR. 1992. Oil sands tailings sludge abandonment dry land reclamation study.

Gumfekar SP, Rooney TR, Hutchinson RA, Soares JBP. 2017. Dewatering oil sands tailings with degradable polymer flocculants. Acs Applied Materials & Interfaces. 9(41):36290-36300.

- Gumfekar SP, Vajihinejad V, Soares JBP. 2019. Advanced polymer flocculants for solid–liquid separation in oil sands tailings. Macromolecular Rapid Communications. 40(1):1800644.
- Guo C. 2009. Rapid densification of the oil sands mature fine tailings (mft) by microbial activity.
- Hamza HA, Stanonik DJ, Kessick MA. 1996. Flocculation of lime-treated oil sands tailings. Fuel. 75(3):280-284.
- Hardy WB. 1900. A preliminary investigation of the conditions which determine the stability of irreversible hydrosols. Proceedings of the Royal Society of London. 66(424-433):110-125.
- Haveroen ME, MacKinnon MD, Fedorak PM. 2005. Polyacrylamide added as a nitrogen source stimulates methanogenesis in consortia from various wastewaters. Water Research. 39(14):3333-3341.
- Headley JV, McMartin DW. 2004. A review of the occurrence and fate of naphthenic acids in aquatic environments. Journal of Environmental Science and Health, Part A. 39(8):1989-2010.
- Herman DC, Fedorak PM, MacKinnon MD, Costerton JW. 1994. Biodegradation of naphthenic acids by microbial populations indigenous to oil sands tailings. Canadian journal of microbiology. 40(6):467-477.
- Holowenko FM, MacKinnon MD, Fedorak PM. 2000. Methanogens and sulfatereducing bacteria in oil sands fine tailings waste. Canadian Journal of Microbiology. 46(10):927-937.

- Holowenko FM, MacKinnon MD, Fedorak PM. 2002. Characterization of naphthenic acids in oil sands wastewaters by gas chromatography-mass spectrometry. Water research. 36(11):2843-2855.
- Huc AY. 2011. Heavy crude oils: From geology to upgrading (introduction and synopsis. Editions Technip, April.27-34.
- Hunter RJ. 2001. Foundations of colloid science. Oxford ; New York: Oxford University Press.
- Jeeravipoolvarn S. 2010. Geotechnical behavior of in-line thickened oil sands tailings. University of Alberta Edmonton.
- Jeeravipoolvarn S, Scott JD, Chalaturnyk RJ. 2009. 10 m standpipe tests on oil sands tailings: Long-term experimental results and prediction. Canadian Geotechnical Journal. 46(8):875-888.
- Ji Y, Lu Q, Liu Q, Zeng H. 2013. Effect of solution salinity on settling of mineral tailings by polymer flocculants. Colloids and Surfaces A: Physicochemical and Engineering Aspects. 430:29-38.
- Johnson RL, Bork P, Allen EA, James WH, Koverny L. 1993. Oil sands sludge dewatering by freeze-thaw and evapotranspiration.
- Kaminsky HAW, Etsell TH, Ivey DG, Omotoso O. 2009. Distribution of clay minerals in the process streams produced by the extraction of bitumen from athabasca oil sands. Can J Chem Eng. 87(1):85-93.

- Kannel PR, Gan TY. 2012. Naphthenic acids degradation and toxicity mitigation in tailings wastewater systems and aquatic environments: A review. Journal of Environmental Science and Health, Part A. 47(1):1-21.
- Kavanagh RJ, Frank RA, Oakes KD, Servos MR, Young RF, Fedorak PM, MacKinnon MD, Solomon KR, Dixon DG, Van Der Kraak G. 2011. Fathead minnow (pimephales promelas) reproduction is impaired in aged oil sands process-affected waters. Aquatic Toxicology. 101(1):214-220.
- Kong JD, Wang H, Siddique T, Foght J, Semple K, Burkus Z, Lewis MA. 2019. Second-generation stoichiometric mathematical model to predict methane emissions from oil sands tailings. Science of The Total Environment. 694:133645.
- Leahy JG, Colwell RR. 1990. Microbial degradation of hydrocarbons in the environment. Microbiological reviews. 54(3):305-315.
- Leishman C, Widdup EE, Quesnel DM, Chua G, Gieg LM, Samuel MA, Muench DG.
 2013. The effect of oil sands process-affected water and naphthenic acids on the germination and development of arabidopsis. Chemosphere. 93(2):380-387.
- Li C, Fu L, Stafford J, Belosevic M, El-Din MG. 2017. The toxicity of oil sands process-affected water (ospw): A critical review. Science of the Total Environment. 601:1785-1802.

- Li, Q., Jonas, U., Zhao, X.S., Kapp, M., 2008. The forces at work in collodial selfassembly: a review on fundamental interactions between collodial particels. Asia Pac. J. Chem. Eng. 3 (3), 255-268.
- Lin F, Zhu X, Luo Y, Yu P, Liu M. 2019. Improvement of activated sludge dewatering properties using green conditioners: Chitosan hydrochloride and lysozyme. RSC advances. 9(12):6936-6945.
- Liu LY, Hu Y, Min FF, Zhang MX, Song SX. 2013. Characterizations and stability of colloidal coal-measure kaolinite in aqueous suspensions: A review. Surf Rev Lett. 20(1).
- Long J, Li H, Xu Z, Masliyah JH. 2006. Role of colloidal interactions in oil sand tailings treatment. AIChE journal. 52(1):371-383.
- Lorentz J, Bara B, Mikula R, Leneve E, Wang N. 2014. Oil sands fluid fine tailings dewatering using rim ditching. Google Patents.
- Lu J, Rao S, Le T, Mora S, Banerjee S. 2011. Increasing cake solids of cellulosic sludge through enzyme-assisted dewatering. Process biochemistry. 46(1):353-357.
- Lu Q. 2016. Understanding two-step polymer flocculation of oil sands tailings. [2016].
- Lu QY, Yan B, Xie L, Huang J, Liu Y, Zeng HB. 2016. A two-step flocculation process on oil sands tailings treatment using oppositely charged polymer flocculants. Science of the Total Environment. 565:369-375.
- MacKinnon M. 2004. Oil sands water quality issues: Properties, treatment and discharge options.

- Mackinnon M, Zubot W. 2013. Chapter 10: Water management. Kingsley Knowledge Publishing: Alberta.
- MacKinnon MD. 1989. Development of the tailings pond at syncrude's oil sands plant: 1978–1987. AOSTRA J Res. 5(2):109-133.
- MacKinnon MD, Boerger H. 1986. Description of two treatment methods for detoxifying oil sands tailings pond water. Water Quality Research Journal. 21(4):496-512.
- MacKinnon MD, Matthews JG, Shaw WH, Cuddy RG. 2000. Water quality issues associated with implementation of composite tailings (ct) technology for managing oil sands tailings.
- MacKinnon MD, Matthews JG, Shaw WH, Cuddy RG. 2001. Water quality issues associated with composite tailings (ct) technology for managing oil sands tailings. International Journal of Surface Mining, Reclamation and Environment. 15(4):235-256.
- Madill REA, Orzechowski MT, Chen G, Brownlee BG, Bunce NJ. 2001. Preliminary risk assessment of the wet landscape option for reclamation of oil sands mine tailings: Bioassays with mature fine tailings pore water. Environmental Toxicology: An International Journal. 16(3):197-208.
- Mahdavi H, Prasad V, Liu Y, Ulrich AC. 2015. In situ biodegradation of naphthenic acids in oil sands tailings pond water using indigenous algae-bacteria consortium. Bioresource technology. 187:97-105.

- Mahdavi H, Ulrich AC, Liu Y. 2012. Metal removal from oil sands tailings pond water by indigenous micro-alga. Chemosphere. 89(3):350-354.
- Mamer M. 2010. Oil sands tailings technology: Understanding the impact to reclamation.
- Masliyah JH, Bhattacharjee S. 2006. Electrokinetic and colloid transport phenomena. Hoboken, N.J.: Wiley-Interscience.
- Matthews JG, Shaw WH, MacKinnon MD, Cuddy RG. 2002. Development of composite tailings technology at syncrude. International Journal of Surface Mining, Reclamation and Environment. 16(1):24-39.
- McQueen AD, Kinley CM, Hendrikse M, Gaspari DP, Calomeni AJ, Iwinski KJ, Castle JW, Haakensen MC, Peru KM, Headley JV. 2017a. A risk-based approach for identifying constituents of concern in oil sands process-affected water from the athabasca oil sands region. Chemosphere. 173:340-350.
- McQueen AD, Kinley CM, Hendrikse M, Gaspari DP, Calomeni AJ, Iwinski KJ, Castle JW, Haakensen MC, Peru KM, Headley JV et al. 2017b. A risk-based approach for identifying constituents of concern in oil sands process-affected water from the athabasca oil sands region. Chemosphere. 173:340-350.
- Meshref MNA, Klamerth N, Islam MS, McPhedran KN, Gamal El-Din M. 2017. Understanding the similarities and differences between ozone and peroxone in the degradation of naphthenic acids: Comparative performance for potential treatment. Chemosphere. 180:149-159.

- . Centrifuge options for production of dry stackable tailings' in surface mined oil sands tailings management. 2008 2008: OnePetro.
- Missana T, Adell A. 2000. On the applicability of dlvo theory to the prediction of clay colloids stability. J Colloid Interf Sci. 230(1):150-156.
- Mohamed MH, Wilson LD, Headley JV, Peru KM. 2011. Sequestration of naphthenic acids from aqueous solution using β-cyclodextrin-based polyurethanes.
 Physical Chemistry Chemical Physics. 13(3):1112-1122.
- Morandi GD, Wiseman SB, Guan M, Zhang XW, Martin JW, Giesy JP. 2017. Elucidating mechanisms of toxic action of dissolved organic chemicals in oil sands process-affected water (ospw). Chemosphere. 186:893-900.
- Nelson LR, MacKinnon M, Gulley JR. 1993. Application of toxicity testing in the evaluation of reclamation options for oil sands fine tails.
- Nix PG. 1983. Detoxification of tailings pond top water. Suncor Incorporated.
- Nix PG, Martin RW. 1992. Detoxification and reclamation of suncor's oil sand tailings ponds. Environmental Toxicology and Water Quality. 7(2):171-188.
- O'Shea J-P, Qiao GG, Franks GV. 2011. Temperature responsive flocculation and solid–liquid separations with charged random copolymers of poly(n-isopropyl acrylamide). Journal of Colloid and Interface Science. 360(1):61-70.
- Penner TJ, Foght JM. 2010. Mature fine tailings from oil sands processing harbour diverse methanogenic communities. Canadian journal of microbiology. 56(6):459-470.

- Pourrezaei P. 2013. Physico-chemical processes for oil sands process-affected water treatment.
- Pourrezaei P, Drzewicz P, Wang Y, Gamal El-Din M, Perez-Estrada LA, Martin JW, Anderson J, Wiseman S, Liber K, Giesy JP. 2011. The impact of metallic coagulants on the removal of organic compounds from oil sands processaffected water. Environmental Science & Technology. 45(19):8452-8459.
- Preez ACd, S Preston J. 1992. The solvent extraction of rare-earth metals by carboxylic acids. Solvent Extraction and Ion Exchange. 10(2):207-230.
- Proskin S, Sego D, Alostaz M. 2010. Freeze-thaw and consolidation tests on suncor mature fine tailings (mft). Cold Regions Science and Technology. 63(3):110-120.
- Ramos-Padrón E, Bordenave S, Lin S, Bhaskar IM, Dong X, Sensen CW, Fournier J, Voordouw G, Gieg LM. 2010. Carbon and sulfur cycling by microbial communities in a gypsum-treated oil sands tailings pond. Environmental science & technology. 45(2):439-446.
- Redfield E, Croser C, Zwiazek JJ, MacKinnon MD, Qualizza C. 2003. Responses of red-osier dogwood to oil sands tailings treated with gypsum or alum. Journal of environmental quality. 32(3):1008-1014.
- Reid ML, Warren LA. 2016. S reactivity of an oil sands composite tailings deposit undergoing reclamation wetland construction. Journal of environmental management. 166:321-329.

- Rogers VV, Liber K, MacKinnon MD. 2002. Isolation and characterization of naphthenic acids from athabasca oil sands tailings pond water. Chemosphere. 48(5):519-527.
- Romaniuk N, Mehranfar M, Selani H, Ozum B, Tate MJ, Scott JD. 2015. Residual bitumen recovery from oil sand tailings with lime. Proceedings tailings and mine waste 2015, vancouver, bc, october 26 to 28, 2015.
- Sabyasachi M, Venkatesh M, Ajay K D, Dena W M, John V H, Kerry M P. 2010. Photocatalysis of naphthenic acids in water. Journal of Water Resource and Protection. 2010.
- Samadi N. 2019. Partitioning of inorganic contaminants between fluid fine tailings and cap water under end pit lake scenario: Biological, chemical and mineralogical processes.
- Sawyer CN, McCarty PL, Parkin GF, Sawyer CN. 1994. Chemistry for environmental engineering. New York: McGraw-Hill.
- Schofield RK, Samson HR. 1954. Flocculation of kaolinite due to the attraction of oppositely charged crystal faces. Discussions of the Faraday Society. 18:135-145.
- Schulze H. 1882. Schwefelarsen in wässriger lösung. Journal für praktische Chemie. 25(1):431-452.
- Scott AC, Mackinnon MD, Fedorak PM. 2005. Naphthenic acids in athabasca oil sands tailings waters are less biodegradable than commercial naphthenic acids. Environmental science & technology. 39(21):8388-8394.

- Seyedy Niasar H. 2017. Treatment of oil sands process-affected water using activated and surface modified petroleum coke for organic compounds recovery.
- Shahimin MFM, Foght JM, Siddique T. 2016. Preferential methanogenic biodegradation of short-chain n-alkanes by microbial communities from two different oil sands tailings ponds. Science of the Total Environment. 553:250-257.
- Shahimin MFM, Siddique T. 2017. Methanogenic biodegradation of paraffinic solvent hydrocarbons in two different oil sands tailings. Science of the Total Environment. 583:115-122.
- Shen W, Li Z, Liu Y. 2008. Surface chemical functional groups modification of porous carbon. Recent Patents on Chemical Engineering. 1(1):27-40.
- Siddique T, Fedorak PM, Foght JM. 2006. Biodegradation of short-chain n-alkanes in oil sands tailings under methanogenic conditions. Environmental science & technology. 40(17):5459-5464.
- Siddique T, Fedorak PM, MacKinnon MD, Foght JM. 2007. Metabolism of btex and naphtha compounds to methane in oil sands tailings. Environmental science & technology. 41(7):2350-2356.
- Siddique T, Gupta R, Fedorak PM, MacKinnon MD, Foght JM. 2008. A first approximation kinetic model to predict methane generation from an oil sands tailings settling basin. Chemosphere. 72(10):1573-1580.
- Siddique T, Kuznetsov P, Kuznetsova A, Arkell N, Young R, Li C, Guigard S, Underwood E, Foght JM. 2014a. Microbially-accelerated consolidation of oil

sands tailings. Pathway i: Changes in porewater chemistry. Front Microbiol. 5:106.

- Siddique T, Kuznetsov P, Kuznetsova A, Arkell N, Young R, Li C, Guigard SE, Underwood E, Foght JM. 2014b. Microbially-accelerated consolidation of oil sands tailings. Pathway i: Changes in porewater chemistry. Frontiers in microbiology. 5:106.
- Siddique T, Kuznetsov P, Kuznetsova A, Li C, Young R, Arocena JM, Foght J. 2014c. Microbially-accelerated consolidation of oil sands tailings. Pathway ii: Solid phase biogeochemistry. Frontiers in Microbiology. 5:107.
- Siddique T, Kuznetsova A. 2020. Linking hydrocarbon biodegradation to greenhouse gas emissions from oil sands tailings and its impact on tailings management. Canadian Journal of Soil Science.1-9.
- Siddique T, Mohamad Shahimin MF, Zamir S, Semple K, Li C, Foght JM. 2015. Long-term incubation reveals methanogenic biodegradation of c5 and c6 isoalkanes in oil sands tailings. Environmental science & technology. 49(24):14732-14739.
- Siddique T, Penner T, Klassen J, Nesbø C, Foght JM. 2012. Microbial communities involved in methane production from hydrocarbons in oil sands tailings. Environmental science & technology. 46(17):9802-9810.
- Siddique T, Penner T, Semple K, Foght JM. 2011. Anaerobic biodegradation of longer-chain n-alkanes coupled to methane production in oil sands tailings. Environmental Science & Technology. 45(13):5892-5899.

- Siddique T, Semple K, Li C, Foght JM. 2020. Methanogenic biodegradation of isoalkanes and cycloalkanes during long-term incubation with oil sands tailings. Environmental Pollution. 258:113768.
- Siddique T, Stasik S, Mohamad Shahimin MF, Wendt-Potthoff K. 2018. Microbial communities in oil sands tailings: Their implications in biogeochemical processes and tailings management. Microbial communities utilizing hydrocarbons and lipids: members, metagenomics and ecophysiology Handbook of hydrocarbon and lipid microbiology Springer Nature, Basel, Switzerland.1-33.
- Small CC, Cho S, Hashisho Z, Ulrich AC. 2015. Emissions from oil sands tailings ponds: Review of tailings pond parameters and emission estimates. Journal of Petroleum Science and Engineering. 127:490-501.
- Small CC, Ulrich AC, Hashisho Z. 2012. Adsorption of acid extractable oil sands tailings organics onto raw and activated oil sands coke. Journal of Environmental Engineering. 138(8):833-840.
- Staples TL, Chatterjee PK. 2002. Synthetic superabsorbents. Textile science and technology. Elsevier. p. 283-322.
- Stasik S, Loick N, Knöller K, Weisener C, Wendt-Potthoff K. 2014. Understanding biogeochemical gradients of sulfur, iron and carbon in an oil sands tailings pond. Chemical Geology. 382:44-53.
- Stasik S, Wendt-Potthoff K. 2014. Interaction of microbial sulphate reduction and methanogenesis in oil sands tailings ponds. Chemosphere. 103:59-66.

- Steffen F, Janzon R, Saake B. 2018. Enzymatic treatment of deinking sludge–effect on fibre and drainage properties. Environmental technology. 39(21):2810-2821.
- Strosher MT, Peake E. 1978. Characterization of organic constituents in waters and wastewaters of the athabasca oil sands mining area.
- Sworska A, Laskowski JS, Cymerman G. 2000. Flocculation of the syncrude fine tailings: Part i. Effect of ph, polymer dosage and mg2+ and ca2+ cations. International journal of mineral processing. 60(2):143-152.
- Tate M, Leikam J, Fox J, Romaniuk N. 2017. Use of calcium hydroxide as a coagulant to improve oil sands tailings treatment. Proceedings tailings and mine waste 2017, banff, alberta, canada, november 2017.
- Tate M, Leikam J, ScotT JD, Mehranfar M, Romaniuk N, Ozum B. 2016. Impacts of calcium compounds on oil sands water chemistry. Conference presentation at the mining and tailings waste conference, keystone, colorado, usa.
- Thomas L, Jungschaffer G, Sprössler B. 1993. Improved sludge dewatering by enzymatic treatment. Water Science and Technology. 28(1):189-192.
- Vajihinejad V, Gumfekar SP, Dixon DV, Silva MA, Soares JBP. 2021. Enhanced dewatering of oil sands tailings by a novel water-soluble cationic polymer. Separation and Purification Technology. 260:118183.
- Vedoy DRL, Soares JBP. 2015. Water-soluble polymers for oil sands tailing treatment: A review. The Canadian Journal of Chemical Engineering. 93(5):888-904.

- Wagner CD, Passoja DE, Hillery HF, Kinisky TG, Six HA, Jansen WT, Taylor JA. 1982. Auger and photoelectron line energy relationships in aluminum–oxygen and silicon–oxygen compounds. Journal of Vacuum Science and Technology. 21(4):933-944.
- Wang C, Alpatova A, McPhedran KN, Gamal El-Din M. 2015. Coagulation/flocculation process with polyaluminum chloride for the remediation of oil sands process-affected water: Performance and mechanism study. Journal of Environmental Management. 160(Supplement C):254-262.
- Wang C, Harbottle D, Liu Q, Xu Z. 2014. Current state of fine mineral tailings treatment: A critical review on theory and practice. Minerals Engineering. 58:113-131..
- Wells, P. S., and D. A. Riley, 2007. MFT drying: case study in the use of rheological modification and dewatering of fine tailings through thin lift deposition in the oil sands of alberta..
- Westcott F, Watson L. 2005. End pit lakes technical guidance document. Clearwater environmental consultants for CEMA end Pit Lakes subgroup, Project. 61.
- White J, Stewart CA, Wareham DG, Wilson TM. 2011. Treatment of volcanic ashcontaminated surface waters through the optimisation of physical and chemical processes. GNS Science.
- Wu B, Chai X, Zhao Y. 2016. Enhanced dewatering of waste-activated sludge by composite hydrolysis enzymes. Bioprocess and biosystems engineering. 39(4):627-639.

- Wu C, Zhang K, Chen J, Li S. 2014. Effect of cellulase and protease pretreatment on dewaterability of waste activated sludge from paper mill. BioResources. 9(3):5198-5207.
- Wu S, Wang B, Sego D, Naeth A. 2011. Responses of wheatgrass species to composite/consolidated oil sands tailings.
- Yang H, Fan M, Liu A, Dong L. 2015. General formulas for drag coefficient and settling velocity of sphere based on theoretical law. International Journal of Mining Science and Technology. 25(2):219-223.
- Young RF, Li C, Semple KM, Baker MG, Kuznetsov P, Kuznetsova A, Siddique T, Foght JM. 2020. Accelerated consolidation of oil sands tailings using an anaerobic bioreactor. Bioresource Technology Reports. 11:100547.
- Yu X, Cao Y, Sampaga R, Rybiak S, Burns T, Ulrich AC. 2018. Accelerated dewatering and detoxification of oil sands tailings using a biological amendment. Journal of Environmental Engineering. 144(9):04018091.
- Yuan X. 2009. Thickened tailings (paste) technology and its applicability in oilsand tailings management. Tailings and Mine Waste 2009, Nov 1-4, Banf, Alberta.
- Zhu R, Liu QX, Xu ZH, Masliyah JH, Khan A. 2011. Role of dissolving carbon dioxide in densification of oil sands tailings. Energy & Fuels. 25(5):2049-2057.
- Zhu S, Li MY, El -Din MG. 2017. Forward osmosis as an approach to manage oil sands tailings water and on-site basal depressurization water. Journal of Hazardous Materials. 327:18-27.

- Zubot W, An Z, Benally C, El-Din MG. 2021. Treatment of oil sands process water using petroleum coke: Field pilot. Journal of Environmental Management. 289:112407.
- Zubot W, MacKinnon MD, Chelme-Ayala P, Smith DW, El-Din MG. 2012. Petroleum coke adsorption as a water management option for oil sands process-affected water. Sci Total Environ. 427:364-372.

3 IMPACT OF LIME TREATMENT ON TAILINGS DEWATERING AND CAP WATER QUALITY UNDER AN OIL SANDS END PIT LAKE SCENARIO²

3.1 Introduction

Large volumes of fluid fine tailings (FFT) and oil sands process-affected water (OSPW) are generated from surface mining of oil sands in northern Alberta. The continuous storage of the massive quantities of FFT in engineered tailing ponds (Allen, 2008b; Clemente and Fedorak, 2005; Siddique et al., 2014) requires oil sands operators to use various reclamation strategies to achieve reclamation obligations. Several remediation approaches (e.g., composite tailings, atmospheric drying, etc.) have been studied, however these methods have drawbacks that hinder their applications. Most of the proposed alternatives are either energy-consuming and costly (e.g., centrifugation; (Wang et al., 2014) or need a large amount of land (e.g., drying requiring freeze-thaw cycles (Proskin et al., 2012) (Allen, 2008a; Board, 2012; Hyndman and Sobkowicz, 2010; Vajihinejad and Soares, 2018; Wells and Riley, 2007). Thus, more economical and effective solutions are being investigated to incorporate FFT into oil sands mine closure landscapes (Famakinwa et al., 2018). Water-capped tailings technologies such as end pit lakes (EPLs) are one of those

² A version of this chapter has been published as: Allam, N.E.; Romaniuk, N.; Tate, M.; Meshref, M.N.A.; Dhar, B.R.; Ulrich, A. (2021): "Impact of lime treatment on tailings dewatering and cap water quality under an oil sands end pit lake scenario", Science of the Total Environment Journal, 781, 146699.

potential approaches to reduce FFT inventories in tailings ponds for subsequent reclamation (Kabwe et al., 2018; Zubot, 2010).

EPLs are engineered water bodies created in the post-mining pit by pumping freshwater and OSPW on top of the FFT (Zubot, 2010). EPLs are predicted to slowly develop into self-sustaining aquatic ecosystems (Hrynyshyn, 2012; Zubot, 2010) and act as permanent features in the final reclaimed landscape (Kabwe et al., 2018). In the first field scale demonstration of pit lakes, Base Mine Lake (BML), the FFT was not treated before being deposited into the lake, and the FFT is settling by self-weight consolidation (natural dewatering through gravity settling; (Dompierre, 2016). The FFT (with initial solids content around 10% solids (wt./wt.) as first deposited into tailing ponds) comprises fine clays suspended in water, and the clay particles slowly settle over decades (MacKinnon, 1989).

There are key concerns and potential challenges relevant to the environmental development of EPLs (COSIA, 2012). For instance, during the slow settling of FFT, the released pore water contributes OSPW-derived constituents to the EPL cap water (Dompierre et al., 2016a), including the chemical flux of constituents of potential concern (COPCs) (Dompierre and Barbour, 2016; Lawrence et al., 2015). The presence of COPCs (e.g., major inorganic salts, dissolved organic compounds [DOC], naphthenic acids [NAs], and petroleum hydrocarbons [PHCs]) are potentially toxic to aquatic microorganisms (Dompierre et al., 2016a; McQueen et al., 2017; Samadi, 2019; White, 2017; White and Liber, 2018).

Earlier studies investigated the influence of vertical mixing (Lawrence et al., 2015) and have used a numerical modeling approach to simulate the movement of COPCs (Dompierre and Barbour, 2016) and identify the key mechanisms controlling mass transport in FFT layers within EPLs. Similarly, White and Liber (2018) monitored the concentrations of inorganic salts in the cap water of BML over three consecutive years and assessed the corresponding toxicological risk of elevated concentrations of inorganic constituents such as sodium (Na⁺) and bicarbonate (HCO₃⁻) on EPL cap water toxicity (White, 2017; White and Liber, 2018; 2020). The studies reported a sufficient level of Daphnia's populations within the BML surface water with a promising improvement trend of BML surface water quality over time (White and Liber, 2018; 2020). However, this was constrained with the less sensitive toxicity endpoints samples used in the study (White and Liber, 2018) that could necessitate further investigations to claim the development into a robust EPL aquatic ecosystem. Several research groups have looked at the ability of the indigenous microbial community in BML to remediate COPCs (Bowman et al., 2016; Bradford et al., 2017; Richardson et al., 2020; Yu et al., 2019). While investigating methanogenic activity, Samadi et al. (2019) reported an enhancement in FFT dewatering with the addition of hydrocarbons and nutrients due to biogenic gas production. Also, Poon et al. (2018) reported a decrease in the turbidity of the EPL cap water through pH reduction by inducing CO₂. However, that study was focused on improving cap water clarity by enhancing the clay flocculation in BML.

Because FFT takes a long time (decades) to settle (Masliyah et al., 2011), coagulation and flocculation were implemented to enhance the dewatering process

(Famakinwa et al., 2018; Wang et al., 2015; Wang et al., 2014). Promising results in the acceleration of tailings dewatering were reported using lime or calcium hydroxide (Ca[OH]₂) as a stand-alone coagulant or in combination with gypsum (CaSO₄·2H₂O) or polymers (Baillie and Malmberg, 1969; Beaty and Lane, 1983; Caughill et al., 1993; Chalaturnyk et al., 2002; Ewin et al., 1981; Hamza et al., 1996; Lorentz et al., 2014; Matthews et al., 2002). Lime addition caused an increase in pH and provided divalent calcium ions (Ca²⁺) that improved FFT coagulation (Chalaturnyk et al., 2002; Tate et al., 2017). To date, chemical treatment, particularly lime addition, has not been thoroughly and comprehensively explored in EPL scenarios. Moreover, the effect of the slow dewatering of FFT on the development of EPLs is still unclear (Risacher et al., 2018) and should be investigated further.

Based on previous studies, lime addition can be hypothesized to accelerate the dewatering of FFT through cation exchange. The increase in pH resulting from lime addition can reduce methanogenesis in the FFT layer, and therefore reduce the chemical flux of ions and organic constituents to the water cap and enhance the cap water quality in EPLs.

Consequently, the key goal of this study is to examine the impact of the addition of various lime doses on FFT dewatering and cap water quality in EPL configurations. The specific objectives are as follows: a) to investigate the feasibility of lime treatment to enhance FFT dewatering in EPLs; b) to evaluate the influence of adding different dosages of lime on the quality of cap water and FFT pore water in EPLs; c) to elucidate the role of cation exchange on the reduction and removal of organic fractions in FFT (DOC and PHCs); and d) to examine the variation of water

chemistry (alkalinity, pH, and cations), bacterial cell counts, and toxicity toward *Vibrio fischeri*.

3.2 Materials and methods

3.2.1 Materials

FFT was collected from an active oil sands tailings pond in northern Alberta. Hydrated lime (Ca[OH]₂) was provided by Graymont Inc. (Calgary, Alberta, Canada). FFT and lime were stored in sealed buckets at 4 °C until use. Raw synthetic OSPW was prepared in 15 L deionized water with: 10.80 g NaHCO₃, 4.35 g Na₂SO₄, 8.70 g NaCl, 0.75 g KCl, 2.10 g MgCl₂·6H₂O, and 0.75 g CaCl₂·2H₂O. The specifications for Raw synthetic OSPW were provided by Graymont Inc., and mimics the solute concentrations in Albian OSPW (Mahdavi, 2014; Poon, 2019). The Raw synthetic OSPW didn't contain any organics to better understand the movement of COPCs from the FFT layer to the cap water.

3.2.2 Experiment

Four different lime dosages (650, 1600, 3500, and 4000 ppm on a wet weight basis) were used. Lime slurry was prepared by adding hydrated lime to Millipore ultrapure water (18.2 M Ω cm) to achieve a 5% solids slurry (wt./wt.). Each lime dose was added to 1 kg FFT in a glass beaker and mixed with a mechanical agitator at moderate speed (1200 rpm). The initial solids content of the raw FFT was 32.8%. To achieve 50% solids content, both control (no lime) and lime-treated FFT were compressed in an air pressure filter (OFITE; Bench-Mount Filter Press with Hose and Regulator; #140-31) at 100 psi.

Experiments were conducted in 1 L glass columns, where 500 ml FFT (50% solids content [wt./wt.]) was placed at the bottom of the column and covered with 500 ml Raw synthetic OSPW. All experiments were conducted at ambient conditions (23°C) using uncovered columns in triplicate, except for the 4000 ppm treatment which had both covered and uncovered conditions. This step was conducted in order to determine the effect of atmospheric CO₂ on the cap water quality and toxicity and to highlight any discrepancies between covered and uncovered conditions. This can be also attributed to the plateau reached at pH 12.5 regardless the increase of lime dose (> 3500 ppm) that requires better understanding about the potential influence of any adverse impact of overdosing.

To better understand the effectiveness of lime addition on FFT dewatering, % water recovery (WR) was used as a comparable index in our assessment. The WR was calculated as follows:

$$WR\% = \frac{\text{Volume of pore water released at day 90}}{\text{total initial volume of FFT}} \times 100$$

The released water from FFT (FFT pore water) at day 0 was collected using the pressure filter, while FFT pore water at day 90 was collected from the FFT samples by centrifugation at 8,000 g for 40 min. FFT pore water was collected in 50 ml plastic tubes and stored at 4 °C before further analyses. For cap water analysis, samples were collected directly from the columns every two weeks. The initial FFT and final treated FFT samples at day 90 were collected directly from the columns, centrifuged at 8000g for 5 mins and stored at -20° C in 50 ml plastic vials for DNA extraction. For PHC analysis, FFT samples were stored at 4 °C until further analysis.

3.2.3 Chemical analysis

Water samples were analyzed for cations including Ca²⁺, Na⁺, K⁺, and Mg²⁺ using a Thermo iCAP 6000 series Inductively Coupled Plasma-Optical Emission Spectrometer (ICP-OES, ThermoFisher). A Mettler Toledo DL53 (Mississauga, Canada) was used to measure alkalinity with 0.02 N H₂SO₄ as a titrant, and pH values were measured using an Accumet® Research AR50 (Fisher Scientific, Lenexa, Kansas). DOC was measured using a TOC analyzer (Shimadzu TOC-LCPH, Japan) and a non-purgeable organic carbon method (Brown et al., 2013). Prior to any analysis and unless otherwise specified, liquid samples were filtered with polytetrafluoroethylene (PTFE) syringe filters (Fisher Scientific; 0.22 μm for cation analysis; 0.45 µm for DOC analysis). For DOC analysis, all water samples were diluted 10× using ultrapure water.

PHCs were analyzed based on the Canada-Wide Standard for Petroleum Hydrocarbons in soil (CCME, 2008). Fractions were grouped as follows: F1 (C6–C10), F2 (C10–C16), F3 (C16–C34), F4 (C34–C50), and F4G-SG (>C50; (CCME, 2008). According to the Dean Stark extraction method (the industry-accepted method) the F4 and F4G-SG fractions were categorized with respect to bitumen content (Dean and Stark, 1920). F1 fractions were non-detectable in all samples and were measured by purge and trap gas chromatography (Column: DB1 0.25 mm ID; 1.0 μm film; 60

m length). Other fractions (F2, F3, F4, and F4G-SG) were analyzed by Maxxam Analytics (Edmonton, Alberta, Canada). Further details on PHC fractions analysis can be found in the Canada-Wide Standard for Petroleum Hydrocarbons in Soil (CCME, 2008) and elsewhere (Yu et al., 2018a). To determine the statistical differences between the treatment conditions, p-values were calculated using a two tailed t-test in Microsoft Excel. The F-test was initially used to determine the variance among sample pools whether it is equal or unequal. The difference between the treatment conditions is considered significant if p-value < 0.05.

3.2.4 Toxicity Bioassay

The toxicity of the aqueous samples toward *V. fischeri* was measured with the Microtox[®] bioassay test (Osprey Scientific, Edmonton, Alberta, Canada). Briefly, the 81.9% Basic Test protocol was followed by 5 min and 15 min incubation in the Microtox[®] 500 Analyzer (Azur Environmental time, (Anderson et al., 2011). The light emission was measured and recorded with MicrotoxOmni software to determine the inhibitory concentration resulting in 20% less light emission (IC₂₀) or inhibitory concentration resulting in 50% less light emission (IC₅₀) values. There were no significant differences in the toxicity measured between 5 min and 15 min tests, therefore only the 5 min data is described (Miles et al., 2019). Toxicity units, derived from IC₅₀ (TU = 100 ÷ IC₅₀), were used to represent high-level toxicity trends (Yu et al., 2018a). In previous studies, 1.0 toxicity unit (TU) or lower is considered completely detoxified (Scott et al., 2008; Yu et al., 2018a).

Both principal component analysis (PCA) and cluster analysis were employed using Minitab 19 in order to evaluate and highlight any variations between the lime treated conditions and control. Implementing the PCA and cluster analysis in addition to Tukey Pairwise comparisons also supported the determination of clustering, similarity, and significant differences between the treatment conditions (i.e., p-values <0.05 is considered significant different).

3.2.5 DNA Extraction and qPCR Analysis

DNA was extracted from the FFT samples (up to 500 mg of FFT used per sample), using the FastDNATM SPIN Kit of Soil (MP Biomedicals) according to the isolation protocol provided by the supplier. The extraction was conducted at the start and end of the experiment to examine the initial and final microbial population density for all conditions. Microbial cell counts were determined by qPCR amplification of the RNA polymerase beta subunit (rpoB) gene, utilizing rpoB 1698f (5'-AACATCGGTTTGCTCAAC-3') 2041r (5'and rpoB CGTTGCATGTTGGTACCCAT-3') primers (Brown et al., 2013; Mahdavi et al., 2015; Yu et al., 2018b). Using a Bio-Rad CFX96 optical reaction module conversion for the C1000 Touch thermal cycler, the qPCR assay was performed on DNA extracted samples from the lime treated/ untreated tailings samples. The amplification data were analyzed using Bio-Rad CFX Manager[™] 3.0 software. All standards and samples were analyzed in triplicate. More details can be found elsewhere (Yu et al., 2018b).

3.3. Results and discussion

3.3.1 Influence of lime treatment on FFT dewatering

The effectiveness of lime addition on the FFT dewatering was assessed based on released water volumes, water recovery (WR), and solids content. The released pore water volumes were 21.7, 30.7, 37.7, and 30 ml for the 650, 1600, 3500, and 4000 ppm lime dosages, respectively. Therefore, an increase in the released volume of FFT pore water was observed after lime treatment.

To better compare our study with others, the WR was also used as a comparable index. In general, WR increased at day 90 for the lime-treated FFT samples, as illustrated in Table 3.1. At the 3500 ppm dose, the maximum WR was 7.9%. However, increasing the lime dose to 4000 ppm decreased the WR to 6.4%, implying that there is an adverse effect from lime overdosing. This can be attributed to the overdose effect (i.e., 4000 ppm), which might create an extra strength on the solids and resist the self-weight settling. Tate et al., (2017) reported the same notion that increasing lime dosages could lead to an increase in the yield stress of the FFT (i.e., developing strength by lime addition) and hence can enhance the geotechnical properties of FFT. Another explanation could be due to the premature cracking as a result of the reduction of solids percentage happened after lime overdosing (Tate et al., 2017).

Comparing our results with previous studies, the 3500 ppm lime treatment achieved a similar WR (7.9%) to Yu et al. (2018a), who reported 8% WR after 100 d using a biological amendment (Ultrazyme). In contrast, Siddique et al. (2014) reported

a higher WR of 26% after 90 d using hydrolyzed canola meal (i.e., as an organic substrate). For solids content in our study, the highest increase was also observed at a dose of 3500 ppm (the final solids content of the FFT increased to 51.6% at day 90; Table 3.1). Despite the increase in the solid content, it can be considered marginal. It is worth to note that the initial solids content had a significant impact on the final WR (or the final volume of released water) where the lowest initial solids content led to a high WR (Yu et al., 2018a). The initial solids content in our study was considerably higher (50%) than those used by Siddique et al. (2014); 25%) and Yu et al. (2018a); 34.6%).

Table 3.1. The FFT pore water recovery % (WR%) at the end of the experiment (day 90) as well as FFT solids content at day 90. The solids content for all columns at day 0 was 50%. All columns consisted of 500 ml lime treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 2 for 0 ppm and n = **3** for 650, 1600, and 3500 ppm) \pm one standard deviation.

Lime Dose (ppm)	WR%	Solid content% Day 90	Volume of FFT Porewater Released (ml)
0	$2.5\%\pm0.4\%$	$50.2\% \pm 0.04\%$	11.5 ± 2.1
650	4.7% ±0.5%	$50.7\% \pm 0.3\%$	21.7 ± 2.8
1600	6.3%± 0.1%	$50.9\% \pm 0.1\%$	30.7 ± 1.2
3500	7.9% ± 0.8%	$51.6\% \pm 2.1\%$	37.7 ± 2.5
4000	6.4%	50.8%	30.0

3.3.2 Variations in water chemistry of FFT pore water and cap water

3.3.2.1 Alkalinity

As shown in Figure 3.1a, pore water alkalinity immediately increased after the addition of high lime doses (day 0; Figure 3.1a). For example, after the addition of 1600 or 3500 ppm lime (day 0), the alkalinity increased to 2390 mg/l and 5720 mg/l (measured as CaCO₃), respectively, compared with the control alkalinity of 450 mg/l (measured as CaCO₃). The increased alkalinity at day 0 with the higher lime dose could be attributed to the increase of the hydroxide (OH⁻) concentrations, which agrees with Caughill et al. (1993).

In contrast, at day 90, the alkalinity sharply decreased at all lime doses, with only a slight change in the control. This could be argued as the pH of the FFT pore water decreased in all the lime-treated columns at day 90 (Figure 3.2a). As a result of the dissolution of carbon dioxide CO₂ into the cap water; the pH of the pore water started to decrease, and the formation of carbonate minerals, such as CaCO₃, also reduces the alkalinity (Hrynyshyn, 2012).

The alkalinity level of the cap water continuously decreased throughout all the lime experiments as well as in the control (Figure 3.1b). However, compared with the control after 90 d (315 mg/l measured as CaCO₃), the alkalinity level of the samples treated with 1600 ppm and 3500 ppm lime significantly decreased to 170 and 200 mg/l (measured as CaCO₃), respectively. The slight change in cap water alkalinity in the control experiment (20% decrease) could be attributed to methanogenesis in the FFT layer (Dompierre et al., 2016b; Siddique et al., 2014) that can increase and

maintain a stable alkalinity level in cap water. Similar observations were reported in BML (White and Liber, 2018) where methanogenesis resulted in elevated and high cap water alkalinity. Based on Figure 3.2b, the pH of the control FFT decreased below pH 7.5 at day 90. This can be attributed to CO_2 production during methanogenesis (Samadi, 2019; Siddique et al., 2014) or the contribution of atmospheric CO_2 to decrease the pH. Dissolved CO_2 is converted to carbonic acid through hydration and then dissociates to HCO_3^- (Li, 2010), which is transported to the cap water through any movement of gas (including CH_4). To confirm such prospect, further investigations are warranted about monitoring of the redox potential for the FFT layer throughout the experiments.



Figure 3.1. Change in alkalinity of (a) FFT pore-water at day 0 and at the end of the experiment (day 90); and (b) cap water at day 0, day 14, day 45, and day 90. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.



Figure 3.2. (a) FFT Pore-water pH at day 0 and at the end of the experiment (day 90). (b) FFT pH at day 0 and at the end of the experiment (day 90). All the columns consisted of 500 ml of treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere except the 4000 ppm treatment, which has two conditions: uncovered and covered with parafilm. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.
An overall decrease in cap water alkalinity in the lime-treated conditions was observed after 90 d, with the highest magnitude of the decrease for the 1600 ppm. For example, the alkalinity decreased after 90 d by 36 %, 58%, and 51% at lime doses of 650, 1600, and 3500 ppm, respectively. The decrease was significantly rapid and sharp in the first 30 d in all lime-treated FFT. A similar trend can be observed for HCO_3^- concentrations (Figure 3.3). There are two explanations for this decrease. The methanogenesis process may have been negatively influenced by lime addition which inhibited the methanogens (Nyberg et al., 2011). The changes in the microbial community are discussed later (section 3.3). Alternatively, the change in pH after lime addition could have led to the predominance of CO_3^{2-} species (MacKinnon et al., 2001). Although further studies are needed to confirm the underlying mechanisms, these results reveal the positive role of lime addition to maintain an appropriate cap water quality for the aquatic life (White, 2017; White and Liber, 2018). The changing pH is correlated with lime dosage, and Figure 3.4 displays the corresponding effect of lime treatment on the pH of cap water with time.



Figure 3.3. Change in bicarbonate (HCO₃⁻) concentration measured as (mg/l CaCO₃) in cap water at day 0, day 14, day 45, and day 90. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.



Figure 3.4. Cap-water pH at day 0, day 14, and at the end of the experiment (day 90). All the columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere except the 4000 ppm treatment which has two conditions: uncovered and covered with parafilm. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.

3.3.2.2 Cations

Figure 3.5 depicts the change in cation concentrations (Ca^{2+} and Na^+) in the FFT pore water with different lime doses. For the low lime dose (650 ppm; pH < 10), the Ca^{2+} concentration was slightly lower than the control (0 ppm lime) at day 0. Similarly, Ca^{2+} concentrations were slightly higher in the 1600 lime dose at day 0 compared to the control. An immediate sharp increase in the Ca^{2+} concentration at day 0 was observed immediately after the addition of high lime dosages 3500 and 4000 ppm (pH > 11.5). At day 90, similar Ca^{2+} concentrations were observed for all lime doses as well as the control. For lime doses 650 and 1600 ppm, there was no significant change in cation concentration (p > 0.05), while a significant reduction in cation concentration was observed at high lime doses (p < 0.05). The summary of the cation results is presented in Table 3.2.

Elucidating the influence of pH on changes in cations concentrations at days 0 and 90 can lead to a better understanding to propose the mechanisms involved. Briefly, there are two main phases of Ca^{2+} availability based on the consumption of carbonate species. In the first phase, as lime is initially added at day 0, (e.g. pH 10, lime dose 650 ppm), the equilibrium of the bicarbonate-carbonate reaction will be shifted toward CO_3^{2-} . Lime reacts with sodium bicarbonate (NaHCO₃) through water softening reaction to produce sodium hydroxide (NaOH) and insoluble calcium carbonate that precipitate as calcite (CaCO₃; (Hamza et al., 1996; MacKinnon et al., 2001). This mechanism explained the lower Ca^{2+} concentrations at day 0 for the 650 ppm lime compared with the control at this pH range (Figure

3.5a). This is the dominant reaction pathway up to a pH of approximately 11.5 (lime dose <1600 ppm), at which the carbonate species are effectively removed. Accordingly, at this phase, no soluble Ca^{2+} is available for cation exchange reactions. It is worth noting that the softening reactions are fast to occur before ion exchange mechanisms. The NaOH produced from water softening reacts with the edges of kaolinite clays, removing hydrogen ions which create new exchange sites that will be occupied by Na⁺. This proposed mechanism can explain the significant reduction in Na⁺ concentration at day 0 in the FFT pore water for the 650 ppm compared to control (Figure 3.5b). With regards to the second phase (i.e., pH >11.5, lime dose >1600 ppm), the immediate increase in the Ca^{2+} concentration at day 0 can be attributed to increased solubility of Ca²⁺ from the dissolution of Ca[OH]₂(MacKinnon et al., 2001; Tate et al., 2016). This provides Ca²⁺ ions that can exchange on mineral surfaces. This can be demonstrated where the soluble Ca²⁺ exchanges and displaces the Na⁺ increasing its concentration in the pore water (i.e., Na⁺ concentrations are 210, 300, and 315 mg/L at 1600, 3500, and 4000 ppm lime dosage respectively).



Figure 3.5. Concentrations of cations for treated and untreated FFT pore water at day 0 and at the end of the experiment (day 90) measured by ICP-OES. a) Ca^{2+} and b) Na⁺. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.

Table 3.2. Concentrations of cations Ca^{2+} and Na^+ for treated and untreated FFT pore water and cap water at day 0 and the end of the experiment (day 90) measured by ICP-OES. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.

		С	a ²⁺			
Sample	Pore water			Cap wat	er	
ID	Day 0	Day 90)	Day 0	Day 90	
0 ppm	48.3 ± 1.3	38.7 ± 0.7		14.1 ± 0.4	15.7 ± 0.6	
650 ppm	15.2 ± 3.9	21.4 ± 35.2		$\begin{array}{c} 13.6 \pm \\ 0.2 \end{array}$	20.2 ± 0.4	
1600 ppm	120.8 ± 20.9	24.6 ± 6	.5	14.5 ± 1.3	6.8 ± 1.6	
3500 ppm	557.4 ± 73.4	60.5 ± 7	.4	$\begin{array}{c} 14.4 \pm \\ 0.8 \end{array}$	4.0 ± 1.6	
4000 ppm	544.6 ± 60.9	49.8 ± 0	.0	12.60	2.2	
	Na ⁺					
Sample ID -	Pore water	Cap water				
	Day 0	Day 90	Ι	Day 0	Day 90	
0 ppm	272.1 ± 5.7	298.9 ± 3.0	294.3 ± 1.0		299.4 ± 3.2	
650 ppm	204.9 ± 1.2	$279.6\pm\!\!1.2$	303.7 ± 3.7		303.3 ± 1.0	
1600 ppm	223.6 ± 12.5	247.4 ± 4.7	298.8 ± 7.7		281.4 ± 4.3	
3500 ppm	277.9 ± 9.3	266.4 ± 17.4	298.8 ± 5.9		303.6 ± 6.8	
4000 ppm	295.4 ± 1.3	240.4 ± 0		283.7	294.5	

With regards to day 90, we can hypothesize that increasing the pH > 12 (i.e., 3500 ppm and 4000 ppm lime) promote the pozzolanic reaction where soluble Ca^{2+} started to react with dissolved silica and alumina from the clay forming hydrated minerals of calcium aluminate and silicate (Tate et al., 2017; Wang, 2017). This can elucidate the reduction of Ca^{2+} concentration at day 90 for the pore water. Lane (1983) suggested that the pozzolanic reaction can occur with lime addition at high pH in which a stabilization process can occur through lime addition (i.e., the authors suggested coagulation process as a good description of the lime stabilization in soil and tailings). Furthermore, the lime addition can modify particle surface charge and lead to agglomeration and interparticle bonding. In the same sense, the study elucidated the stabilization reactions as the pozzolanic reactions and carbonation reactions.

In terms of the cap water, Na⁺ concentration was almost stable throughout the experiment at different lime doses (Figure 3.6b). In contrast, Ca²⁺ levels decreased in the cap water with increasing lime dose (Figure 3.6a). The change in Ca²⁺concentrations confirms that CO₂ is being absorbed by the system to react with alkaline Ca²⁺. The Student's t-test results suggested significant differences in Ca²⁺ concentrations at day 90 at high lime doses (>1600 ppm) compared with lower lime doses (p < 0.05). Our investigations generally depicted a marginal drop in K⁺ concentration for all lime doses throughout the experiment (Figure 3.7) that slightly varied with the control. Although this could partially correspond with the findings of White and Liber (2018), who reported a slight change in K⁺ levels in the surface water of BML over three consecutive years, further investigations are warranted to confirm this phenomenon. In summary, confirming cation exchange and pozzolanic reactions occurrence would require a more focused study.



Figure 3.6. Variation of cation (Ca²⁺ and Na⁺) concentrations in cap water at day 0 and at the end of the experiment (day 90) measured by ICP-OES. Illustrated results are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW (Raw SPW). All the columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.



Figure 3.7. Variation of cation K^+ concentrations in cap water at day 0 and the end of the experiment (day 90) measured by ICP-OES. Illustrated results are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) ± one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW (Raw SPW). All the columns were open to the atmosphere. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.

3.3.3 Variations in bacterial cell counts

The initial indigenous microbial cell counts are represented by the number found in the control sample on day 0 (8.15×10^{10} cells/µl), and is used to compare the effect of lime treatment on the microorganisms. The addition of lime resulted in an immediate adverse impact on the bacterial cell counts at day 0 after lime addition, and the microbial count was considerably reduced for all lime conditions (Figure 3.8). On day 90, the bacterial cell counts were significantly decreased at high lime doses (pH > 11.5), with the greatest reduction of cell counts occurring at 4000 ppm lime under covered conditions (4.41 ×10⁴ cells/µl; 98% reduction) resulting from the sharp pH increase. In contrast, at a lime dose of 650 ppm (pH < 10) there was a minimal reduction in the microbial cell counts after 90 d, highlighting the adverse impact of increased pH, particularly pH > 11.5, on the cell counts. Overall, there is significant difference in the microbial cell counts of the lime-treated FFT (i.e., lime doses > 1600 ppm; p < 0.05), but no significant difference in the control or low lime-treated FFT (i.e., p > 0.05 for both control and 650 ppm lime).

Our findings correspond with previous studies that showed a similar adverse impact of a rapid increase in pH (pH > 12) on bacterial cell membranes, which inhibited their growth (Burns and Gremminger, 1994; Nyberg et al., 2011; Wong et al., 2001; Wong and Selvam, 2009). Furthermore, the addition of lime (pH 11) was demonstrated to effectively inactivate pathogens in municipal biosolids (Bennett et al., 2003).



Figure 3.8. Bacterial cell counts for treated and untreated FFT at day 0 and at the end of the experiment (day 90) measured by qPCR of the rpoB gene. Results illustrated an average (n = 2 for control and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere except the 4000 ppm which has two conditions: uncovered and covered with parafilm. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.

3.3.4 Impact of lime treatment on organic fractions

To determine the impact of lime treatment on organic fractions, we characterized both PHCs and DOC. Figure 3.9a shows the concentrations of PHC fractions after different lime treatments at day 90 compared with the initial concentration in untreated FFT (control) at day 0 and day 90. The greatest reduction in PHC concentrations is observed at 650 ppm lime, with removal efficiencies of 54%, 57%, 60%, and 66% for F2, F3, F4, and bitumen, respectively. However, for lime doses of 1600, 3500, and 4000 ppm, the PHC reductions were minimal compared with the reduction in the control sample after 90 d. The high bacterial cell counts in the

sample treated with 650 ppm lime could explain the better degradation of PHCs compared with the degradation in the high lime doses (>1600 ppm).

As stated earlier, at a lime dose of 650 ppm, Ca²⁺ reacts with carbonate and precipitates as CaCO₃ (Hamza et al., 1996; MacKinnon et al., 2001). In comparison to control, both F4 and bitumen fractions (i.e., the heaviest fractions of PHCs) might complex with calcite precipitates (CaCO₃) and then settle down. Nevertheless, the F2 fraction (i.e., demonstrated slight reduction, Figure 3.9a), which is the highly volatile fraction of PHC (Brickner, 2013), seems to solubilize from the bitumen and dissolve in the FFT pore water.

Our findings highlight the negative influence of high doses of lime on microbial cell counts that subsequently affect the degradation of the PHCs. Various studies have reported the effectiveness of various microbial communities such as hydrocarbon-degrading microorganisms, sulfate-reducing bacteria, and iron-reducing bacteria on the degradation of PHCs (Allen et al., 2007; Pandey et al., 2009). Similarly, Yu et al. (2018a) suggested that the hydrocarbons present in tailings serve as a carbon source for the indigenous microorganisms. These findings can explain the reduction of the PHC fractions in the control columns after 90 d.

The variations in DOC concentrations in the pore water throughout the experiment are illustrated in Fig. 3.9b. No significant change was observed between day 0 and day 90 at 650 ppm; however, at lime doses of 1600, 3500, and 4000 ppm, a significant increase in DOC was observed. The changes in DOC agree with the minimal decrease in the PHC fraction; in which at high lime doses, bacterial cell numbers were decreased and were not able to degrade either the PHC or DOC. Additionally, during cation exchange when pH > 11.5, the F2 fraction might dissolve in the FFT pore water and increase the DOC concentration. This mechanism is similar

to one proposed by Fine et al. (1997) that highly volatile fractions of PHCs can dissolve into soil pore water or groundwater. The increase in DOC may also be ascribed to the release of organic compounds into the cap water during FFT dewatering (Yu et al., 2018a).



Figure 3.9. (a) Petroleum hydrocarbons (PHC) concentrations for lime-treated FFT at the end of experiment (day 90) compared with the untreated/0 ppm FFT at day 0 and day 90. (b) Dissolved organic carbon (DOC) concentrations for pore water at day 0 and at the end of the experiment (day 90). All columns consisted of 500 ml treated or untreated FFT covered with 500 ml of Raw synthetic OSPW. All the columns were open to the atmosphere except the 4000 ppm treatment, which was covered with parafilm. Results illustrated are an average (n = 2 for 0 ppm and n = 3 for 650, 1600, and 3500 ppm) \pm one standard deviation. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.

3.3.5 Toxicity and implications of pH influence

Table 3.3 summarizes the impact of lime addition on the toxicity to *V. fischeri* of the FFT pore water (days 0 and 90) and cap water (day 14 and day 90). High lime doses (i.e., > 1600 ppm) caused higher toxicity of the FFT pore water (above instrument detection limit) compared with the lower toxicity in the control and in the 650 ppm lime dose (0.1 and 1.2 TU, respectively). These results are consistent with the negative impact of lime addition (i.e., > 1600 ppm) on bacterial cell counts measured in the FFT.

With respect to the acute toxicity effect of the cap water to *V. fischeri* over time (Table 3.3), for lime doses ≤ 1600 ppm (pH ≤ 11.5), the cap water at day 14 was non-toxic after lime addition. At high lime doses (3500 and 4000 ppm; pH > 11.5), the toxicity at day 14 slightly increased to 0.8 and 1.1 TU at 3500 and 4000 ppm, respectively, then decreased at day 90 to 0.6 and 0.4 TU for 3500 and 4000 ppm, respectively.

Even though the lime treated samples (uncovered conditions after 90 days) in comparison to the control (Table 3.3), showed a slight increase in toxicity; the toxicity of the cap water for all lime doses did not exceed 1 TU (Table 3.3). This translates as a nontoxic effect toward *V. Fisheri*. To confirm our observations, principal component analysis (PCA, Figure 3.10) and cluster analysis (Figure 3.11) were assessed to pinpoint the influence of lime treatment on cap water toxicity after 14 and 90 days. The PCA data support very minimal variations and high similarity levels (i.e., statistically confirmed as insignificant different at 95% confidence interval and significance level of 0.05) between the toxicity effect of lime treated samples and control samples in cap water. It is worth noting that our rationale was to monitor the toxicity of the cap water after FFT lime treatment which necessitate using the synthetic OSPW in the experiment without any organics to better understand the movement of COPCs. According to our earlier observations about the adverse effects on WR occurred due to overdosing (i.e., 4000 ppm), we have decided to elucidate the potential influence of any adverse impact of overdosing or discrepancies between covered and uncovered conditions with regards to toxicity. For the 4000 ppm covered condition, a significant increase in toxicity occurred at both days 14 and 90 (14.83 and 16.13 TU, respectively). This can be initially attributed to the significant effect on pH at high lime doses, as shown in Figure 3.4. Although at day 14, both of the 4000 ppm conditions (covered and uncovered) had a similar pH (\approx 9.8), the TUs were quite different (1.1 TU for the uncovered condition; 14.8 TU for the covered condition). Therefore, high toxicity is not only related to the pH increase in the covered condition, but also possibly to COPCs moving from the FFT into the cap water. These results highlight the significant impact of atmospheric CO₂ to reduce cap water pH and generate a carbonate buffer in the uncovered conditions.

In summary, the slight increase in toxicity of cap water samples can be related to the movements of COPCs from FFT to the released pore water and cap water. This would necessitate more toxicological investigations to examine the critical concentrations, species, and the environmental impact of lime treatment on live-dead viability assays. This would develop better understanding of the influence of end pit lake surface water on acute toxicity and in vitro assays (e.g. RAW 264.7 mouse macrophage cell line and the response of bone marrow-derived macrophages).

Table 3.3. Toxicity of FFT pore water presented in Toxicity Units (TU) at day 0 and the end of the experiment (day 90) and cap water toxicity at day 14 and the end of the experiment (day 90) assessed using Microtox. TU < 1.0 indicates complete detoxification (Scott et al., 2008). All columns consisted of 500 ml treated or untreated FFT covered with 500 ml Raw synthetic OSPW. All the columns were open to the atmosphere except the 4000 ppm treatment, which was covered with parafilm. The untreated FFT represents a 0 ppm treatment and acts as the control experiment.

Sample ID	Toxicity Unit (TU)					
	Pore water		Cap water			
	Day 0	Day 90	Day 14	Day 90		
0 ppm	Non-toxic*	0.1	0	0.1		
650 ppm	1.3	1.2	0	0.4		
1600 ppm	High level	4.3	0	0.3		
3500 ppm	High level	High level	0.8	0.6		
4000 ppm	High level**	High level	1.1	0.4		
4000 ppm	High level	High level	14.9	16.2		
Covered						



Figure 3.10. (a) Score plot for PCA analysis of all lime treated samples and control, PC 1 (99.9%) and PC 2 (0.1%), (b) Biplot and loading plot of PC1 and PC2 with project lines of all samples. The vectors radiating from the origin represent the sample loadings. Samples that are chemically similar are plotted near to each other (clustered together), while sample scores are color-coded based on treatment conditions and symbols indicated by square or circle.



Figure 3.11. Dendrogram of the cluster analysis for toxicity effects of cap water.

3.4 Conclusions

While mimicking the EPL approach, this study investigated lime application to control the water quality in EPLs and aid the dewatering of FFT from oil sands operations. In terms of the better performance of WR, alkalinity of cap water, and cation exchange, lime dose of 3500 ppm achieved 7.9% WR despite the relatively high initial solids content of FFT (50% wt./wt.). The alkalinity of the cap water decreased after adding lime dose of 3500 ppm, compared with the untreated FFT, promoting and maintaining an appropriate, low alkalinity environment for an aquatic ecosystem in EPLs. Furthermore, the cation concentrations indicate that cation exchange occurs at high lime doses (pH > 11.5), which aids in clay settling. However, high lime doses (1600, 3500, and 4000 ppm; pH > 10) had a negative impact on microbial cell numbers in the underlying FFT, resulting in minimal reduction of organics (PHC) at high lime doses compared with moderate reduction

at a low lime dose (650 ppm). Assessing the toxicity of the FFT pore water after lime treatment demonstrated high toxicity effects to V. fischeri because of the increased pH. Nevertheless, the toxicity of the cap water was not affected by the high toxicity of the FFT, except under the covered conditions. This highlights a considerable influence of atmospheric CO₂ on regenerating the carbonate buffer and reducing pH of the cap water over time. As such any pH transport from the FFT will rapidly be neutralized due to the associated impact of dissolution of atmospheric CO_2 into the cap water. Overall, the performance metrics and indicative parameters (e.g. increase the WR% from the FFT, increase the possibility of the cation exchange of Ca²⁺ with the clay particles, and decrease the alkalinity of the cap water) revealed the role of lime addition towards the reclamations benefits within EPLs. Future work is still required to build upon the current findings in order to further optimize the lime treatment and the occurrence of cation exchange. The movement of COPCs into the cap water should be further investigated with a focus on chemical characterization of the FFT organic fractions and their associated potential risks toward the aquatic life in the EPLs. This better understanding will infer how to proceed to larger-scale and successful applications of lime treatment in EPLs.

3.5 References

- Allen, E.W. 2008a. Process water treatment in Canada's oil sands industry: II. A review of emerging technologies. Journal of Environmental Engineering and Science 7(5), 499-524.
- Allen, E.W. 2008b. Process water treatment in Canada's oil sands industry: I. Target pollutants and treatment objectives. Journal of Environmental Engineering and Science 7(2), 123-138.
- Allen, J.P., Atekwana, E.A., Atekwana, E.A., Duris, J.W., Werkema, D.D. and Rossbach, S. 2007. The microbial community structure in petroleumcontaminated sediments corresponds to geophysical signatures. Appl. Environ. Microbiol. 73(9), 2860-2870.
- Anderson, J.C., Wiseman, S.B., Wang, N., Moustafa, A., Perez-Estrada, L., Gamal El-Din, M., Martin, J.W., Liber, K. and Giesy, J.P. 2011. Effectiveness of Ozonation Treatment in Eliminating Toxicity of Oil Sands Process-Affected Water to Chironomus dilutus. Environ. Sci. Technol. 46(1), 486-493.
- Baillie, R.A. and Malmberg, E.W. 1969 Removal of clay from the water streams of the hot water process by flocculation, Google Patents.
- Beaty, N.B. and Lane, M.D. 1983. The polymerization of acetyl-CoA carboxylase. Journal of Biological Chemistry 258(21), 13051-13055.

- Bennett, D.D., Higgins, S.E., Moore, R.W., Beltran, R., Caldwell, D.J., Byrd, J.A. and Hargis, B.M. 2003. Effects of lime on Salmonella enteritidis survival in vitro. Journal of applied poultry research 12(1), 65-68.
- Board, E.R.C. 2012 Energy Resources Conservation Board Tailings Management Assessment Report: Oil
- Sands Mining Industry, (posted date: June 12, 2013), from Alberta Energy Regulator, Accessed online

April

2017: http://osipfiles.alberta.ca/datasets/158/TailingsManagementAssess mentReport2011-2012.pdf.

- Bowman, D.T., Arriaga, D., Morris, P., Risacher, F., Warren, L.A., McCarry, B.E. and Slater, G. 2016. Chemical fingerprinting of naphthenic acids at an oil sands end pit lake by comprehensive two-dimensional gas chromatography/time-of-flight mass spectrometry (GC× GC/TOFMS). AGUFM 2016, H23C-1563.
- Bradford, L.M., Ziolkowski, L.A., Goad, C., Warren, L.A. and Slater, G.F. 2017. Elucidating carbon sources driving microbial metabolism during oil sands reclamation. Journal of environmental management 188, 246-254.
- Brickner, H. 2013. The mobility of petroleum hydrocarbons in Athabasca oil sands tailings.
- Brown, L.D., Pérez-Estrada, L., Wang, N., El-Din, M.G., Martin, J.W., Fedorak, P.M. and Ulrich, A.C. 2013. Indigenous microbes survive in situ ozonation

improving biodegradation of dissolved organic matter in aged oil sands process-affected waters. Chemosphere 93(11), 2748-2755.

- Burns, H. and Gremminger, L. 1994 Lime and fly ash stabilization of wastewater treatment sludge, Google Patents.
- Caughill, D.L., Morgenstern, N.R. and Scott, J.D. 1993. Geotechnics of nonsegregating oil sand tailings. Canadian Geotechnical Journal 30(5), 801-811.
- CCME 2008. CANADA-WIDE STANDARD FOR PETROLEUM HYDROCARBONS (PHC) IN SOIL.
- Chalaturnyk, R.J., Don Scott, J. and Özüm, B. 2002. MANAGEMENT OF OIL SANDS TAILINGS. Petroleum Science and Technology 20(9-10), 1025-1046.
- Clemente, J.S. and Fedorak, P.M. 2005. A review of the occurrence, analyses, toxicity, and biodegradation of naphthenic acids. Chemosphere 60(5), 585-600.
- COSIA 2012 Technical Guide for Fluid Fine Tailings Management pp. Report, 1-131., Canadian Oil Sands Innovation Alliance, Edmonton, AB. August2012, Edmonton, Alberta, Canada.
- Dean, E.W. and Stark, D.D. 1920. A Convenient Method for the Determination of Water in Petroleum and Other Organic Emulsions. Industrial & Engineering Chemistry 12(5), 486-490.

- Dompierre, K.A. and Barbour, S.L. 2016. Characterization of physical mass transport through oil sands fluid fine tailings in an end pit lake: a multi-tracer study. Journal of Contaminant Hydrology 189, 12-26.
- Dompierre, K.A., Lindsay, M.B.J., Cruz-Hernández, P. and Halferdahl, G.M. 2016a. Initial geochemical characteristics of fluid fine tailings in an oil sands end pit lake. Science of The Total Environment 556(Supplement C), 196-206.
- Dompierre, K.A., Lindsay, M.B.J., Cruz-Hernández, P. and Halferdahl, G.M. 2016b. Initial geochemical characteristics of fluid fine tailings in an oil sands end pit lake. Science of The Total Environment 556, 196-206.
- Dompierre, K.A.H. 2016. Controls on mass and thermal loading to an oil sands end pit lake from underlying fluid fine tailings.
- Ewin, G.J., Erno, B.P. and Hepler, L.G. 1981. Clay chemistry: Investigation of thermodynamics of ion exchange reactions by titration calorimetry. Canadian Journal of Chemistry 59(20), 2927-2933.
- Famakinwa, T.E., Su, Y., Wang, J. and Gates, I.D. 2018. An in-situ process to consolidate oil sands mine tailings. Journal of Environmental Chemical Engineering 6(2), 3295-3305.
- Ferris, A.P. and Jepson, W.B. 1975. The exchange capacities of kaolinite and the preparation of homoionic clays. Journal of Colloid and Interface Science 51(2), 245-259.

- Fine, P., Graber, E.R. and Yaron, B. 1997. Soil interactions with petroleum hydrocarbons: abiotic processes. Soil Technology 10(2), 133-153.
- Hamza, H.A., Stanonik, D.J. and Kessick, M.A. 1996. Flocculation of lime-treated oil sands tailings. Fuel 75(3), 280-284.
- Hrynyshyn, J. 2012. End pit lakes guidance document 2012. Cumulative Environmental Management Association, Fort McMurray, Alberta. CEMA Contract (2010-0016).
- Hyndman, A. and Sobkowicz, J. 2010, Oil sands tailings: reclamation goals and the state of technology, pp. 12-16.
- Kabwe, L.K., Scott, J.D., Beier, N.A., Wilson, G.W. and Jeeravipoolvarn, S. 2018. Environmental Implications of End Pit Lakes at Oil Sands Mine. Environmental Geotechnics, 1-25.
- Lawrence, G.A., Tedford, E.W. and Pieters, R. 2015. Suspended solids in an end pit lake: potential mixing mechanisms. Canadian Journal of Civil Engineering 43(3), 211-217.
- Li, C. (2010) Methanogenesis in oil sands tailings. [electronic resource] : an analysis of the microbial community involved and its effects on tailings densification, 2010.
- Lorentz, J., Bara, B., Mikula, R., Leneve, E. and Wang, N. 2014 Oil sands fluid fine tailings dewatering using rim ditching, Google Patents.
- Ma, C. and Eggleton, R.A. 1999. Cation exchange capacity of kaolinite. Clays and Clay minerals 47(2), 174-180.

- MacKinnon, M.D. 1989. Development of the tailings pond at Syncrude's oil sands plant: 1978–1987. AOSTRA J. Res 5(2), 109-133.
- MacKinnon, M.D., Matthews, J.G., Shaw, W.H. and Cuddy, R.G. 2001. Water quality issues associated with composite tailings (CT) technology for managing oil sands tailings. International Journal of Surface Mining, Reclamation and Environment 15(4), 235-256.

Mahdavi, H. (2014) THE APPLICATION OF ALGAE FOR THE REMOVAL OF METALS AND NAPHTHENIC ACIDS FROM OIL SANDS TAILINGS POND WATER.

- Mahdavi, H., Prasad, V., Liu, Y. and Ulrich, A.C. 2015. In situ biodegradation of naphthenic acids in oil sands tailings pond water using indigenous algae– bacteria consortium. Bioresource technology 187, 97-105.
- Masliyah, J.H., Xu, Z. and Czarnecki, J.A. (2011) Handbook on theory and practice of bitumen recovery from Athabasca oil sands, Kingsley Knowledge Pub.
- Matthews, J.G., Shaw, W.H., MacKinnon, M.D. and Cuddy, R.G. 2002.Development of composite tailings technology at Syncrude. International Journal of Surface Mining, Reclamation and Environment 16(1), 24-39.
- McQueen, A.D., Kinley, C.M., Hendrikse, M., Gaspari, D.P., Calomeni, A.J., Iwinski, K.J., Castle, J.W., Haakensen, M.C., Peru, K.M. and Headley, J.V. 2017. A risk-based approach for identifying constituents of concern in oil sands process-affected water from the Athabasca Oil Sands region. Chemosphere 173, 340-350.

- Miles, S.M., Hofstetter, S., Edwards, T., Dlusskaya, E., Cologgi, D.L., Gänzle, M. and Ulrich, A.C. 2019. Tolerance and cytotoxicity of naphthenic acids on microorganisms isolated from oil sands process-affected water. Science of The Total Environment 695, 133749.
- Nyberg, K.A., Vinnerås, B., Lewerin, S.S., Kjellberg, E. and Albihn, A. 2011. Treatment with Ca(OH)2 for inactivation of Salmonella Typhimurium and Enterococcus faecalis in soil contaminated with infected horse manure. Journal of Applied Microbiology 110(6), 1515-1523.
- Pandey, J., Chauhan, A. and Jain, R.K. 2009. Integrative approaches for assessing the ecological sustainability of in situ bioremediation. FEMS Microbiology Reviews 33(2), 324-375.
- Poon, H.Y. 2019 An Examination of the Effect of Diluent on Microbial Dynamics in Oil Sands Tailings and the Mechanistic Insight on Carbon Dioxidemediated Turbidity Reduction in Oil Sands Surface Water.
- Poon, H.Y., Brandon, J.T., Yu, X. and Ulrich, A.C. 2018. Turbidity Mitigation in an Oil Sands Pit Lake through pH Reduction and Fresh Water Addition. Journal of Environmental Engineering 144(12), 04018127.
- Proskin, S., Sego, D. and Alostaz, M. 2012. Oil sands MFT properties and freezethaw effects. Journal of Cold Regions Engineering 26(2), 29-54.
- Richardson, E., Bass, D., Smirnova, A., Paoli, L., Dunfield, P. and Dacks, J.B. 2020. Phylogenetic estimation of community composition and novel

eukaryotic lineages in Base Mine Lake: an oil sands tailings reclamation site in Northern Alberta. Journal of Eukaryotic Microbiology 67(1), 86-99.

- Risacher, F.F., Morris, P.K., Arriaga, D., Goad, C., Nelson, T.C., Slater, G.F. and Warren, L.A. 2018. The interplay of methane and ammonia as key oxygen consuming constituents in early stage development of Base Mine Lake, the first demonstration oil sands pit lake. Applied Geochemistry 93, 49-59.
- Samadi, N. 2019. Partitioning of inorganic contaminants between fluid fine tailings and cap water under end pit lake scenario: Biological, Chemical and Mineralogical processes.
- Scott, A.C., Zubot, W., MacKinnon, M.D., Smith, D.W. and Fedorak, P.M. 2008.Ozonation of oil sands process water removes naphthenic acids and toxicity.Chemosphere 71(1), 156-160.
- Siddique, T., Kuznetsov, P., Kuznetsova, A., Arkell, N., Young, R., Li, C., Guigard, S.E., Underwood, E. and Foght, J.M. 2014. Microbially-accelerated consolidation of oil sands tailings. Pathway I: changes in porewater chemistry. Frontiers in microbiology 5, 106.
- Tate, M., Leikam, J., Fox, J. and Romaniuk, N. 2017 Use of calcium hydroxide as a coagulant to improve oil sands tailings treatment. Proceedings Tailings and Mine Waste 2017, Banff, Alberta, Canada, November 2017.
- Tate, M., Leikam, J., ScotT, J.D., Mehranfar, M., Romaniuk, N. and Ozum, B. 2016 Impacts of Calcium Compounds on Oil Sands Water Chemistry.

Conference presentation at the Mining and Tailings Waste Conference, keystone, Colorado, USA.

- Vajihinejad, V. and Soares, J.B.P. 2018. Monitoring polymer flocculation in oil sands tailings: A population balance model approach. Chemical Engineering Journal 346, 447-457.
- Wang, C., Alpatova, A., McPhedran, K.N. and Gamal El-Din, M. 2015. Coagulation/flocculation process with polyaluminum chloride for the remediation of oil sands process-affected water: Performance and mechanism study. Journal of Environmental Management 160(Supplement C), 254-262.
- Wang, C., Harbottle, D., Liu, Q. and Xu, Z. 2014. Current state of fine mineral tailings treatment: A critical review on theory and practice. Minerals Engineering 58, 113-131.
- Wells, P.S. and Riley, D.A. 2007, MFT drying—case study in the use of rheological modification and dewatering of fine tailings through thin lift deposition in the oil sands of Alberta, pp. 13-15.
- White, K.B. 2017. Characterizing annual changes in the chemistry and toxicity of surface water from Base Mine Lake, an Alberta oil sands end pit lake.
- White, K.B. and Liber, K. 2018. Early chemical and toxicological risk characterization of inorganic constituents in surface water from the Canadian oil sands first large-scale end pit lake. Chemosphere 211, 745-757.

- White, K.B. and Liber, K. 2020. Chronic Toxicity of Surface Water from a Canadian Oil Sands End Pit Lake to the Freshwater Invertebrates Chironomus dilutus and Ceriodaphnia dubia. Archives of Environmental Contamination and Toxicology 78(3), 439-450.
- Wong, J.W.C., Fang, M. and Jiang, R. 2001. Persistency of bacterial indicators in biosolids stabilization with coal fly ash and lime. Water environment research 73(5), 607-611.
- Wong, J.W.C. and Selvam, A. 2009. Reduction of indicator and pathogenic microorganisms in pig manure through fly ash and lime addition during alkaline stabilization. Journal of Hazardous Materials 169(1-3), 882-889.
- Yu, X., Cao, Y., Sampaga, R., Rybiak, S., Burns, T. and Ulrich, A.C. 2018a. Accelerated dewatering and detoxification of oil sands tailings using a biological amendment. Journal of Environmental Engineering 144(9), 04018091.
- Yu, X., Lee, K., Ma, B., Asiedu, E. and Ulrich, A.C. 2018b. Indigenous microorganisms residing in oil sands tailings biodegrade residual bitumen. Chemosphere 209, 551-559.
- Yu, X., Lee, K. and Ulrich, A.C. 2019. Model naphthenic acids removal by microalgae and Base Mine Lake cap water microbial inoculum. Chemosphere 234, 796-805.

Zubot, W. 2010. Water use and aquatic reclamation at Syncrude Canada Ltd. Integrated oil sands mining operation. The Association of Professional Engineers, Geologists, and Geophysicists of Alberta (APEGA).

4 ENZYME-ASSISTED DEWATERING OF OIL SANDS TAILINGS: SIGNIFICANCE OF WATER CHEMISTRY AND BIOLOGICAL ACTIVITY³

4.1 Introduction

The oil sands reserves in Alberta are the third largest reserves of crude oil worldwide. Surface-mined oil sands ore in north-eastern Alberta produces almost two-thirds of the total bitumen production in Alberta (Kaminsky et al. 2009). Hotwater bitumen extraction currently demands large volumes of water, ranging from 2 to 2.5 m³ per m³ of crude oil produced (Zhu et al. 2017; Zubot et al. 2012). The mining companies reuse the water to a certain extent based on water quality; however, a large amount of slurry (known as tailings) is continuously generated from the process and stored in engineered tailings ponds (Government of Alberta 2015; Congressional Research Service 2008). These tailings and process water cannot be directly discharged into the environment due to a zero-discharge practice following environmental legislation (ERCB 2012; McQueen et al. 2017). Until 2017, the total area occupied by oil sands tailings (AER 2019; Foght et al. 2017; Miles et al. 2020; Vajihinejad et al. 2017). Oil sands tailings typically contain 20–

³ A version of this chapter has been published as "Allam, N.E.; Anwar, M. N.; Kuznetsov, P.V.; Ulrich, A. C.; Dhar, B. R. (2022) "Enzyme-assisted dewatering of oil sands tailings: Significance of water chemistry and biological activity" Chemical Engineering Journal, 437, 135162.

30 wt% solids (i.e., sand, silt, clays, dissolved salts, organics, and minerals) and 1– 3 wt% residual bitumen. In the tailings ponds, the coarse sands can quickly settle when deposited, while fine solids and residual bitumen remain suspended. Mackinnon and Zubot (2013) suggested that the tailings will settle into three layers, with the course sands at the bottom, tailings in the middle (known as fluid fine tailings [FFT] or mature fine tailings [MFT]), and water on top. Within a few weeks, the fine tailings can settle up to 20 wt% solids; however, several years are required to reach 30–35 wt% solids (Allen 2008b). In order to achieve regulatory guidelines, oil sands operators need advanced reclamation strategies to manage such massive volumes of tailings (Allen 2008a; Allen 2008b; Clemente and Fedorak 2005; Siddique et al. 2014).

A number of engineering strategies, including biological, chemical, physical, and/or their combinations, have been examined during the last decades to achieve sustainable reclamation, emphasizing dewatering of tailings. Although these methods developed the tailings stream and enhanced the solids content to 50–60% (Devenny 2010), subsequent dewatering of tailings is still required for eventual reclamation and safe deposition as a soil (Owolagba and Azam 2015). The dewatering processes based only on chemical methods could treat much larger volumes with less energy; however, some chemical methods could introduce toxic and/or recalcitrant species into the water that are difficult to remove. For example, despite the NA reduction after ozonation treatment of OSPW reported by Anderson et al. (2011), there was increasing toxicity due to the formation of toxic by-products post ozonation. Additionally, traditional methods can be applied to dewater the

FFT, including consolidated tailings (CT), centrifugation, thin lift dewatering, filtering, rim ditching, and the freeze and thaw method, but the cost of these methods is high (Kasperski and Mikula 2011; Masliyah et al. 2011). Therefore, several studies have addressed the combined effect of natural processes (e.g., evaporation) with chemical process (e.g., coagulation and flocculation) to increase the FFT strength (Yuan and Shaw 2007). Combinations of mechanical and chemical dewatering techniques, such as flocculation using polymers followed by filtration or centrifugation, have also been investigated. However, such energy-intensive solutions are beneficial only if the recovered sediments have little water (Gumfekar et al. 2017).

In comparison with physical and chemical treatment, biological treatment is considered more cost-effective and associated with better water quality (Yu et al. 2018a). However, limited research studies have investigated biological treatment for accelerating the dewatering of tailings. Siddique et al. (2014) investigated the effectiveness of hydrolyzed canola meal as an organic substrate to improve microbial activity and improve the dewatering of tailings. They reported an 8% increase in the water recovery compared with the untreated condition. Recently, Young et al. (2020) reported an increase in solids content from 23.2 wt% to 39.4 wt% for FFT amended with hydrolyzed canola meal and acclimatized in an anaerobic environment. Yu et al. (2018a) achieved 8% water recovery from FFT using Ultrazyme, a proprietary digestive enzyme. These recent studies primarily observed cracks in the tailings during the dewatering process. These cracks could be attributed to biogenic gas, facilitating pathways for pore water release. Previous
studies reported the presence of methanogens in oil sands tailings ponds that can slowly biodegrade BTEX and residual hydrocarbons to biomethane (Penner and Foght 2010; Siddique et al. 2011). Consistent with the report by Yu et al. (2018a) (i.e., biogenic gas production due to enzymatic treatment of tailings), several studies previously reported enhanced methane production due to improved hydrolysis via enzymes amendment in conventional anaerobic bioreactor studies (Ehimen et al. 2013; Kim et al. 2019; Odnell et al. 2016; Parawira 2012; Passos et al. 2016).

In conventional wastewater treatment, many studies have examined the feasibility of enzymatic treatment to improve the dewaterability of wastewater sludge. Parmar et al. (2001) investigated the influence of composite hydrolysis enzymes (cellulase, protease, and lipase) to enhance the dewatering of waste activated sludge. Ayol and Dentel (2005) investigated the use of a commercial product called Enviro-Zyme 216 which contains protease, lipidase, anaerobic bacteria, Aspergillus oryzae, and a mixture of enzyme complex and polymer to increase the dewaterability of anaerobically digested biosolid samples. Their study demonstrated enhanced final solid contents, filtrate turbidity, and suspended solids based on the capillary suction test. Moreover, a study by Dursun et al. (2006) using enzymatic treatment showed a 27% increase in the cake solids content in simulated belt filter press dewatering anaerobically digested sludge at laboratory scale. In addition to sewage sludge, enzymatic treatment (cellulase, protease, and lysozyme) was also effective for treating pulp and paper mill sludge (Bonilla et al. 2015). Notably, lysozyme could increase the solids content in pulp and paper sludge from 5.6% to 8.9% (Bonilla et al. 2015). Despite these positive reports on the effectiveness of enzymes on dewatering, enzymatic treatment has not yet been thoroughly explored for accelerating the dewatering of oil sands tailings. Based on an extensive literature search, only one study (2018a) could be found examining the feasibility of a commercial enzyme (Ultrazyme) as an amendment for the dewatering of oil sands tailings. However, detailed insights on the effects of enzymes on water chemistry and biogenic activity are yet to be revealed to understand the subsequent impact on oil sands tailings dewatering.

Thus, this study investigated three different enzymes (cellulase, protease, and lysozyme) to aid in the reclamation of oil sands tailings and identify the significance of water chemistry and biogenic activity in dewatering. The detailed objectives of this study are a) to examine the technical feasibility of enzymatic treatment to enhance FFT dewatering; b) to investigate the corresponding mechanism for the dewatering; c) to study the influence of enzymatic treatment on the reduction or removal of organic fractions such as naphthenic acid fractions (NAFCs); and d) to determine shifts in microbial community diversity and toxicity toward *Vibrio fischeri*.

4.2 Materials and methods

4.2.1 Materials

FFT was collected from the oil sands region in northern Alberta and stored in sealed buckets at 4 °C until use. Three different enzymes were used in this study: cellulase from *Trichoderma reesei* (>700 U/g; Sigma-Aldrich, Canada); Biomedicals[™], a protease from *Bacillus* sp. (>16 U/g; Sigma-Aldrich, Canada); and MP Biomedicals[™], a lysozyme from chicken egg white (20,000 U/mg; Fisher Scientific, Canada). Based on previous reports, these enzymes could effectively dewater sewage sludge and pulp and paper mill sludge (Bonilla et al. 2015; Lin et al. 2019; Liu et al. 2018); however, never been examined for oil sands tailings based on the authors' knowledge. Sterilized FFT was used as an abiotic control. Raw FFT was autoclaved for 1 h each day for 8 consecutive days to ensure the control was completely abiotic (i.e., no biological activity).

4.2.2 Experiment

Two dosages of each enzyme (0.5% and 1%) were used. Lysozyme slurry was prepared by adding lysozyme powder to Millipore Ultrapure water (18.2 MΩ cm) to achieve a 0.5% or 1% solids slurry (wt./wt.). Cellulase and protease were purchased in liquid form and were added directly to FFT. Each enzyme dose was added to 800 g of FFT in a glass beaker and mixed with a mechanical agitator at moderate speed (1200 rpm). The initial solids content of the raw FFT (30%) was measured. Experiments were conducted in 1 L glass columns, with 800 mL FFT placed in the column and covered with parafilm and aluminum foil to avoid evaporation of the FFT pore water. All experiments were conducted in triplicate at room temperature. Sterilized FFT was used as an abiotic control; FFT was autoclaved for 1 h per day for 8 successive days. In the abiotic control samples, only the 0.5% dose was examined for each enzyme. The same procedures were repeated for the abiotic control. The experiment period was 12 weeks for all the treatment conditions; this is when the volume of released FFT porewater began to

plateau for all the biotic conditions. The experiment was conducted in the dark by covering the columns with a black colored tarp to prevent light penetration. The tarp was temporarily removed during monitoring mudlines and sampling.

To test the effect of enzymes on microbial activities in the tailings, a parallel experiment was conducted with microcosms (serum bottles with 158 mL total volume and 100 mL working volume) to monitor biogenic methane gas (CH₄) production for each test condition. Each microcosm consisted of 100 mL FFT (treated or untreated FFT); bottles were sealed, crimped, and purged with 80% $N_2/20\%$ CO₂ to maintain methanogenic conditions.

4.2.3 Analysis and calculations

4.2.3.1 Water recovery and solids content

The changes in the FFT mudline were recorded over time for all columns. The mudline represents the interface between the FFT and FFT pore water. Water recovery % (WR) has been used to determine the amount of water recovered from FFT with time (Allam et al. 2021; Siddique et al. 2014; Yu et al. 2018a) after enzyme treatment. More details on the WR calculations can be found elsewhere (Allam et al. 2021; Yu et al. 2018a). For day 0 samples, the FFT samples were centrifuged at 8,000 g for 40 min, and the released porewater was collected and filtered using polytetrafluoroethylene (PFTE) filters (Fisher Scientific; 0.45 μ m), then stored at 4 °C before further analyses. Chemical and physical parameters were measured, including pH, soluble cations and anions, zeta potential, and toxicity for

FFT porewater. The headspace composition of the serum bottles was analyzed to determine the methane production for each condition.

4.2.3.2 Chemical analysis

A Thermo iCAP 6000 series inductively coupled plasma–optical emission spectrometer (ICP-OES, ThermoFisher) was used to analyze FFT pore water for cations including Ca²⁺, Na⁺, K⁺, and Mg²⁺. Dionex ion chromatography (ICS-2100, ThermoFisher) using the AS18 32mM ISO method for Anions was used to analyze the anion concentrations in the FFT pore water (Cl⁻, NO₂⁻, SO₄²⁻, Br⁻, NO₃⁻, PO₄³⁻). Alkalinity and HCO₃⁻ (measured as CaCO₃ mg L⁻¹) were measured using a Mettler Toledo DL53 (Mississauga, Canada) with 0.02 N H₂SO₄ as a titrant. An Accumet® Research AR50 (Fisher Scientific, Lenexa, Kansas) was used to measure pH values. Prior to any analysis and unless otherwise specified, liquid samples were filtered with PFTE filters (Fisher Scientific; 0.22 μ m for cation analysis; all water samples were diluted 10 × using Ultrapure water).

NAFCs comprise a broad class of organic compounds including O_2 compounds, nitrogen-containing species (NO_n and N₂O_n), and sulfur-containing species (O_nS and O_nS₂). O₂-NAFCs compounds were measured using Orbitrap high-resolution mass spectrometry and high-performance liquid chromatography at the University of Saskatchewan Toxicology Centre; the detailed protocols can be found elsewhere (Headley et al. 2011). Briefly, using the liquid-liquid extraction method as described by Ross et al. (2012) and Pereira et al. (2013a), the acidified sample (reduced to pH < 2 or 1 with sulfuric acid or formic acid) was extracted

twice into 200 mL of dichloromethane (DCM). The combined 400 mL of DCM extract was evaporated to 5 mL via rotary evaporation, reduced to dryness with nitrogen gas, then the dry extract was dissolved in 1 mL methanol for chemical analysis via Orbitrap mass spectrometry. The sample could also be reconstituted in 500 μ L of 50:50 ACN/H₂O with 0.1% NH₄OH. Further details are described by Morandi et al. (2015). The methane content was measured with an Agilent 7890A gas chromatograph equipped with a flame ionization detector (GC-FID) and Agilent Technologies HP-5MS column (30 m × 0.25 mm × 0.25 μ m). A typical injection volume was 100 μ L.

4.2.3.3 Ionic strength (I) and Diffuse Double Layer (DDL)

The following formula was used to calculate the ionic strength of the FFT porewater (adapted from Essington, 2004):

 $I = 1/2 \Sigma_{\rm i} (M_{\rm i} Z^2_{\rm i})$

where M_i : the molar concentration of charged species (i); Z_i : charge in the porewater.

The following formula was used to calculate the thickness of the DDL (adapted from Essington, 2004):

$$k^{-1} = (3.042 \times 10^{-8}) / Z I^{0.5}$$

where k^{-1} : the thickness of the DDL (cm); Z: the average mean charge of the counterions (exchangeable cations); and *I*: the ionic strength of the porewater.

4.2.3.4 Zeta potential

The zeta potential was measured using a Zetasizer Nano ZSP (Malvern Instruments, Malvern, UK). Approximately 1 mL of the FFT pore water sample obtained from each column was injected into a folded capillary zeta cell (DTS 1070).

4.2.3.5 Toxicity Bioassay

The Microtox[®] bioassay test (Osprey Scientific, Edmonton, Alberta, Canada) was used to measure the toxicity of the FFT porewater samples to *V. fischeri*. The 81.9% Basic Test protocol was applied with 5 min incubation with the Microtox[®] 500 Analyzer (Azur Environmental time) (Anderson et al. 2011). More details are published elsewhere (Allam et al. 2021; Siddique et al. 2014; Yu et al. 2018a). Previous studies reported 1.0 toxicity unit (TU) or lower as detoxified (Scott et al. 2008; Yu et al. 2018a).

4.2.3.6 Microbial community and diversity

The FastDNATM SPIN Kit for Soil (MP Biomedicals) was used to extract DNA from the untreated and enzyme-treated FFT samples (FFT sample weight \simeq 500 mg); more information can be found elsewhere (Allam et al. 2021). DNA extraction was conducted at the start and end of the experiment for the untreated/control sample: however, for the enzyme-treated samples only the final sample was analyzed (week 12). To determine the total microbial community, universal bacterial primers 341F: 5' CCTACGGGNGGCWGCAG 3' and 805R: 5' GACTACHVGGGTATCTAATCC 3' were used to target the 16S rRNA gene (Zakaria and Dhar 2021a). To determine the methanogen gene copies after enzyme archaeal-specific treatment, both primers [517F: 5' GCYTAAAGSRNCCGTAGC 3' and 909R: 5' TTTCAGYCTTGCGRCCGTAC 3'] and methyl coenzyme-M reductase gene (mcrA) primers (*mcrAf*: 5' GGTGGTGTMGGATTCACACARTAYGCWACAGC 3' and mcrAr: 5' TTCATTGCRTAGTTWGGRTAGTT 3') were used (Lin et al. 2020; Zakaria and Dhar 2021a). The quantitative polymerase chain reaction (qPCR) assay was performed on extracted DNA samples (1 µL) using a Bio-Rad CFX96 optical reaction module conversion for the C1000 Touch thermal cycler. Bio-Rad CFX ManagerTM 3.0 software was used to analyze the amplification data. All standards and samples were analyzed in triplicate. Detailed protocols for preparing the qPCR mixtures can be found elsewhere (Lin et al. 2020).

To determine the microbial community diversity, 16S rRNA gene sequencing was performed by Microbiome Insights (Vancouver, BC, Canada) on the DNA samples. PCR amplification of the V4 region of the 16S rRNA gene in bacteria and archaea primers using the 515F (5' GTGCCAGCMGCCGCGGTAA 3') and 806R (5' GGACTACHVGGGTWTCTAAT 3') was performed. The Illumina PE250 platform (Illumina, CA, USA) was used for amplicon sequencing. More information can be found elsewhere (Cossey et al. 2021).

128

4.3 Results and discussion

4.3.1 Effects of enzymes on FFT dewatering

The WR and the volume of the released water from the FFT were used to assess the effectiveness of enzyme treatment for FFT dewatering (Fig. 4.1a and 4.1b). All three enzymes accelerated the amount of the released water after 12 weeks compared with the untreated control (WR: 12% and 11.5% for 0.5% and 1% cellulase; 8.5% and 6% for 0.5% and 1% protease; 19.8% and 20% for 0.5% and 1% lysozyme; control was 2%); (volume of pore water: 97 and 93 mL for cellulase 0.5% and 1%; 67 and 49 mL for protease 0.5% and 1%; 158 and 159.5 mL for lysozyme 0.5% and 1%; control was 15 mL). Both doses of lysozyme (0.5% and 1%) achieved the highest WR (~ 20%) and highest volume of released pore water (~ 159 mL) after 12 weeks. Despite the rapid initial improvement (i.e., first 2 weeks) for cellulase 0.5% and 1% (which achieved 12.5% and 12% WR, and 95 mL and 90 mL of released water), cellulase was relatively less effective than lysozyme at the end of the experiment. Among the three enzymes, the least dewatering was observed with protease 1% (6% WR and 49 mL of released water after 12 weeks).

As mentioned earlier, enzymes have rarely been investigated for oil sands tailings dewatering. Yu et al. (2018a) recently reported the effectiveness of a commercial enzyme (Ultrazyme) for enhanced dewatering of oil sands tailings. Nevertheless, our findings are consistent with studies by Bonilla et al. (2015) that reported enhanced dewaterability of pulp and paper mill sludge with lysozyme treatment compared with cellulase and protease. Their study suggested that lysozyme could act as a cationic polymer rather than a bio-stimulant. Based on another report by Liu et al. (2018), lysozyme could degrade bacterial cells and extracellular polymeric substances (EPS) matrix in sludge, leading to the release of trapped intracellular water. This mechanism is relevant to our study as EPS and microbial cells were shown to be present in oil sands tailings under methanogenic and nitrate-reducing conditions (Bordenave et al. 2010). Our findings agreed with Wu et al. (2014) who found a 0.5% dose of cellulase or protease is more effective for dewatering paper mill sludge than 1% dose.

Interestingly, Yu et al. (2018a) found that Ultrazyme could stimulate microbial activity in tailings followed by biogenic gas production. The produced biogenic gases (e.g., CH₄, CO₂) provided trapped pore water release pathways, which were apparent from the visible cracks developed within the tailings. In our study, many cracks were also observed in the enzyme-treated tailings samples (see Figure 4.2). Therefore, further experiments were conducted with autoclaved tailings to acquire more insights into the underlying mechanisms. The biotic experiments provided no significant difference between the performance of 0.5% and 1% enzyme dosages, with only a slight increase in the performance of the 0.5%treatment compared with the 1% treatment in terms of WR and volume of pore water released. Therefore, for the abiotic control, only the 0.5% dose for each enzyme was examined. As shown in Figure 4.1c, all autoclaved abiotic samples showed considerably lower dewaterability (\leq 5% WR) than the biotic samples (see Figure 4.1a and 4.1b), suggesting the significance of microbial activity in FFT dewatering. Cellulase and protease treatment of the sterile samples resulted in slightly higher WR than the lysozyme treated and control samples. Thus, these enzymes were still capable of slightly improving the dewatering by a physicochemical mechanism, but the results suggest that the enzymes tested here acted more as bio-stimulants and improved the dewaterability of tailings via enhanced microbial activity (i.e., biogenic methane production). Notably, a few studies previously reported that enzymes (e.g., lysozyme) could increase methane production in conventional anaerobic digestion (Bonilla 2017; Kim et al. 2019; Odnell et al. 2016). Therefore, further assessment of methanogenic activity in microcosms with enzyme-treated tailings was performed.



Figure 4.1. (a) Volume of FFT porewater released (mL); (b) The FFT pore water recovery % (WR) after 2 weeks and at the end of the experiment (12 weeks) for the biotic condition; and (c) The FFT pore water recovery % (WR) after 2 weeks and at the end of the experiment (12 weeks) for the abiotic condition. All columns consisted of 800 mL enzyme-treated or untreated FFT. All columns were sealed with parafilm and aluminum foil. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.



Figure 4.2. Different cracks appeared in the FFT columns treated with cellulase, protease, and lysozyme throughout the experiment compared to the control.

4.3.2 Effects on biogenic methane production

Figure 4.3 shows the effects of enzymes on methane production from FFT. The addition of enzymes (except for protease 1%) resulted in an increase in methane production from FFT. The addition of cellulase 0.5% and 1% led to higher methane production after 12 weeks (1.6–2.4 mmol) than the other two enzymes (i.e., lysozyme 1.2–1.5 mmol, and protease 1.8–0.06 mmol for each dose 0.5% and 1% respectively) and control (0.002 mmol). In contrast, methane production from lysozyme and protease-treated tailings had relatively lower methane production than cellulase. Although lysozyme exhibited the highest effectiveness in dewatering, methane production from lysozyme-treated tailings with either 0.5% and 1% doses

was slightly lower than for either dose of cellulase, (1.2–1.5 mmol vs. 1.6– 2.4 mmol). Both doses of lysozyme resulted in similar methane production and similar dewatering. In contrast, different doses of cellulase and protease resulted in different methane production. Notably, 0.5% cellulase and protease treatments had considerably higher methane production than the 1% treatments, indicating the negative impact of higher enzyme doses on methanogenic activity. Also, the methane production results are similar to the WR, which slightly decreased after increasing the enzyme dose for both cellulase and protease. Nonetheless, it was apparent that methanogenesis might have occurred during the column dewatering tests. Previous reports also suggested the formation of anaerobic conditions in the settled tailings at the bottom of the columns during the dewatering tests (Yu et al. 2018a). Our finding agrees with the literature, showing enzymatic treatment with cellulase, protease, or lysozyme increases methanogenesis (Ehimen et al. 2013; Kim et al. 2019; Odnell et al. 2016; Parawira 2012; Passos et al. 2016). These studies suggested that enhanced methane production in anaerobic bioreactors could be attributed to improved hydrolysis due to enzymatic activity. Particularly, the EPS matrix in sludge flocs could be disintegrated by enzymes (Liu et al. 2018; Parawira 2012). A previous study also reported the presence of EPS in oil sands tailings, particularly under methanogenic and nitrate-reducing environments (Bordenave et al. 2010). As discussed later, enzymes could also increase the richness and diversity of bacterial communities, which might improve the degradation of complex organics in tailings to simple substrates that methanogens could utilize.



Figure 4.3. Temporal changes in methane production throughout the experiment for all enzyme-treated FFT compared with the control/untreated FFT. All bottles consisted of 75 mL enzyme-treated or untreated FFT and 25 mL anaerobic media. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.

4.3.3 Variations in water chemistry

4.3.3.1 Cations and anions

Figures. 4.4a, b, and c illustrate the changes in HCO_3^- , Mg^{2+} , and Ca^{2+} concentrations, respectively, in the FFT pore water for the control and enzyme-treated conditions after 4, 6, 9, and 12 weeks. In the control, the concentrations of the three ions were stable throughout the experiment. For enzyme-treated samples, the concentrations of ions increased until week 4. However, after week 4, the

changes in ions concentrations showed different trends for the different enzymes. For the FFT treated with cellulase 1%, after 4 weeks the Ca²⁺ concentration started to decrease until the end of the experiment. Similarly, the Mg²⁺ concentration reached a plateau from 4–9 weeks and then started to decrease up to week 12. This reduction in Ca²⁺ and Mg²⁺ concentrations could be related to the increase in the pH after 9 weeks to pH 7.6 at week 12. At higher pH, Ca²⁺ and Mg²⁺ can form calcium and magnesium bicarbonate (Siddique et al. 2014). However, the HCO₃⁻ concentration reached a plateau after 4 weeks. Such observations could be attributed to the reduction of bicarbonate to methane by methanogens, as suggested previously in the literature (Cervini-Silva et al. 2020; Yang et al. 2016). In the lysozyme-treated FFT, the concentrations of these ions increased up to week 12 (p-value < 0.05; significant difference).

For most of the enzyme-treated samples, the pH profiles showed declining trends. The methanogenic activity in the FFT might cause a pH reduction below 7.5 in FFT porewater after 4 weeks for the cellulase (0.5 and 1%) and protease (0.5%) treated samples (Figure 4.4d); CO₂ dissolve in FFT porewater and cause pH reduction (Samadi 2019; Siddique et al. 2014; Yu et al. 2018a). The slight increase in the pH after 4 weeks for the cellulase treatment comply with the reduction in Ca^{2+} concentrations in the porewater after 4 weeks. With lysozyme treatment, the pH took longer to reduce below 7.5 (6 weeks); this explains the sharp increase in HCO₃⁻⁷, Ca²⁺, and Mg²⁺ concentrations in the porewater. As discussed earlier, the enzymatic treatment also enhanced methanogenic activity in the tailings in this study. During the ebullition of biogenic gases, the release of CO₂ caused a drop in

the FFT pH below 7.5 and the dissolution of carbonate minerals occurred which increase HCO_3^- and cation (Ca²⁺ and Mg²⁺) concentrations in the FFT pore water (Siddique et al. 2014). The dissolution of carbonate minerals had a direct impact in the increase of ionic strength (will be discussed in the next section). Thus, methanogenic activities from the enzyme-treated samples agreed with the changes in pH and ions concentrations.



Figure 4.4. Change in (a) HCO_3^- , (b) Mg^{2+} , (c) Ca^{2+} , concentrations and (d) pH throughout the experiment for all enzyme-treated FFT compared with the control/ untreated FFT. All columns consisted of 800 mL enzyme-treated or untreated FFT. All columns were sealed with parafilm and aluminum foil. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.

4.3.3.2 Ionic strength, diffuse double layer, and zeta potential

Figure 4.5a shows the changes in ionic strength (*I*) of porewater and the diffuse double layer (DDL) of FFT over the 12 week dewatering test. Due to the increase in cation (Ca²⁺ and Mg²⁺) and anion (HCO₃⁻) concentrations throughout the experiment (Figure 4.4) in the enzyme treatment columns, an increase in the *I* values of the pore water and a decrease in the DDL were observed. In contrast, the control sample showed minimal changes. The highest *I* value was reported for the lysozyme 1% treatment ($I = 0.1 \text{ mol } L^{-1}$) compared with the control ($I = 0.038 \text{ mol } L^{-1}$); p-value < 0.05, (significant difference). Siddique et al. (2014) also previously reported an increase in *I* values due to the dissolution of carbonate minerals.

When comparing different doses of each enzyme, increasing the enzyme dose mostly increased the *I* value and DDL (Figure 4.5a). For instance, increasing the cellulase dose from 0.5% to 1% increased the *I* values (0.055 vs. 0.065 mol L⁻¹). Nevertheless, increasing the enzyme dose to 1% had an adverse impact on methane production and hence on WR from FFT (Figure 4.1). Protease treatment achieved the lowest *I* (0.055 and 0.045 mol L⁻¹) for the 0.5% and 1% doses, respectively. Additionally, the lysozyme 1% treatment resulted in the lowest DDL of clay particles (9.40 × 10⁻⁸ cm), while the DDL was as high as 1.54×10^{-7} cm for the control. These results agree with the *I* results, where *I* has a negative effect on the DDL: increasing the *I* strength of the solution decreases the DDL thickness of the clay particles (Siddique et al. 2014). Similarly, the protease 1% treatment had a higher DDL (1.39 × 10⁻⁷ cm) than the other two enzymes, but it was still lower than the control.



Enzyme Dose (wt/wt%)





Figure 4.5. Changes of the a) ionic strength concentrations and b) the thickness of the diffuse double layer for all enzyme-treated FFT compared with the control/ untreated FFT at the beginning of the experiment (0 weeks) and at the end of the experiment (12 weeks). All columns consisted of 800 mL enzyme-treated or untreated FFT. All columns were sealed with parafilm and aluminum foil. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.

The enzymatic treatment resulted in lower DDL thickness and shifted zeta potential to less negative values at the end of the experiment (Figure 4.6). Lysozyme treatment significantly increased the zeta potential (-34.7 to -14.8 mV). However, there was a minor difference between the initial and final zeta potential values for the control (Figure 4.6). These findings agree with a previous study that suggested an increase in the Zeta potential of paper mill sludge from -44.2 mV to -26.1 mV due to lysozyme addition (Bonilla 2017). The changes in Zeta potential can be related to charge neutralization, facilitating effective flocculation. For instance, a shift of zeta potential to less negative values might be because the lysozyme carries a positive charge at pH \sim 7 (Bonilla 2017), neutralizing the negative charge surrounding the clay particles in FFT. Thus, the key reason for the decrease in the DDL thickness is the charge neutralization between the clay surface and positively charged divalent cations such as Ca²⁺ and Mg²⁺. Also, the electrostatic repulsion between the clay particles decreased at a high I value, followed by a decrease in DDL thickness. The reduction in the DDL thickness for enzyme-treated samples suggested that enzymes facilitated clay flocculation and improved tailings consolidation.



Figure 4.6 Variations of the zeta potential for all enzyme-treated FFT compared with the control/ untreated FFT at the beginning of the experiment (0 weeks) and at the end of the experiment (12 weeks). All columns consisted of 800 mL enzyme-treated or untreated FFT. All columns were sealed with parafilm and aluminum foil. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.

4.3.4 Naphthenic acid fractions

To determine the efficiency of enzymatic treatment of FFT for the degradation of naphthenic acids fractions (NAFCs), the concentration of the classical O₂ species in the released porewater were determined using Orbitrap–MS. NAFCs were measured in untreated columns at the beginning of the experiment, and in all treated and untreated columns at the end of experiment. In this study, we focused only on the classical O₂-NAFC because these are considered the main contributors to acute toxicity in OSPW (Balaberda and Ulrich 2021; Hughes et al. 2017; Meshref et al. 2017b) and FFT porewater. Figure 4.7a and 4.7b illustrate the concentrations of

total O₂-NAFC species in the control sample (untreated), indicating a slight decrease in the O₂-NAFC of approximately 5% within 90 d. This result agrees with the literature, which reports slow degradation of NAFCs by indigenous microbial communities (e.g., ~16% decrease in 5 years) (Allen 2008b; Schoof 2015; Wu et al. 2019). When comparing the two dosage levels (low versus high dose) for both cellulase and protease, the 1% dose resulted in higher O₂-NAFC residuals than lower doses (0.5%). For lysozyme, increasing the dose from 0.5% to 1% resulted in higher degradation of O₂-NAFC. Overall, the highest decreases in O₂-NAFC concentration were 344 and 243 ng mL⁻¹ achieved by the lysozyme 0.5% and 1% treatments, respectively. The lowest reduction in the total O₂-NAFC species was observed with the protease 1% treatment after 12 weeks (1138 ng mL⁻¹).

One possible reason for the reduction in O₂-NAFC species in the lysozymetreated FFT is the high amount of FFT WR after 12 weeks compared with the other two enzymes (Figure 4.1a). This might dilute the NAFC concentration in the FFT porewater compared with the control. Another reason could be the ability of lysozyme to degrade NAFCs; Guo and Xu (2011) reported previously that lysozyme could degrade organic molecules as well as lysing the bacterial cell wall. Additionally, the composition of NAFCs reported by Headley and McMartin, (2004) revealed that NAFCs have smaller amounts of acyclic aliphatic / paraffinic or fatty acids together with alkyl-substituted cycloaliphatic carboxylic acids. In the same sense, Peng et al. (2016) suggested that some NAFC structures are similar to fatty acids. Previous studies reported a reduction of volatile fatty acids after enzymatic treatment (Roman et al. 2006).



Figure 4.7. The concentrations of NAFCs for all enzyme-treated FFT compared with the control/untreated FFT. (a) Control at Day 0, (b) Control Day 90, (c) cellulase 0.5%, (d) cellulase 1%, (e) protease 0.5%, (f) protease 1%, (g) lysozyme 0.5% and (h) lysozyme 1%. All columns consisted of 800 mL enzyme-treated or untreated FFT. All columns were sealed with parafilm and aluminum foil. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.

4.3.5 Toxicity

Figure 4.8a summarizes the influence of enzyme treatments on the released FFT porewater toxicity to V. fischeri throughout the experiment at weeks 0, 4, 6, 9, and 12. Figure 4.8b summarizes the results from the abiotic control. Overall, cellulase 0.5% was the only treatment which resulted in a non-toxic effect for the released porewater after 12 weeks (0.25 TU) compared with other enzymes or the control (1.05 TU)TU). However, treatment with lysozyme resulted in the highest toxic effect for the FFT porewater to V. fischeri: 27 TU and 32 TU for the 0.5% and 1% doses, respectively. For cellulase, increasing the dose to 1% increased the toxicity of the released water to 12 TU. For protease, although both doses resulted in porewater that was considered toxic (i.e., TU > 1), the 0.5% dose resulted in lower toxicity (2.5 TU) compared with the 1% dose (6.8 TU). For lysozyme, the toxicity increased sharply after 6 weeks, which could be ascribed to the release of organic fractions from the FFT. These findings are in agreement with Lin et al. (2019) who reported that lysozyme destroyed the microbial cell walls and released the substances inside the cells. In contrast, based on the O2-NAFCs results, lysozyme-treated FFT had the lowest O₂-NAFCs after 12 weeks compared with the control and cellulase treatment. Many studies have reported that NAFC species, including classical and oxidized (Ox-NAFCs) species, are the main contributors to toxicity in OSPW (Grewer et al. 2010; Headley et al. 2009). In our study, only the classical NAFCs were measured. Other studies have reported that other organic compounds, such as alkylated phenols, are toxic compounds in OSPW (Hargesheimer et al. 1984). With cellulase 0.5% treatment, the toxicity reduction after 12 weeks might be attributed to the reduction in the lower molecular weight constituents compared with the control. This

observation is in agreement with Frank et al. (2008) who reported that the toxicity of the NAFCs toward V. fischeri correlated with the concentration of lower molecular weight constituents. However, for abiotic controls, all the enzyme treatments resulted in no toxicity to V. fischeri at the end of the experiment (Fig. 4.8b), indicating that the toxicity increase in the biotic columns is related to the distortion of microbial cell walls and release of the substances from inside the cells. Despite the fact that V. fischeri (Microtox assay[®]) is currently used as a standard method for measuring OSPW toxicity (Meshref et al. 2017a; Pourrezaei et al. 2014; Zubot et al. 2012), other organisms such as Pseudomonas sp., algae, invertebrates (e.g., Daphnia magna), and fish can also be used (Barceló et al. 2020). The level of toxicity may be different for different organisms, leading to different sensitivities (Bartlett et al. 2017) and minor inconsistencies between the toxicity results and NAFC concentrations and CH₄ production. Moreover, this study focused on NAFCs (O_2^- species), which are often considered the primary source of toxicity in OSPW (Balaberda and Ulrich 2021; Meshref et al. 2017a). In contrast, a few studies reported that the non-naphthenic acids $(O_2^+$ species) could also be toxic (Pereira et al. 2013b). As discussed later, lysozyme could adversely impact the microbial population while resulting in better NAFC degradation. Thus, the release of non-naphthenic acids $(O_2^+ \text{ species})$ in response to enzymatic treatment should be further investigated.



Figure 4.8. The effect of the released pore water from (a) biotic and (b) abiotic enzymatic treatment at various enzyme doses on the toxicity of *V. fisheri*. The impact of toxicity was monitored throughout the experiment for all enzyme-treated FFT compared with the control/untreated FFT. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.

4.3.6 Microbial communities

In this study, the quantitative and qualitative characteristics of microbial communities were analyzed to evaluate the impact of enzymes on microbial communities. The untreated FFT sample from the beginning of the test served as the control (i.e., initial). After 12 weeks, FFT samples were collected from all columns, including untreated and enzyme-treated FFT samples. Figure 4.9 shows the qPCR results for total microorganisms using the 16S rRNA gene, archaeal gene copies, and *mcr*A. As reported in previous studies, *mcrA* gene copies are considered a biomarker for hydrogenotrophic methanogenesis (Wilkins et al. 2015; Zakaria and Dhar 2021b). The week 0 results for the untreated/control sample represent the indigenous microorganisms that existed in the raw tailings sample before treatment.

With the cellulase 0.5% and 1% treatments, the total microorganisms increased from 7.60×10^6 cell g⁻¹ (measured at the control week 0) to 1.53×10^7 cell g⁻¹ and 2.90×10^7 cell g⁻¹, respectively. The *mcr*A gene also increased significantly from 9.84×10^3 cell g⁻¹ measured at the control week 0 to 3.72×10^4 cell g⁻¹ and 6.84×10^4 cell g⁻¹ for the cellulase 0.5% and 1% treatments, respectively. These results agree with the methane production results where high CH₄ was observed after 12 weeks for both cellulase doses. Also, as discussed earlier in the toxicity results, the cellulase 0.5% treatment resulted in no toxicity to *V. fischeri*, agreeing with the increase in the qPCR results observed from cellulase-treated FFT. In contrast, the cellulase 1% treatment resulted in high toxicity after 12 weeks (12 TU) despite the increase in the microbial community. Yu et al. (2018b) reported the same observation where the biogenic CO₂ increased and the bacterial density decreased by 70%. For the lysozyme-treated FFT, a slight change in the microbial community was observed after 12 weeks; the total microorganisms decreased to 5.60 $\times 10^{6}$ cell g⁻¹ and 5.11 $\times 10^{6}$ cell g⁻¹ for the lysozyme 0.5% and 1% treatments, respectively, and to 5.00 $\times 10^{3}$ cell g⁻¹ and 3.05 $\times 10^{3}$ cell g⁻¹ for the methanogen *mcr*A gene, respectively. The protease 1% treatment had the highest reduction in the gene copies of 16S rRNA and *mcr*A. This agrees with the CH₄ production results where the protease 1% treatment resulted in the lowest CH₄ production. Nevertheless, despite low methane production, the control treatment did not have a substantial decrease in gene copies, suggesting that the differences in methane production.



Figure 4.9. Change in gene copies g^{-1} FFT at the beginning and end of the experiment for the control conditions and at week 12 for all enzyme-treated conditions. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) ± one standard deviation.

With regards to the microbial community diversity, it presents a diverse community of microbial species dominated by Firmicutes and Bacteroidetes at the phylum level (Figure 4.10). During the start-up (Week 0), Proteobacteria dominated in the control but their relative abundance significantly decreased after 12 weeks (3.9%). Firmicutes was observed as most dominant bacterial phylum at week 12 (66.9%) along with Bacteroidetes (20.3%). These bacterial phyla were previously identified in oil refinery activated sludge (Misiti et al. 2013) and OSPW (Wilson et al. 2016). For all the enzymes, increasing the enzyme dose from 0.5% to 1% directly impacted the relative abundance of the microbial community after 12 weeks (Figure 4.10). Proteobacteria and Actinobacteria have widely been observed in OSPW, despite variations in abundance (Johnson et al. 2012; Penner and Foght 2010); (Foght et al. 2017; Ramos-Padrón et al. 2011; Siddique et al. 2018). Some other important phyla, such as Acidobacteria, Bacteroidetes, and Firmicutes, are reported to be efficient NAFC-degrading microbes in bioreactors treating OSPW (Zhang et al. 2020).

At the class level (Figure 4.10 a), a complex and diverse microbial community can be seen more prominently. The control week 0 sample is mainly dominated by Betaproteobacteria (66.5%) and Deltaproteobacteria (17.2%) at the class level. In the control sample at week 12, three different classes pointedly appeared: Bacilli (44.4%), Clostridia (15.4%), and Bacteroidia (13.8%). A diverse microbial community facilitates synergistic relationships between species, as one species may be able to degrade metabolites more effectively or use substrates more efficiently (Demeter et al. 2015; Folwell et al. 2020; Yu et al. 2019). In the cellulase 0.5%

treatments, Bacteroidia (39.7%) and Clostridia (26.8%) were among the dominant classes. In the cellulase 1% treatment, the relative abundance of Bacteroidia dropped to 3.7%. However, Clostridia (28.3%) and Negativicutes (27%) prevailed. Through the use of cellulases, organized into cellulosomes, *Clostridium* is able to hydrolyze cellulose efficiently (Gold and Martin 2007; Zverlov et al. 2005). Gammaproteobacteria were also prominent in this sample, accounting for 2.66% for the cellulase 1% treatment. In aerobic cultures derived from froth treatment tailings, Gammaproteobacteria and other taxa played a critical role in hydrocarbon degradation (Foght et al. 2014). Among the well-known methanotrophic aerobic bacteria, Gammaproteobacteria are widespread (Danilova and Dedysh 2014; Op den Camp et al. 2009). Methane oxidation by these bacteria results in the release of methanol and formaldehyde, which are the main sources of nutrients for other organisms (Op den Camp et al. 2009; Saidi-Mehrabad et al. 2013). In the protease 0.5% and 1% treatments, the relative abundance of Clostridia decreased dramatically from 58.3% to 9.3%. Negativicutes (16.6% to 0.4%) were also considerably reduced in the protease 0.5% and 1% treatments. There was a noticeable increase in the relative abundance of Actinobacteria (3.5% to 30.8%) (Figure 4.10). With lysozyme 0.5% and 1% treatment, the microbial community remained consistent, with the relative abundance of Bacteroidia increasing (38% to 69.6%) and Clostridia decreasing (34.9% to 12.1%). Several studies reported that the presence of symbiotic *Clostridium* and *Bacteroides* consortia enhances methane production (Tukanghan et al. 2021). It is well known that both of these genera produce xylanase and cellulase (Robert et al. 2007; Thomas et al. 2014).



Figure 4.10. Change in bacterial-phylum and bacterial-class at the beginning and end of the experiment for the control conditions and at week 12 for all the enzyme-treated conditions. The untreated FFT represents a 0 ppm treatment and acts as the control experiment. Results illustrated are an average (n = 3 for all treatment conditions) \pm one standard deviation.

In the archaeal community, Methanomicrobia was the most abundant enriched class (Figure 4.11 a). However, Methanobacteria (61.1%) and Methanomicrobia (36.8%) were found in the lysozyme 1% treatment. Both Methanobacteria and Methanomicrobia are methanogens from Euryarchaota and get their energy by converting specific substrates to methane gas (Choi et al. 2018; Matsuda and Ohtsuki 2016; Whitman et al. 2006). Methanomicrobia usually function as the main hydrogenotrophic/methylotrophic methanogens, according to several studies (Figure 4.11).

Among all samples, Methanolinea and Methanosaeta were significantly more abundant than any other archaeal genus. Protease treatments (0.5% and 1%) were mostly dominated by Methanosaeta (56.9% and 71.9%). The cellulase 0.5% treatment was dominated by Methanolinea (60.7%) and Methanosaeta (36.9%). In response to a shift in concentration, Methanosaeta (56.9%) became more abundant with the cellulase 1% treatment. The genus Methanolinea belongs to the newly identified Methanoregulaceae family and is hydrogenotrophic (Oren 2014). In the lysozyme 0.5% and 1% treatments, *Methanolinea* accounted for 45.7% and 61.6% of total methanogen abundance. Furthermore, Methanosaeta were found with an abundance of 44.6% and 33.8% in the same samples (Figure 4.11). The genus Methanosaeta belongs to the class Methanomicrobia and was previously known as Methanothrix (Anderson et al. 2003; Smith and Ingram-Smith 2007). Methanosaeta is one of the most active methanogens found in wetlands, and is responsible for large amounts of global methane production. In residual oil, palmitic acid, and hexadecane amended enrichments, the acetotrophic methanogen Methanosaeta was the most

abundant (Fowler et al. 2016; Liang et al. 2015). Clostridiales, Bacteroidia, Methanobacteria, and Methanomicrobia act synergistically in anaerobic digestion to produce hydrogen as well as methane (Vanwonterghem et al. 2014). Additionally, *Methanosaeta* and *Methanolinea* were abundant in communities that had syntrophic interactions with *Bacteroides* and *Clostridium* (Jaenicke et al. 2011; Wang et al. 2017).

We evaluated the alpha diversity of microbial communities (Table 4.1 and 4.2). The observed OTU, Chao1, and Shannon indices were relatively higher in the cellulase 1% and protease 0.5% and 1% treatments than in the control, reflecting their positive impact on the richness and diversity of these bacterial communities. As evidenced by the Chao1 index, methanogens were found in higher abundance in the cellulase 0.5%, protease 0.5%, and lysozyme 0.5% and 1% treatments, and coincided with high methane production under these conditions. Comparatively, in the enzymatically treated samples, the bacterial community was far more abundant and diverse than the archaeal community. In spite of this, their mutual interaction has vital benefits for accelerating oil sand tailings dewatering.





Table 4.1. The bacterial diversity index for the control conditions at week 0 and week12 for all enzymes treated conditions. The untreated FFT represents a 0 ppmtreatment and acts as the control experiment.

Bacterial diversity Index					
Samples	Observed OTU	Chao1	Shannon	Simpson	Read count
Control Week 0	155	168.6	3.26	0.91	12984
Control Week 12	150	167.6	2.82	0.89	11552
Cellulase 0.5%	131	166.0	2.88	0.89	9211
Cellulase 1%	162	170.5	3.29	0.93	10848
Protease 0.5%	158	183.1	3.29	0.93	9126
Protease 1%	162	162.9	3.38	0.92	14338
Lysozyme 0.5%	156	165.6	2.92	0.90	14549
Lysozyme 1%	150	157.7	2.64	0.88	23465
Table 4.2. The archaeal diversity index for the control conditions at week 0 and week

 12 for all enzyme-treated conditions. The untreated FFT represents a 0 ppm treatment

 and acts as the control experiment.

Archaeal diversity Index					
Samples	Observed OTU	Chao1	Shannon	Simpson	Read count
Control Week 0	7	7	0.63	0.35	2910
Control Week 12	8	8	1.16	0.67	2452
Cellulase 0.5%	10	10.5	0.76	0.49	5075
Cellulase 1%	6	6	0.82	0.52	369
Protease 0.5%	9	9.5	0.76	0.43	884
Protease 1%	7	7	0.71	0.38	1254
Lysozyme 0.5%	9	9	0.97	0.58	5619
Lysozyme 1%	9	9.5	0.75	0.48	1196

4.3.7 Dewatering mechanisms

This study demonstrated the feasibility of using enzymes for enhanced dewatering of oil sands tailings. Based on the results, improved FFT dewatering involved multiple mechanisms. First, enzymes could stimulate the biogenic CH₄ production, which provided release pathways for trapped pore water in tailings. Second, interactions between enzymes and clay particles enabled charge neutralization, facilitating flocculation.

Our results showed that the addition of enzymes increased CH₄ production (Figure 4.3). Moreover, columns amended with enzymes exhibited visible cracks developed within the tailings. Molecular biology analysis suggested that enzymes increased the richness and diversity of bacterial communities. Thus, enhanced bacterial activities could possibly improve biodegradation of tailings organics to simple substrates, which subsequently enhanced methanogenic activity. Nonetheless, as suggested in the previous literature, the disintegration of the EPS matrix by enzymes could also contribute to enhanced methanogenesis, which warrants further investigation.

During the release of the biogenic gases, CO₂ dissolved in the treated FFT pore water and the pH dropped below 7.5. This drop in the pH resulted in the dissolution of carbonate minerals. The dissolution of minerals can increase the ionic strength (Siddique et al. 2014); in which our results confirmed this notion. Thus, the addition of enzymes could increase ionic strength and decrease the DDL thickness (Figure 4.5). Increased ionic strength in the pore fluid can promote inter-particle attraction, leading to enhanced dewatering (Mishra et al. 2019). Zeta potential changes to less negative values (Figure 4.6) indicated that the shrinkage of the DDL was attributed to the charge neutralization of the clay surface. Negatively charged clay particles can attract ions of opposite charge. Based on the literature, enzymes like lysozyme can carry positive charges (Bonilla 2017), potentially interacting with negatively charged clay particles. A schematic representation of the above mechanisms associated with enhanced FFT dewatering aided by enzymes is provided in Figure 4.12.



Figure 4.12. Proposed mechanisms of tailings dewatering aided by enzymes.

4.4 Conclusions

In summary, the results of this study demonstrated that enzyme addition could increase the methanogenic activity, which consequently influences the dewatering process. All enzymes (except for protease 1%) increased methane production. The ebullition of CH₄ caused cracks in the FFT, which created pathways for the trapped water to be released. Additionally, during the release of CH₄, the pH of the FFT decreased to <7.5, leading to the dissolution of carbonate minerals, which consequently increased ionic strength. Lysozyme was the best for improving FFT dewatering among the three enzymes. Lysozyme treatment provided the highest WR of 20%, highest increase in ionic strength and highest decrease in the DDL thickness,

following the highest shift of Zeta potential to less negative values (-34.7 to -14.8 mV). Thus, lysozyme was the most effective in charge neutralization of clay particles, which could be attributed to the positive charge carried by lysozyme under pH conditions of tailings. Both doses of lysozyme (0.5% and 1%) reduced the NAFC concentrations in pore water. However, the released pore water from the lysozyme treatment had high toxicity compared with the other enzyme treatments. Further investigation is still warranted to investigate the release of organic fractions from FFT and their associated potential risks.

4.5 References

- AER. 2019. State of fluid tailings management for mineable oil sands, 2018. Alberta Energy Regulator: Calgary, AB, Canada.
- Allam NE, Romaniuk N, Tate M, Meshref MNA, Dhar BR, Ulrich AC. 2021. Impact of lime treatment on tailings dewatering and cap water quality under an oil sands end pit lake scenario. Science of The Total Environment. 781:146699.
- Allen EW. 2008a. Process water treatment in canada's oil sands industry: Ii. A review of emerging technologies. Journal of Environmental Engineering and Science. 7(5):499-524.
- Allen EW. 2008b. Process water treatment in canada's oil sands industry: I. Target pollutants and treatment objectives. Journal of Environmental Engineering and Science. 7(2):123-138.
- Anderson JC, Wiseman SB, Wang N, Moustafa A, Perez-Estrada L, Gamal El-Din M, Martin JW, Liber K, Giesy JP. 2011. Effectiveness of ozonation treatment in eliminating toxicity of oil sands process-affected water to chironomus dilutus. Environ Sci Technol. 46(1):486-493.
- Anderson K, Sallis P, Uyanik S. 2003. Anaerobic treatment processes. V: The handbook of water and wastewater microbiology. Mara d., horan n. London, Academic Press.
- Ayol A, Dentel SK. 2005. Enzymatic treatment effects on dewaterability of anaerobically digested biosolids-ii: Laboratory characterizations of drainability and filterability. Process Biochemistry. 40(7):2435-2442.

- Balaberda AL, Ulrich AC. 2021. Persulfate oxidation coupled with biodegradation by pseudomonas fluorescens enhances naphthenic acid remediation and toxicity reduction. Microorganisms. 9(7).
- Barceló D, Žonja B, Ginebreda A. 2020. Toxicity tests in wastewater and drinking water treatment processes: A complementary assessment tool to be on your radar. Journal of Environmental Chemical Engineering. 8(5):104262.
- Bartlett AJ, Frank RA, Gillis PL, Parrott JL, Marentette JR, Brown LR, Hooey T, Vanderveen R, McInnis R, Brunswick P. 2017. Toxicity of naphthenic acids to invertebrates: Extracts from oil sands process-affected water versus commercial mixtures. Environmental pollution. 227:271-279.
- Bonilla S. 2017. Protein-based conditioners for enhancing biosludge dewaterability.
- Bonilla S, Tran H, Allen DG. 2015. Enhancing pulp and paper mill biosludge dewaterability using enzymes. Water research. 68:692-700.
- Bordenave S, Kostenko V, Dutkoski M, Grigoryan A, Martinuzzi RJ, Voordouw G. 2010. Relation between the activity of anaerobic microbial populations in oil sands tailings ponds and the sedimentation of tailings. Chemosphere. 81(5):663-668.
- Cervini-Silva J, Ruíz G, Hernández JM, Palacios E, Balmaseda J, Orta M, Hernández S. 2020. Coupled reduction of bicarbonate to methane and generation/sequestration of carbon dioxide in natural basaltic-and clay-matrices at 25° c≤ t≤ 100° c. Applied Clay Science. 198:105844.

- Choi J-M, Han S-K, Lee C-Y. 2018. Enhancement of methane production in anaerobic digestion of sewage sludge by thermal hydrolysis pretreatment. Bioresource technology. 259:207-213.
- Clemente JS, Fedorak PM. 2005. A review of the occurrence, analyses, toxicity, and biodegradation of naphthenic acids. Chemosphere. 60(5):585-600.
- Cossey HL, Anwar MN, Kuznetsov PV, Ulrich AC. 2021. Biofilms for turbidity mitigation in oil sands end pit lakes. Microorganisms. 9(7):1443.
- Danilova OV, Dedysh SN. 2014. Abundance and diversity of methanotrophic gammaproteobacteria in northern wetlands. Microbiology. 83(1):67-76.
- Demeter MA, Lemire J, Yue G, Ceri H, Turner RJ. 2015. Culturing oil sands microbes as mixed species communities enhances ex situ model naphthenic acid degradation. Frontiers in microbiology. 6:936.
- Devenny DW. 2010. A screening study of oil sand tailings technologies and practices. Prepared for Alberta Energy Research Institute.
- Dursun D, Turkmen M, Abu-Orf M, Dentel SK. 2006. Enhanced sludge conditioning by enzyme pre-treatment: Comparison of laboratory and pilot scale dewatering results. Water science and technology. 54(5):33-41.
- Ehimen EA, Holm-Nielsen JB, Poulsen M, Boelsmand JE. 2013. Influence of different pre-treatment routes on the anaerobic digestion of a filamentous algae. Renewable energy. 50:476-480.
- ERCB. 2012. Alberta's energy reserves 2011 and supply/demand outlook 2012-2021. Government of alberta, c., ab, canada, 2012 (ed).

- Foght JM, Gieg LM, Siddique T. 2017. The microbiology of oil sands tailings: Past, present, future. Fems Microbiology Ecology. 93(5):22.
- Foght JM, Siddique T, Gieg LM. 2014. Protocols for handling, storing, and cultivating oil sands tailings ponds materials for microbial and molecular biological study. Hydrocarbon and lipid microbiology protocols. Springer. p. 175-197.
- Folwell BD, McGenity TJ, Whitby C. 2020. Diamondoids are not forever: Microbial biotransformation of diamondoid carboxylic acids. Microbial biotechnology. 13(2):495-508.
- Fowler SJ, Toth CRA, Gieg LM. 2016. Community structure in methanogenic enrichments provides insight into syntrophic interactions in hydrocarbonimpacted environments. Frontiers in microbiology. 7:562.
- Frank RA, Kavanagh R, Burnison BK, Arsenault G, Headley JV, Peru KM, Van Der Kraak G, Solomon KR. 2008. Toxicity assessment of collected fractions from an extracted naphthenic acid mixture. Chemosphere. 72(9):1309-1314.
- Gold ND, Martin VJJ. 2007. Global view of the clostridium thermocellum cellulosome revealed by quantitative proteomic analysis. Journal of bacteriology. 189(19):6787-6795.
- Government of Alberta. 2015. Lower athabasca region : Tailings management framework for the mineable athabasca oil sands.

- Grewer DM, Young RF, Whittal RM, Fedorak PM. 2010. Naphthenic acids and other acid-extractables in water samples from alberta: What is being measured? Science of the Total Environment. 408(23):5997-6010.
- Gumfekar SP, Rooney TR, Hutchinson RA, Soares JBP. 2017. Dewatering oil sands tailings with degradable polymer flocculants. ACS Applied Materials & Interfaces. 9(41):36290-36300.
- Guo J, Xu Y. 2011. Review of enzymatic sludge hydrolysis. Journal of Bioremediation and Biodegradation. 2(5).
- Hargesheimer EE, Coutts RT, MacKinnon MD. 1984. Characterization of simple phenols in oil sands extraction-process water. Environmental Technology. 5(1-11):433-440.
- Headley JV, Barrow MP, Peru KM, Fahlman B, Frank RA, Bickerton G, McMaster ME, Parrott J, Hewitt LM. 2011. Preliminary fingerprinting of athabasca oil sands polar organics in environmental samples using electrospray ionization fourier transform ion cyclotron resonance mass spectrometry. Rapid Communications in Mass Spectrometry. 25(13):1899-1909.
- Headley JV, McMartin DW. 2004. A review of the occurrence and fate of naphthenic acids in aquatic environments. Journal of Environmental Science and Health, Part A. 39(8):1989-2010.
- Headley JV, Peru KM, Barrow MP. 2009. Mass spectrometric characterization of naphthenic acids in environmental samples: A review. Mass spectrometry reviews. 28(1):121-134.

- Holowenko FM, MacKinnon MD, Fedorak PM. 2000. Methanogens and sulfatereducing bacteria in oil sands fine tailings waste. Canadian Journal of Microbiology. 46(10):927-937.
- Hughes SA, Mahaffey A, Shore B, Baker J, Kilgour B, Brown C, Peru KM, Headley JV, Bailey HC. 2017. Using ultrahigh-resolution mass spectrometry and toxicity identification techniques to characterize the toxicity of oil sands process-affected water: The case for classical naphthenic acids. Environmental toxicology and chemistry. 36(11):3148-3157.
- Humphries M, Library of Congress. Congressional Research Service. 2008. North american oil sands history of development, prospects for the future. Updated January 17, 2008 ed. [Washington, D.C.]: Congressional Research Service. http://fas.org/sgp/crs/misc/RL34258.pdf.
 http://www.scribd.com/doc/2402059/CRS-Report-for-US-Congress-NA-

Oil-Sands-Jan-2008.

- Jaenicke S, Ander C, Bekel T, Bisdorf R, Dröge M, Gartemann K-H, Jünemann S, Kaiser O, Krause L, Tille F. 2011. Comparative and joint analysis of two metagenomic datasets from a biogas fermenter obtained by 454pyrosequencing. PloS one. 6(1):e14519.
- Johnson RJ, West CE, Swaih AM, Folwell BD, Smith BE, Rowland SJ, Whitby C. 2012. Aerobic biotransformation of alkyl branched aromatic alkanoic naphthenic acids via two different pathways by a new isolate of mycobacterium. Environmental microbiology. 14(4):872-882.

- Kaminsky HAW, Etsell TH, Ivey DG, Omotoso O. 2009. Distribution of clay minerals in the process streams produced by the extraction of bitumen from athabasca oil sands. Can J Chem Eng. 87(1):85-93.
- Kasperski KL, Mikula RJ. 2011. Waste streams of mined oil sands: Characteristics and remediation. Elements. 7(6):387-392.
- Kim S, Woo S-G, Lee J, Lee D-H, Hwang S. 2019. Evaluation of feasibility of using the bacteriophage t4 lysozyme to improve the hydrolysis and biochemical methane potential of secondary sludge. Energies. 12(19):3644.
- Liang B, Wang L-Y, Mbadinga SM, Liu J-F, Yang S-Z, Gu J-D, Mu B-Z. 2015. Anaerolineaceae and methanosaeta turned to be the dominant microorganisms in alkanes-dependent methanogenic culture after long-term of incubation. Amb Express. 5(1):1-13.
- Lin F, Zhu X, Luo Y, Yu P, Liu M. 2019. Improvement of activated sludge dewatering properties using green conditioners: Chitosan hydrochloride and lysozyme. RSC advances. 9(12):6936-6945.
- Lin L, Chowdhury B, Zakaria BS, Dhar BR. 2020. Temperature-dependent (20–55°
 c) electrocatalytic characteristics during ethanol/propionate degradation by methanogenic communities grown on conductive carbon fibers. Chemical Engineering Journal. 391:123566.
- Liu W, Zhong X, Cheng L, Wang J, Sun Y, Deng Y, Zhang Z. 2018. Cellular and compositional insight into the sludge dewatering process using enzyme treatment. Environmental Science and Pollution Research. 25(29):28942-28953.

- Mackinnon M, Zubot W. 2013. Chapter 10: Water management. Kingsley Knowledge Publishing: Alberta.
- Masliyah JH, Xu Z, Czarnecki JA. 2011. Handbook on theory and practice of bitumen recovery from athabasca oil sands. Kingsley Knowledge Pub.
- Matsuda S, Ohtsuki T. 2016. Establishment and analysis of microbial communities capable of producing methane from grass waste at extremely high c/n ratio. Int J New Technol Res. 2:81-86.
- McQueen AD, Kinley CM, Hendrikse M, Gaspari DP, Calomeni AJ, Iwinski KJ, Castle JW, Haakensen MC, Peru KM, Headley JV et al. 2017. A risk-based approach for identifying constituents of concern in oil sands process-affected water from the athabasca oil sands region. Chemosphere. 173:340-350.
- Meshref MNA, Chelme-Ayala P, Gamal El-Din M. 2017a. Fate and abundance of classical and heteroatomic naphthenic acid species after advanced oxidation processes: Insights and indicators of transformation and degradation. Water Research. 125:62-71.
- Meshref MNA, Klamerth N, Islam MS, McPhedran KN, Gamal El-Din M. 2017b. Understanding the similarities and differences between ozone and peroxone in the degradation of naphthenic acids: Comparative performance for potential treatment. Chemosphere. 180:149-159.
- Miles SM, Asiedu E, Balaberda A-l, Ulrich AC. 2020. Oil sands process affected water sourced trichoderma harzianum demonstrates capacity for mycoremediation of naphthenic acid fraction compounds. Chemosphere. 258:127281.

- Mishra PN, Scheuermann A, Bore T, Li L. 2019. Salinity effects on soil shrinkage characteristic curves of fine-grained geomaterials. Journal of Rock Mechanics and Geotechnical Engineering. 11(1):181-191.
- Misiti T, Tezel U, Pavlostathis SG. 2013. Fate and effect of naphthenic acids on oil refinery activated sludge wastewater treatment systems. Water Research. 47(1):449-460.
- Morandi GD, Wiseman SB, Pereira A, Mankidy R, Gault IGM, Martin JW, Giesy JP. 2015. Effects-directed analysis of dissolved organic compounds in oil sands process-affected water. Environmental science & technology. 49(20):12395-12404.
- Odnell A, Recktenwald M, Stensén K, Jonsson B-H, Karlsson M. 2016. Activity, life time and effect of hydrolytic enzymes for enhanced biogas production from sludge anaerobic digestion. Water Research. 103:462-471.
- Op den Camp HJM, Islam T, Stott MB, Harhangi HR, Hynes A, Schouten S, Jetten MSM, Birkeland NK, Pol A, Dunfield PF. 2009. Environmental, genomic and taxonomic perspectives on methanotrophic verrucomicrobia. Environmental Microbiology Reports. 1(5):293-306.
- Oren A. 2014. The family methanoregulaceae. The Prokaryotes: Other Major Lineages of Bacteria and the Archaea.253-258.
- Owolagba J, Azam S. 2015. Effect of seasonal weather variations on the desiccation behavior of treated oil sand fine tailings. Environmental Earth Sciences. 74(2):1711-1717.

- Parawira W. 2012. Enzyme research and applications in biotechnological intensification of biogas production. Critical Reviews in Biotechnology. 32(2):172-186.
- Parmar N, Singh A, Ward OP. 2001. Enzyme treatment to reduce solids and improve settling of sewage sludge. Journal of industrial microbiology and biotechnology. 26(6):383-386.
- Passos F, Hom-Diaz A, Blanquez P, Vicent T, Ferrer I. 2016. Improving biogas production from microalgae by enzymatic pretreatment. Bioresource Technology. 199:347-351.
- Peng H, Sun J, Alharbi HA, Jones PD, Giesy JP, Wiseman S. 2016. Peroxisome proliferator-activated receptor γ is a sensitive target for oil sands processaffected water: Effects on adipogenesis and identification of ligands. Environmental science & technology. 50(14):7816-7824.
- Penner TJ, Foght JM. 2010. Mature fine tailings from oil sands processing harbour diverse methanogenic communities. Canadian journal of microbiology. 56(6):459-470.
- Pereira AS, Bhattacharjee S, Martin JW. 2013a. Characterization of oil sands process-affected waters by liquid chromatography orbitrap mass spectrometry. Environmental science & technology. 47(10):5504-5513.
- Pereira AS, Islam MDS, Gamal El-Din M, Martin JW. 2013b. Ozonation degrades all detectable organic compound classes in oil sands process-affected water; an application of high-performance liquid chromatography/obitrap mass

spectrometry. Rapid Communications in Mass Spectrometry. 27(21):2317-2326.

- Pourrezaei P, Alpatova A, Chelme-Ayala P, Perez-Estrada LA, Jensen-Fontaine M, Le XC, El-Din MG. 2014. Impact of petroleum coke characteristics on the adsorption of the organic fractions from oil sands process-affected water. International Journal of Environmental Science and Technology. 11(7):2037-2050.
- Ramos-Padrón E, Bordenave S, Lin S, Bhaskar IM, Dong X, Sensen CW, Fournier J, Voordouw G, Gieg LM. 2011. Carbon and sulfur cycling by microbial communities in a gypsum-treated oil sands tailings pond. Environmental Science & Technology. 45(2):439-446.
- Robert C, Chassard C, Lawson PA, Bernalier-Donadille A. 2007. Bacteroides cellulosilyticus sp. Nov., a cellulolytic bacterium from the human gut microbial community. International journal of systematic and evolutionary microbiology. 57(7):1516-1520.
- Roman HJ, Burgess JE, Pletschke BI. 2006. Enzyme treatment to decrease solids and improve digestion of primary sewage sludge. African Journal of Biotechnology. 5(10).
- Ross MS, Pereira AdS, Fennell J, Davies M, Johnson J, Sliva L, Martin JW. 2012. Quantitative and qualitative analysis of naphthenic acids in natural waters surrounding the canadian oil sands industry. Environmental Science & Technology. 46(23):12796-12805.

- Saidi-Mehrabad A, He Z, Tamas I, Sharp CE, Brady AL, Rochman FF, Bodrossy L, Abell GCJ, Penner T, Dong X. 2013. Methanotrophic bacteria in oilsands tailings ponds of northern alberta. The ISME journal. 7(5):908-921.
- Samadi N. 2019. Partitioning of inorganic contaminants between fluid fine tailings and cap water under end pit lake scenario: Biological, chemical and mineralogical processes.
- Schoof A. 2015. Enzymatic treatment of phenolic industrial wastewater with nitrogen management.
- Scott AC, Young RF, Fedorak PM. 2008. Comparison of gc-ms and ftir methods for quantifying naphthenic acids in water samples. Chemosphere. 73(8):1258-1264.
- Siddique T, Kuznetsov P, Kuznetsova A, Arkell N, Young R, Li C, Guigard SE, Underwood E, Foght JM. 2014. Microbially-accelerated consolidation of oil sands tailings. Pathway i: Changes in porewater chemistry. Frontiers in microbiology. 5:106.
- Siddique T, Penner T, Semple K, Foght JM. 2011. Anaerobic biodegradation of longer-chain n-alkanes coupled to methane production in oil sands tailings. Environmental Science & Technology. 45(13):5892-5899.
- Siddique T, Stasik S, Mohamad Shahimin MF, Wendt-Potthoff K. 2018. Microbial communities in oil sands tailings: Their implications in biogeochemical processes and tailings management. Microbial communities utilizing hydrocarbons and lipids: members, metagenomics and ecophysiology

Handbook of hydrocarbon and lipid microbiology Springer Nature, Basel, Switzerland.1-33.

- Smith KS, Ingram-Smith C. 2007. Methanosaeta, the forgotten methanogen? Trends in microbiology. 15(4):150-155.
- Thomas L, Joseph A, Gottumukkala LD. 2014. Xylanase and cellulase systems of clostridium sp.: An insight on molecular approaches for strain improvement. Bioresource technology. 158:343-350.
- Tukanghan W, Hupfauf S, Gómez-Brandón M, Insam H, Salvenmoser W, Prasertsan P, Cheirsilp B, Sompong O. 2021. Symbiotic bacteroides and clostridium-rich methanogenic consortium enhanced biogas production of high-solid anaerobic digestion systems. Bioresource Technology Reports. 14:100685.
- Vajihinejad V, Guillermo R, Soares JBP. 2017. Dewatering oil sands mature fine tailings (mfts) with poly(acrylamide-co-diallyldimethylammonium chloride): Effect of average molecular weight and copolymer composition. Industrial & Engineering Chemistry Research. 56(5):1256-1266.
- Vanwonterghem I, Jensen PD, Dennis PG, Hugenholtz P, Rabaey K, Tyson GW. 2014. Deterministic processes guide long-term synchronised population dynamics in replicate anaerobic digesters. The ISME journal. 8(10):2015-2028.
- Wang S, Hou X, Su H. 2017. Exploration of the relationship between biogas production and microbial community under high salinity conditions. Scientific reports. 7(1):1-10.

- Whitman WB, Bowen TL, Boone DR. 2006. The methanogenic bacteria. Prokaryotes. 3(Chapter 9):165-207.
- Wilkins D, Lu X-Y, Shen Z, Chen J, Lee PKH. 2015. Pyrosequencing of mcra and archaeal 16s rrna genes reveals diversity and substrate preferences of methanogen communities in anaerobic digesters. Applied and environmental microbiology. 81(2):604-613.
- Wilson SL, Li C, Ramos-Padrón E, Nesbø C, Soh J, Sensen CW, Voordouw G, Foght J, Gieg LM. 2016. Oil sands tailings ponds harbour a small core prokaryotic microbiome and diverse accessory communities. Journal of Biotechnology. 235:187-196.
- Wu C, De Visscher A, Gates ID. 2019. On naphthenic acids removal from crude oil and oil sands process-affected water. Fuel. 253:1229-1246.
- Wu C, Zhang K, Chen J, Li S. 2014. Effect of cellulase and protease pretreatment on dewaterability of waste activated sludge from paper mill. BioResources. 9(3):5198-5207.
- Yang G-C, Zhou L, Mbadinga SM, Liu J-F, Yang S-Z, Gu J-D, Mu B-Z. 2016. Formate-dependent microbial conversion of co2 and the dominant pathways of methanogenesis in production water of high-temperature oil reservoirs amended with bicarbonate. Frontiers in microbiology. 7:365.
- Young RF, Li C, Semple KM, Baker MG, Kuznetsov P, Kuznetsova A, Siddique T, Foght JM. 2020. Accelerated consolidation of oil sands tailings using an anaerobic bioreactor. Bioresource Technology Reports. 11:100547.

- Yu X, Cao Y, Sampaga R, Rybiak S, Burns T, Ulrich AC. 2018a. Accelerated dewatering and detoxification of oil sands tailings using a biological amendment. Journal of Environmental Engineering. 144(9):04018091.
- Yu X, Lee K, Ma B, Asiedu E, Ulrich AC. 2018b. Indigenous microorganisms residing in oil sands tailings biodegrade residual bitumen. Chemosphere. 209:551-559.
- Yu X, Lee K, Ulrich AC. 2019. Model naphthenic acids removal by microalgae and base mine lake cap water microbial inoculum. Chemosphere. 234:796-805.
- Yuan XS, Shaw W. 2007. Novel processes for treatment of syncrude fine transition and marine ore tailings. Canadian Metallurgical Quarterly. 46(3):265-272.
- Zakaria BS, Dhar BR. 2021a. An intermittent power supply scheme to minimize electrical energy input in a microbial electrolysis cell assisted anaerobic digester. Bioresource Technology. 319:124109.
- Zakaria BS, Dhar BR. 2021b. Characterization and significance of extracellular polymeric substances, reactive oxygen species, and extracellular electron transfer in methanogenic biocathode. Scientific reports. 11(1):1-13.
- Zhang L, Zhang Y, Patterson J, Arslan M, Zhang Y, El-Din MG. 2020. Biofiltration of oil sands process water in fixed-bed biofilm reactors shapes microbial community structure for enhanced degradation of naphthenic acids. Science of the Total Environment. 718:137028.

- Zhu S, Li MY, El -Din MG. 2017. Forward osmosis as an approach to manage oil sands tailings water and on-site basal depressurization water. Journal of Hazardous Materials. 327:18-27.
- Zubot W, MacKinnon MD, Chelme-Ayala P, Smith DW, El-Din MG. 2012. Petroleum coke adsorption as a water management option for oil sands process-affected water. Sci Total Environ. 427:364-372.
- Zverlov VV, Kellermann J, Schwarz WH. 2005. Functional subgenomics of clostridium thermocellum cellulosomal genes: Identification of the major catalytic components in the extracellular complex and detection of three new enzymes. Proteomics. 5(14):3646-3653.

5 MITIGATING METHANE EMISSION FROM OIL SANDS TAILINGS USING ENZYMATIC AND LIME TREATMENTS⁴

5.1 Introduction

Surface-mined oil sands in north-eastern Alberta (Canada) represent the third largest production and reserves of global crude oil (Kaminsky et al. 2009). Regardless of the reuse of extraction water by the mining companies, large volumes of water are continuously produced during the hot-water bitumen extraction, and a massive quantity of tailings are stored in engineered tailings ponds (Government of Alberta 2015; Zhu et al. 2017). Environmental legislation mandating a zero-discharge practice prevents companies from directly discharging oil sands tailings and oil sands process water (ERCB 2012; McQueen et al. 2017). The surface area of tailings ponds is continuously increasing and new tailing ponds are being developed (Vedoy and Soares 2015; Young et al. 2020). The composition of oil sands tailings is 20-30 wt% water, and 1-3 wt% residual bitumen and hydrocarbon diluent such as naphtha (Allen 2008). Naphtha is a mixture of low molecular weight aliphatic and monoaromatic hydrocarbons (C₆–C₁₀) and insoluble and complex asphaltenes (Holowenko et al.

⁴ A version of this chapter has been published as: Allam, N.E.; Zakaria, B.S.; Kuznetsov, P.V.; Ulrich A. C.; Dhar, B. R. (2023): "Mitigating Methane Emission from Oil Sands Tailings Using Enzymatic and Lime Treatments" Chemosphere, 313, 137455.

2000; Siddique et al. 2020). Previous studies have illustrated the existence of complex anaerobic microbial communities with the ability to biodegrade a portion of diluent hydrocarbons and produce greenhouse gases (GHGs) such as methane (CH₄) and carbon dioxide (CO₂) under methanogenic conditions (Siddique et al. 2007; Small et al. 2015).

The remediation of oil sands tailings is problematic because of the increase in GHG emissions from the ponds. Many factors affect the presence and activity of different microorganisms in tailings, such as age, type, depth of tailings, and the additives used during bitumen extraction (Penner and Foght 2010; Ramos-Padrón et al. 2010). Within a single milliliter of fluid fine tailings (FFT) from Syncrude's Mildred Lake Settling Basin (MLSB), 10³ anaerobic heterotrophs, 10⁴ sulfatereducing prokaryotes, and methanogens (concentrations below detection limits) were detected using the conventional most probable number (MPN) method (Foght et al. 1985; Penner and Foght 2010; Ramos-Padrón et al. 2011). Many biogeochemical processes have been derived due to the methanogenesis processes in tailings ponds where hydrocarbon metabolism leads to GHGs emissions (Siddique et al., 2015). In the 1990s, methane bubbles started to appear at the surface of MLSB from methanogenic activity (Guo 2009; Sobolewski 1997), and methane production from MLSB in 2000 was estimated at up to 43 million L/day (Holowenko et al. 2000). The methane emissions from tailings ponds is currently uncontrolled and is considered an input to atmospheric GHGs (Siddique and Kuznetsova 2020).

Although natural detoxification of the ponds (i.e., degradation of contaminants) is beneficial, methane emissions pose significant environmental

178

concerns. The Government of Canada established regulations to assist the oil and gas sector reduce methane emissions by 40–45% by 2025 (Government of Canada 2018). Previous studies have contributed to the fundamental understanding of methanogenic biodegradation of hydrocarbon contaminants in tailings and the kinetics of methane emissions from tailing ponds (Liggio et al. 2019; Siddique et al. 2007; Simpson et al. 2010; You et al. 2021). However, engineered strategies to reduce GHGs emissions by methanogenic inhibition in tailings have rarely been examined. This information will allow the development of sustainable technologies for oil sands companies to achieve carbon emission reduction goals.

Methanogenic activity can be suppressed using various inhibitors such as 2bromoethanesulfonate (BES), acetylene (C_2H_2), ethylene (C_2H_4), methyl fluoride (CH₃F), lime, and various enzymes (Biswas et al. 2016; Chan and Parkin 2000; Grawert et al. 2014; Liang et al. 2011; Magdalena et al. 2018; Park et al. 2019; Zinder et al. 1984). BES is a widely reported selective inhibitor for methanogens (Grawert et al. 2014; Zinder et al. 1984); however, BES failed to inhibit methanogenesis over a sustained period during a few *in vivo* trials (Henderson et al. 2016; Immig et al. 1996). Among various enzymes, protease pre-treatment of Chlorella biomass could inhibit methane production in anaerobic digestion due to the accumulation of volatile fatty acids (VFAs) and an increase in Actinobacteria species (Magdalena et al. 2018). Biswas et al. (2016) also reported that lysozyme addition could inhibit fermentation in an *in vitro* rumen and reduce methane production. This reduction was due to the increase in the amounts of acetic acid, propionic acid and total VFA after lysozyme treatment. Adding lime inhibited the dry digestion of smooth cordgrass (Liang et al. 2011). Lime treatment causes an increase in the pH and alkalinity which have an adverse impact on the microbial activity resulting in an elimination of biogas production (Wong & Fang, 2000). Interestingly, treatment of oil sands tailings with various enzymes and lime is effective for dewatering oil sands tailings (Allam et al. 2022; Allam et al. 2021; Romaniuk et al. 2015; Tate et al. 2017; Tate et al. 2019). However, existing literature provides limited information on the effectiveness of enzymes and lime in inhibiting methane emissions from oil sands tailings.

This study focuses on investigating the effects of chemical treatment with lime and biological treatment with two different enzymes (lysozyme and protease) on biomethane emissions from oil sands tailings. The main objectives are as follows: 1) to investigate the feasibility of enzymatic and lime treatment to inhibit methane emissions from tailings ponds; 2) to investigate the corresponding mechanism for the inhibition; 3) to study the effect on the reduction of hydrocarbons; and, 4) to determine the changes in microbial community diversity and toxicity toward *Vibrio fischeri*. These objectives will provide insights to guide mitigation planning and evaluate any ongoing progress towards meeting the Canadian emission reduction goals.

5.2 Materials and methods

5.2.1 Materials

FFT was collected from the oil sands region in northern Alberta and stored in sealed buckets at 4 °C until use. Two different enzymes (protease and lysozyme) and lime were used as potential methanogenic inhibitors for tailings. Protease from *Bacillus* sp. (>16 U/g) was purchased from Sigma-Aldrich (Canada), MP

BiomedicalsTM Lysozyme from Chicken Egg White (20,000 U/mg) was purchased from Fisher Scientific (Canada), and lime (Ca[OH]₂) was purchased from Fisher Scientific (Canada). Based on previous studies (Navarro et al. 2014; Zinder et al. 1984), BES was used as an additive to the negative control for methanogens. Notably, Zinder et al. (1984) reported complete inhibition of methanogens in sludge anaerobic digestion with the addition of 50 μ M of BES. Sterilized FFT was used as an abiotic control condition; FFT was sterilized by autoclaving for 1 h per day for 8 successive days.

5.2.2 Experiment

The experiment was conducted in 158 mL microcosm bottles (Figure 5.1). Each bottle was sealed with a butyl rubber stopper and crimped, then purged with N₂:CO₂ (80:20%) to ensure anaerobic conditions. FFT (65 mL) was injected into each bottle using a 50 mL syringe and purged with N₂:CO₂ (80:20%). The bottles were then moved to an anaerobic chamber flushed with N₂ to perform the following steps. Anaerobic media (35 mL) was injected into each microcosm. The anaerobic media consisted of minerals, resazurin, phosphate solution, sodium bicarbonate, vitamin B solution, and sterilized sodium sulphide. Hexane (500 ppm) and toluene (150 ppm) were added as substrates. Further details regarding the media can be found in the literature (Holowenko et al. 2000). The abiotic control microcosms contained only 65 mL sterilized FFT and 35 mL anaerobic media purged with N₂:CO₂ (80:20%). The microcosms were incubated at 30 °C in the dark, and methane production was monitored. The lag phase for methane production was ~6 weeks. After the initiation of methane production, enzyme, lime, or BES were added to the microcosms. Three

dosages of protease (1%, 1.5%, and 3%), two doses of lysozyme (1.5% and 3%), three doses of lime (1600, 3500, and 5000 ppm), and one dose of BES (50 μ M) were tested. To prepare lime and lysozyme slurry, the powder was dissolved in Millipore ultrapure water (18.2 M Ω cm) to achieve the required dose (wt/wt). Protease and BES were purchased in liquid form and were added directly to the inoculum.



Figure 5.1. Photographs of experimental setup (a) 158 mL microcosm consists of 65 mL FFT and 35 mL of anaerobic media control, treated with enzyme, lime or BES, (b) the microcosms incubated in dark conditions at 30 °C.

5.2.3 Analysis

5.2.3.1 Chemical analysis

The methane content and hydrocarbons (i.e., hexane and toluene) concentrations were measured in the headspace of each microcosm with a typical injection volume of 100 μ L. The gas chromatograph (Agilent 7890A) was equipped with an Agilent Technologies HP-5MS column (30 m × 0.25 mm × 0.25 μ m) and a flame ionization detector (GC-FID). Soluble chemical oxygen demand (sCOD) was measured using HACH reagent kit (High range COD kit, 0–1500 ppm; HACH, Loveland, CO, USA). The sample was filtered through a 0.45 μ m membrane filter before using the COD kit. pH was measured with a LAQUAtwin pH meter (HORIBA Scientific, USA).

5.2.3.2 Toxicity bioassay

The toxicity of the inoculum samples to *V. fischeri* was measured with the Microtox[®] bioassay test (Osprey Scientific, Edmonton, Alberta, Canada). *V. fischeri*_is a luminescent bacterium used as a test organism in the Microtox bioassay. Microtox is a popular bioassay in the oil sands field to measure toxicity as the method is quick, easy, less expensive, and more reproducible than other toxicity bioassays (Brown & Ulrich, 2015). More details on the analytical method can be found elsewhere (Allam et al. 2021; Yu et al. 2018). According to the literature, 1.0 toxicity unit (TU) or lower is considered as not toxic (Scott et al. 2008; Yu et al. 2018).

5.2.3.3 Microbial community and diversity

Genomic DNA was extracted from the control and treated samples (sample weight ≈ 500 mg) with the FastDNATM SPIN Kit for Soil (MP Biomedicals, USA); more information can be found elsewhere (Allam et al. 2021). The DNA extraction was conducted for the initial inoculum (at start up: week 0) and at the end of the experiment (week 23) for the control sample and treated samples. The extracted genomic DNA was sequenced using Illumina Miseq by the Research and Testing Laboratory (Lubbock, TX, USA). The universal primers 515F (5' GTGCCAGCMGCCGCGGTAA 3') 806R and (5' GGACTACHVGGGTWTCTAAT 3') were used to target the 16S rRNA gene (Zakaria and Dhar 2021b). The 16S rRNA and mcrA gene copies were quantified with quantitative polymerase chain reaction (qPCR) assay using universal bacterial primers, specific archaeal primers, and methyl coenzyme-M reductase gene (mcrA) primers; Wilkins et al. (2015) used mcrA gene copies as a biomarker for hydrogenotrophic methanogenesis. The primer sequences and detailed qPCR protocol can be found elsewhere (Zakaria and Dhar 2021a). To evaluate the microbial taxonomy and diversity, the sequenced raw data were processed using QIIME2 (Bolyen et al. 2019). The detailed methodology is available elsewhere (Zakaria and Dhar 2021a; 2021b).

The relative abundances of the microbial community revealed by 16S rRNA gene sequencing were statistically computed by principal component analysis (PCA) using weighted Unifrac metrics to evaluate the relationship between microbial genera and treatment conditions. PCAs were computed and visualized using JMP software (v.11.0.0, SAS Institute Inc., US, <u>https://www.jmp.com</u>).

5.3 Results and discussion

5.3.1 Impact of enzymes and lime on biogenic methane production

Figure 5.2 illustrates the effects of enzymes (protease and lysozyme), lime, and BES on biogenic methane production compared with initial methane production after an incubation period of 23 weeks. The control microcosm had an increasing rate of methane production up to week 19, with a maximum change in methane production of ~1.3 mmol. After that, methane production started to decrease dramatically until the weekly change in methane production was 0.33 mmol at the end of the experiment (i.e., week 23). This reduction in the rate of methane production could be attributed to the lack of substrates (i.e., hexane and toluene) available for methanogens after week 19. BES (i.e., negative control) decreased the rate of methane production up to week 11 (0.25 mmol); after this, the effect of BES on the inoculum started to diminish, and the weekly methane production reached a maximum on week 21 with a 0.1 mmol change in methane production. A possible reason for this trend could be an insufficient dose of BES (Siddique et al. 2014).

The addition of 1.5% lysozyme led to higher change methane production throughout the experiment (2.3 mmol) than the lower dosages of protease, lime, and the control (0.9, 0.2, and 1.3 mmol change in methane, respectively). These results agree with our previous study where 1% lysozyme increased CH₄ production and

enhanced the water recovery from FFT (Allam et al. 2022). Kim et al. (2019) also reported an increase in CH₄ production from secondary sludge after lysozyme treatment. Lysozyme 1.5% increased the rate of CH₄ production up to week 17, then decreased the rate until the end of experiment at week 23 (0.55 mmol). The reduction in CH₄ production with lysozyme 1.5% treatment is similar to the control microcosm, where the CH₄ production rate reduction started at week 17 because of insufficient substrate. For lysozyme 3%, the change in CH₄ production decreased starting from 0.12 mmol at week 3 up to 0.48 mmol at week 23. Park et al. (2019) reported the significant inhibition of rumen protozoa with lysozyme but minimal or marginal inhibition with other peptidases. Another study by Biswas et al. (2016) reported the effect of lysozyme treatment on improving fermentation (i.e., increasing acetate, propionate, and volatile fatty acids (VFA) in an in vitro rumen and reducing CH₄ production. Lysozyme 3% had more of a negative impact on CH₄ production than BES after week 11. In contrast, protease 3% caused the highest reduction in CH₄ production throughout the experiment compared with lysozyme, lime, and BES. In the same sense, Allam et al. (2022) reported a low CH₄ production after protease treatment than other enzymes (i.e., lysozyme and cellulase). Increasing the treatment dose (i.e., lysozyme, protease, or lime) increased the inhibition of CH₄ production.

The protease 1% and 1.5% treatments had increased CH_4 production at the beginning of the experiment at weeks 3 and 5 (i.e., 0.375 and 0.4 for protease 1% and 0.86 and 0.89 for protease 1.5%, change in methane respectively). A slight decrease in CH_4 production was observed with both doses after week 5. As

illustrated in Figure 5.2, increasing the protease dose from 1% to 1.5% increases CH₄ production; however, increasing the dose to 3% has a negative influence on CH₄ production throughout the experiment. A decreasing trend was observed for protease 3%, with a maximum decrease in the change in CH₄ production at week 19 (0.72 mmol). A reduction in pH was observed from 6.8 at week 0 to 4.9 at week 23 with protease 3%. The reduction in pH might promote the accumulation of VFA and decrease CH₄ production, as previously reported by Magdalena et al. (2018) at pH 5.5. In addition, our results agree with Mahdy et al. (2015) who reported that protease pre-treatment leads to a high accumulation of VFA and inhibition of biogas production. The high nitrogen content produced after protease pre-treatment is toxic to the methanogenic microorganisms.

The addition of lime at all doses decreased the methane production from the inoculum. The highest reduction occurred with lime 5000 ppm at week 0 (pH \simeq 12; Figure 5.3). Methane production with lime 1600 ppm (pH 8.5) increased up to week 4 then decreased to the end of the experiment. Lime addition and BES addition had different effects; up to week 9, BES has higher inhibition of CH₄ production, and after week 11, lime had more inhibition. This trend is similar to the pattern described by Georgiou et al. (2019) who reported substantial killing of pathogenic microorganisms at pH > 11.5 after lime pre-treatment of anaerobically digested animal manure. Liang et al. (2011) also reported an inhibition of anaerobic microorganisms with simultaneous lime treatment and dry anaerobic digestion of smooth cordgrass under high lime loadings.



Figure 5.2. Temporal changes in methane production throughout the experiment for all treatment conditions, control/untreated FFT, and negative control 2-bromoethanesulfonate (BES) treatment. The temporal change represents the methane production at any week relative to the initial methane production for each microcosm. All microcosms consisted of 65 mL FFT and 35 mL anaerobic media. The control/untreated FFT represents a 0 ppm treatment and acts as the control treatment.



Figure 5.3. Change in pH throughout the experiment for all treatment conditions, the control/untreated FFT, and the negative control 2bromoethanesulfonate (BES). All microcosms consisted of 65 mL FFT and 35 mL anaerobic media. The control/ untreated FFT represents a 0 ppm treatment and acts as the control experiment.

5.3.2 Organics and hydrocarbon degradation

The impact of different treatment conditions on organics biodegradation was characterized by changes in sCOD and hydrocarbon concentration. The increasing enzyme doses increased the sCOD in the enzyme-treated samples above the sCOD observed in the controls and lime treatment (Figure 5.4). Previous studies reported an increase in the sCOD concentrations of excess sludge hydrolysis using different enzymes, including lysozyme and protease (Liu et al. 2019). Notably, enzymes can hydrolyze complex insoluble organics.

Protease treatment increased the initial sCOD from 690 ppm to 30000, 18600, and 12500 ppm for the 3%, 1.5%, and 1% treatments, respectively, at week 16, followed by a slight decrease at week 23 for all protease doses (22000, 13000, and 7500 ppm for 3%, 1.5%, and 1% treatments) (Figure 5.4b). For lysozyme treatment, the 3% dose caused a rapid increase in sCOD to 28100 ppm at week 8, which then gradually decreased until the end of the experiment, reaching 22500 ppm at week 23. With 1.5% lysozyme, a significant decrease in sCOD occurred from 18000 ppm at week 16 to 5600 ppm at week 23. The substantial increase in sCOD for both lysozyme dosages could be a result of the release of large amounts of soluble organic matter produced from lysozyme activity. Lysozyme has been previously reported to directly disrupt cell walls and release cellular organics into solution (Chen et al. 2008; Xin et al. 2015). The reduction observed after week 8 and week 16 for 3% and 1.5% dosages, respectively, can also be explained by the consumption of organic matter by lysozyme, also previously reported by Wang et al. (2022); this explains the decrease in sCOD at the end of the experiment. These results suggest that addition of enzymes initially increases sCOD in the inoculum and then degrades some of the organics by the end of experiment. In contrast, BES treatment resulted in a slight increase in sCOD after 8 weeks before reaching a plateau.

No significant change in sCOD concentration was observed with lime treatment (1290, 500, and 430 ppm sCOD for 1600, 3500, and 5000 ppm lime, respectively) compared with the control (192 ppm sCOD). This trend was opposite to the pattern described by Georgiou et al. (2019) who reported a 9% reduction in sCOD after lime treatment (pH = 12.1) of anaerobic digestate. However, the BES treatment resulted in a higher sCOD (5000 ppm) than the control and lime treatment.

Toluene was rapidly degraded in all treatment and control conditions. Siddique et al. (2007) previously reported that short chain *n*-alkanes (i.e., *n*-hexane) and BTEX compounds such as toluene can sustain methanogenesis; toluene is metabolized faster than other BTEX compounds. Lysozyme 1.5% had the highest hexane degradation at the end of the experiment compared with other treatment conditions (reduced to 45 ppm compared with 0 ppm in the control) (Figure 5.5). These results agree with the CH₄ production results where the lysozyme produced the highest CH₄ concentration after 23 weeks. Increasing the lysozyme dose to 3% reduced the hexane concentration to 157 ppm. The protease has the same effect on hexane degradation as CH₄ production, where increasing the protease dose from 1%, 1.5%, to 3% decreased the hexane degradation to 87, 171, and 194 ppm respectively. The negative control (BES) had higher degradation of hexane compared with other treatment conditions.
dose used in this experiment was not sufficient to completely inhibit the methanogens, as observed in other studies (Cárdenas-Manríquez et al. 2020; Siddique et al. 2014). Lime treatment resulted in less degradation of hexane throughout the experiment with a maximum degradation of 380 ppm occurring in the lime 1600 ppm treatment. Lime treatment reduced the hydrocarbon utilization performance of methanogens in the microcosm.



Figure 5.4. Changes in soluble chemical oxygen demand (sCOD) concentrations throughout the experiment for all treatment conditions, the control/untreated FFT, and the negative control 2-bromoethanesulfonate (BES). All microcosms consisted of 65 mL FFT and 35 mL anaerobic media. The control/ untreated FFT represents a 0 ppm treatment and acts as the control experiment.



Figure 5.5. Changes in hexane concentrations throughout the experiment for all treatment conditions, the control/untreated FFT, and the negative control 2-bromoethanesulfonate (BES). All microcosms consisted of 65 mL FFT and 35 mL anaerobic media. The control/ untreated FFT represents a 0 ppm treatment and acts as the control experiment.

5.3.3 The effect of different treatment conditions on toxicity

Table 5.1 summarizes the influence of different treatment conditions on the toxicity to *V. fischeri*, as measured with the Microtox[®] bioassay. The control and BES microcosms had the lowest toxicity of all treatment conditions; toxicity decreased from 6.6 TU to 1.3 TU and 0.8 TU, respectively, at week 23. The lime treatment had relatively lower toxicity than the lysozyme and protease treatments. Nonetheless, increasing both enzyme and lime dosages consistently increased the

toxicity. For instance, increasing the protease dose from 1% to 3% increased the toxicity of the microcosm from 7.0 TU to 124 TU. These finding agrees with Allam et al. (2022), where increasing the enzyme dose (protease and lysozyme) from 0.5%to 1% increased the toxicity of FFT porewater. Lysozyme dosages (1.5% and 3%) demonstrated more toxic conditions after week 12 (21 and 76 TU, respectively) compared with other treatment conditions. For lime, although all doses were toxic (i.e., TU > 1), the 1600 ppm dose resulted in the lowest toxicity (6.3 TU) compared with the 3500 ppm and 5000 ppm doses (9.7 and 11.3 TU, respectively). These results are consistent with the negative impact of lime addition (i.e., pH > 10) on bacterial cell counts in the FFT (Allam et al. 2021). Moreover, the toxicity of the enzyme and lime-treated conditions was consistent with the methane production and hydrocarbon degradation profiles. For instance, lysozyme 3% had lower methane production, lower hydrocarbon degradation, and higher toxicity than lysozyme 1.5%. Despite the inhibition of methanogenesis, BES had the lowest toxicity among all conditions, including the control. These results agree with methane production, where BES treatment produced less methane than the control. However, further investigation is still warranted regarding toxicity reduction after BES treatment.

Table 5.1. The effect of different treatment conditions on the toxicity to *Vibrio fisheri*. The impact of toxicity was monitored at week 0, week 12, and week 23 for all treatment conditions and compared with the control/untreated FFT and negative control BES.

Condition/ Time	Toxicity (TU)		
	Initial (Week 0)	Week 12	Week 23
Control	6.60 ± 0.5	2.2 ± 0.0	1.3 ± 0.4
Protease 1%		5.2 ± 0.4	7.0 ± 0.3
Protease 1.5%		6.3 ± 0.8	15.5 ± 0.4
Protease 3%		9.9 ± 0.9	124.8 ± 4.3
Lysozyme 1.5%		21.4 ± 0.7	41.1 ± 7.0
Lysozyme 3%		76.3 ± 1.3	81.3 ± 2.0
Lime 1600 ppm		1.6 ± 0.4	6.3 ± 0.1
Lime 3500 ppm		4.9 ± 0.1	9.7 ± 2.6
Lime 5000 ppm		7.6 ± 2.0	11.3 ± 3.9
BES 50 µmol/l		1.2 ± 0.2	0.8 ± 0.0

5.3.4 Microbial quantity

Figure 5.6 shows the quantitative analysis of microbial communities in the control and treated conditions using qPCR. For the control, the bacterial cell numbers were not significantly different between weeks 0 and 23 (5.7×10^6 vs. 5.4 $\times 10^6$ cells g⁻¹). Lime doses resulted in the highest reduction in 16S rRNA gene copies compared with the control and other conditions (2.7×10^5 , 1.71×10^5 , and

1.4 × 10⁵ cells g⁻¹ for 1600, 3500, and 5000 ppm lime, respectively). As aforementioned, the changing pH during lime treatment (pH to 8.5, 10, and 12 for 1600, 3500, and 5000 ppm lime, respectively) caused a sharp decline in the bacterial cell numbers. The 16S rRNA gene copies were higher in the control than in the protease 1% (9.9 × 10⁵ cells g⁻¹) and protease 1.5% treatments (1.0 × 10⁶ cells g⁻¹), with a higher reduction in the protease 3% treatment (3.3 × 10⁵ cells g⁻¹). Likewise, BES 50 μ M reduced the bacterial cell numbers to 3.6 × 10⁵ cells g⁻¹. In contrast, the lysozyme 1.5% and 3% treatments increased the bacterial cell numbers to 12.5 × 10⁷ and 1.6 × 10⁷ cells g⁻¹, respectively, which were higher than the control and other treatment conditions.

Similar to the bacterial cell number trends, lime resulted in the highest decrease in archaeal cell numbers. Using lime dose of 1600 ppm had the highest reduction in archaeal cell numbers (5.8×10^2 cells g⁻¹) compared with the control (3.76×10^5 cells g⁻¹) and other treatment conditions. The changing pH during lime treatment (pH 8.5, 10, and 12 for 1600, 3500, and 5000 ppm lime, respectively) caused a sharp decline in the archaeal cell numbers, which reduced the methane production. Protease 1% and 3% and BES 50 µM resulted in the second highest reduction in archaeal cell numbers, after that of the lime treatment condition. However, lysozyme treatment had almost no effect on the archaeal cell numbers when compared with the control. *mcr*A gene copies were quantified as a biomarker for hydrogenotrophic methanogenesis (Wilkins et al. 2015). Similar trends were observed for *mcr*A gene copies, with the lime 1600 ppm treatment having the highest decrease in *mcrA* gene copy numbers.



Figure 5.6. Changes in cell number or gene copy number per g inoculum at the beginning and end of the experiment for the control condition, and at week 23 for all treatment conditions and the negative control. All microcosms consisted of 65 mL of FFT and 35 mL anaerobic media. The control/ untreated FFT represents a 0 ppm treatment and acts as the control experiment.

Figure 5.7 represents the alpha diversity metrics of the microbial community computed using QIIME 2. The different treatments had an impact on the microbial diversity. The lowest richness of the microbial community (measured by Chao1 and observed taxonomic units; OTUs) was observed in the lysozyme 3% treatment, with a notable increase in diversity with other treatments such as protease 1% and 1.5%; lime 5000, 1600, and 3500; and BES 50 μ M. The lime 3500 ppm treatment had the highest microbial community richness of all other treatment conditions. The highest microbial diversity (measured by Shannon index) was found in the BES 50 μ M (6.6) and lime 1600 ppm treatments (6.6), followed by lime 5000 ppm (6.5) and lime 3500 ppm treatments (6.3); the lysozyme 1.5% treatment had a significantly decreased Shannon index (2.7), followed by lysozyme 3% (4.0) and protease 3% (4.8). These results indicate that the treatment conditions significantly shifted the microbial diversity.



Figure 5.7. The alpha diversity metrics of the microbial community computed using QIIME 2 for all treatment conditions, the control/untreated FFT, and the negative control 2-bromoethanesulfonate (BES). All microcosms consisted of 65 mL FFT and 35 mL anaerobic media. The control/ untreated FFT represents a 0 ppm treatment and acts as the control experiment.

5.3.4.1 Microbial community

The microbial community composition of the control and treatment conditions are shown in Figure 5.8. The relative abundances of the microbial community were directly impacted by the treatment conditions after 23 weeks incubation. At the domain level, archaea decreased in the control microcosm after 23 weeks from 27.7% to 10.9%. All treatment conditions had lower relative abundance of archaea than in the control. The highest reduction in the relative abundance of archaea was 2.2% for the lime 5000 ppm treatment, followed by 3.3% for lysozyme 1.5% treatment and 4.9% for lysozyme 3% treatment.



Figure 5.8. Change in the relative abundance at domain level for the control at the beginning of the experiment (Initial), the control at the end of the experiment (Control), and week 23 for all other treatment conditions.

The microbial community comprised diverse phyla mainly dominated by Firmicutes, Proteobacteria and Chloroflexi (Figure 5.9). In the control microcosm at week 0, Proteobacteria (40%) was the most abundant phylum, followed by Euryarchaeota (28%) and Firmicutes (16%). At week 23, the abundance of Proteobacteria increased to 67%, however, the Euryarchaeota and Firmicutes abundances decreased to 11% and 7%, respectively.

After the treatments, Firmicutes, Proteobacteria, and Chloroflexi remained the most abundant phyla. After protease treatment (1%, 1.5%, and 3%), Firmicutes increased and became the most dominant phylum with 62%, 66%, and 80% abundance, respectively. After lysozyme treatment, Bacteroidetes was enriched and became the most abundant phyla in lysozyme 1.5% with 79% abundance; however, increasing the dose to lysozyme 3% shifted the dominant phylum to Firmicutes (69% abundance). After lime treatment (1600, 3500, and 5000 ppm), Firmicutes (19%, 37%, and 31%) and Chloroflexi (19%, 21%, and 25%) were the most abundant phyla. After 50 µM BES treatment, Proteobacteria (40%) and Chloroflexi (27%) were the dominant phyla. These results agree with the increased CH₄ production with increased protease dose up to 1.5%. Previous studies reported the enrichment of the microbial community during methanogenic activity (Kuznetsova et al. 2021; Young et al. 2020). Chloroflexi was first reported by Ficker et al. (1999) to be related to toluene degradation in a toluene-degrading methanogenic consortium.



Figure 5.9. Changes in the relative abundance at the phylum level for the control at the beginning of the experiment (Initial), the control at the end of the experiment (Control), and week 23 for all other treatment conditions.

At the genus level (Figure 5.10a), a more diverse microbial community was observed. The control microcosm at week 0 was mainly dominated by *Desulfotomaculum* (16%); however, the abundance of this genus declined to 7% after 23 weeks of incubation. Another dominant genus was *Acidovorax* (14%) at week 0, which increased significantly to 53% after 23 weeks of incubation. *Acidovorax* was previously reported in Shell Albian sands tailings (Cárdenas-Manríquez et al. 2020).

Both Desulfotomaculum and Acidovorax significantly decreased in all other treatment conditions. In contrast, Siddique et al. (2020) reported a significant increase in the genus Desulfotomaculum after 120 d incubation after the degradation of isoalkanes and cycloalkanes. Toth and Gieg (2018) also reported an increase in *Desulfotomaculum* after methanogenic biodegradation of short chain *n*alkanes, cyclohexane, and mono-aromatics. After protease treatments, *Clostridium*, a member of the Deltaproteobacteria, was dominant and increased in relative abundance with increasing protease dose (26%, 31%, and 58% for the 1%, 1.5%, and 3% doses, respectively). As reported previously by Fowler et al. (2016), an increase in the relative abundance of the Clostridium results in increased downstream conversion of smaller molecules to methanogenic intermediates. For the lysozyme 1.5% treatment, two different genera appeared: *Clostridium* (23%) and Anaerovorax (37%). However, in the lysozyme 1.5% treatment, Marinilabiaceae was the dominant family with a relative abundance of 79%. The dominant genera observed in most of the lime treatment conditions and BES were Desulfosporosinus (29%-4%), Desulfotomaculum (9%-3%), and some genera of the family of *Anaerolinaceae* (14%–6%). Previous reports by Sutton et al. (2013) linked the presence of the family of *Anaerolineaceae* with the anaerobic degradation of oil-related compounds such as *n*-alkanes.

The dominant archaeal genera in the control microcosms were Methanosaeta (3%–9%), Methanolinea (3%–8%), Methanoculleus (3%–7%), and Candidatus Methanoregula (0.2%-4%) (Figure 5.10b). A slight decrease in most of the methanogen-affiliated genera was observed in the control condition after 23 weeks of incubation. For the protease treatment (1% and 2%), Candidatus Methanoregula, a hydrogenotrophic methanogen, was dominant over the other genera that were suppressed after treatment compared with control. Most of the treatments reduced the relative abundance of Methanosaeta, an acetoclastic methanogen, after 23 weeks incubation; the highest reduction occurred with protease 1%, from 8% to 0.55% relative abundance. Methanosaeta is mainly found in wetlands and is considered one of the most active methanogens responsible for producing large amounts of CH₄ (Fowler et al. 2016). Methanolinea (a known hydrogenotrophic methanogen) decreased more in most treatment conditions than in the control; with protease 1%, Methanolinea decreased from 3% to 0.4% and with BES 50 µM it decreased to 0.3%. The lime doses of 1600 ppm and 3500 ppm resulted in the lowest reduction in archaea genera compared with other treatment conditions.



Figure 5.10. (a) Changes in the relative abundance at the genus level for bacteria, (b) for the archaea for the control at the beginning of the experiment (Initial), the control at the end of the experiment (Control), and week 23 for all other treatment conditions.

5.3.5 Principal Component Analysis (PCA)

PCA was performed to analyze the relationship between the treatment conditions and microbial community abundances at the phylum and genus levels (Figure 5.11). The treatment conditions differently impacted the abundances of the microbial communities. At the phylum level, the control and lime 1600 ppm treatments clustered together in the same quadrant (top right) (Figure 5.11a). Euryarchaeota had a strong relationship with the control and lime 1600 ppm treatments (Figure 5.11b). The relative abundance of *Euryarchaeota* agreed with previous studies that focused on biomethane production (Magdalena et al. 2019). Another study reported that *Euryarchaeota* comprises hydrogentrophic and acetotrophic methanogens (Fowler et al. 2016). Other treatment conditions, such as lysozyme 1.5% and 3% and protease 1.5% and 3%, were closely associated and had a strong relationship to the abundances of Bacteroidetes and Firmicutes. The lime 3500 and 5000 ppm and BES 50 μ M treatments were located in the same quadrant (right bottom) and were closely associated with Chloroflexi, Actinobacteria, and Proteobacteria.

The relationship between the treatment conditions and archaeal genera was also evaluated using PCA (Figure 5.11c and b). Similar to the phylum level, the control and lime 1600 ppm treatments were located in the same quadrant (right bottom), closely correlated with hydrogenotrophic methanogens (*Methanobacterium*, *Methanoculleus*, and *Methanolinea*) (Zakaria and Dhar 2019). High CH₄ production in the control condition may be attributed to the abundance of the hydrogenotrophic methanogens *Methanobacterium*, *Methanoculleus*, and Methanolinea. Therefore, hydrogenotrophic methanogens provide the predominant pathway for CH₄ production in the control microcosm, with lower activity of acetoclastic methanogens. Similarly, the lime 3500 ppm and protease 1.5% treatments were clustered together in the right-top quadrant with the hydrogenotrophic methanogen Candidatus Methanoregula (Yamamoto et al. 2014). The protease 1% and BES 50 µM treatments were located on the top-left quadrant and were associated with Candidatus Methanoregula. Methanosarcina was located in a separate quadrant and showed no association with any other genera. These results illustrated that both hydrogenotrophic (i.e., Methanoregula) and acetoclastic (i.e., Methanosarcina) methanogens governed the pathway for CH4 production after protease 1% treatment. The lysozyme 1.5%, lysozyme 3%, protease 3%, and lime 5000 ppm treatments were very weakly associated with archaeal abundance (Figure 5.10c). These findings agree with the CH₄ results where protease 3%, lysozyme 3%, and lime 1500 ppm treatments resulted in inhibition of CH₄ production.



Figure 5.11. PCA loading plot and biplot between different treatment conditions and abundances of the microbial communities at the (a, b) phylum levels, and (c, d) archaeal genera level.

5.3.6 Significance of Results

Despite the effectiveness of enzymes (e.g., protease) in inhibiting methanogenic activity, due to their high costs as compared to lime the feasibility of enzymatic treatment is questionable (Cammarota & Freire, 2006; Ramkumar et al., 2016; Silvetti et al., 2017). However, there has been considerable research on the low-cost production of various enzymes for different industrial applications (including wastewater treatment), which may make enzymatic treatment economically sound in the near future (Cammarota & Freire, 2006; Ramkumar et al., 2016; Silvetti et al., 2017). While this study focused on understanding the fundamental roles of enzymes and lime treatment in mitigating methane emission from oil sands tailings management, a detailed economic and life-cycle assessment in future research would be critical for selecting the most viable method.

5.4 Conclusions

In summary, 3% protease was the most effective for methanogenesis inhibition. Microcosm acidification (pH of 4.9 after week 23) due to VFAs generation in protease-treated tailings may have contributed to the methanogenic inhibition. Alternatively, the formation of toxic compounds, including high nitrogen content after protease treatment, could adversely affect the microorganisms. Among different lime doses, 5000 ppm resulted in the lowest CH₄ production, while enzyme treatments resulted in relatively less CH₄ production. All lime doses resulted in the lowest hydrocarbon degradation rates compared with the control and other treatment conditions. However, at the end of 23 weeks of incubation, 5000 ppm lime provided the lowest toxicity (11.3 TU) to *V. fischeri* than enzyme treatments at higher doses (124.8 TU for protease 3% and 81.3 TU for lysozyme 3%). PCA analysis revealed that lysozyme 3%, protease 3%, and lime 5000 ppm have a weak association with archaeal abundance; therefore, these treatments inhibited CH_4 production. Further investigation is still warranted on the effect of enzyme doses and identifying the methanogenesis inhibition mechanisms.

5.5 References

- Allam NE, Anwar MN, Kuznetsov PV, Ulrich AC, Dhar BR. 2022. Enzymeassisted dewatering of oil sands tailings: Significance of water chemistry and biological activity. Chemical Engineering Journal.135162.
- Allam NE, Romaniuk N, Tate M, Meshref MNA, Dhar BR, Ulrich AC. 2021. Impact of lime treatment on tailings dewatering and cap water quality under an oil sands end pit lake scenario. Science of The Total Environment. 781:146699.
- Allen EW. 2008. Process water treatment in canada's oil sands industry: I. Target pollutants and treatment objectives. Journal of Environmental Engineering and Science. 7(2):123-138.
- Anderson JC, Wiseman SB, Wang N, Moustafa A, Perez-Estrada L, Gamal El-Din M, Martin JW, Liber K, Giesy JP. 2011. Effectiveness of ozonation treatment in eliminating toxicity of oil sands process-affected water to chironomus dilutus. Environ Sci Technol. 46(1):486-493.
- Biswas AA, Lee SS, Mamuad LL, Kim S-H, Choi Y-J, Bae G-S, Lee K, Sung H-G, Lee S-S. 2016. Use of lysozyme as a feed additive on in vitro rumen fermentation and methane emission. Asian-Australasian journal of animal sciences. 29(11):1601-1607.
- Bolyen E, Rideout JR, Dillon MR, Bokulich NA, Abnet CC, Al-Ghalith GA, Alexander H, Alm EJ, Arumugam M, Asnicar F. 2019. Reproducible,

interactive, scalable and extensible microbiome data science using qiime 2. Nature biotechnology. 37(8):852-857.

- Cárdenas-Manríquez M, Young RF, Semple KM, Li C, Coy D, Underwood E, Siddique T, Guigard S, Bressler DC, Gupta R et al. 2020. Microbiallymediated de-watering and consolidation ("biodensification") of oil sands mature fine tailings, amended with agri-business by-products. Nova Scientia. 12(24).
- Chan ASK, Parkin TB. 2000. Evaluation of potential inhibitors of methanogenesis and methane oxidation in a landfill cover soil. Soil Biology and Biochemistry. 32(11):1581-1590.
- Chen L, Guangxia WU, Huanzhen MU, Zonghuan Y, Lianyi T, Xiangwei LIN. 2008. Synthesis and application of lignin-based copolymer lsaa on controlling non-point source pollution resulted from surface runoff. Journal of Environmental Sciences. 20(7):820-826.
- ERCB. 2012. Alberta's energy reserves 2011 and supply/demand outlook 2012-2021. Government of alberta, c., ab, canada, 2012 (ed).
- Ficker M, Krastel K, Orlicky S, Edwards E. 1999. Molecular characterization of a toluene-degrading methanogenic consortium. Applied and environmental microbiology. 65(12):5576-5585.
- Foght JM, Fedorak PM, Westlake DWS, Boerger HJ. 1985. Microbial content and metabolic activities in the syncrude tailings pond. AOSTRA J Res. 1:139-146.

- Fowler SJ, Toth CRA, Gieg LM. 2016. Community structure in methanogenic enrichments provides insight into syntrophic interactions in hydrocarbonimpacted environments. Frontiers in microbiology. 7:562.
- Georgiou D, Liliopoulos V, Aivasidis A. 2019. Investigation of an integrated treatment technique for anaerobically digested animal manure: Lime reaction and settling, ammonia stripping and neutralization by biogas scrubbing. Bioresource Technology Reports. 5:127-133.
- Government of Alberta. 2015. Lower athabasca region : Tailings management framework for the mineable athabasca oil sands.
- Government of Canada. 2018. Technical backgrounder: Federal methane regulations for the upstream oil and
- gas sector canada.Ca, https://www. Canada.Ca/en/environment-climatechange/news/2018/04/federalmethane-regulations-for-the-upstream-oiland-gas-sector.Html, (accessed jan 28, 2022).
- Grawert T, Hohmann H-P, Kindermann M, Duval S, Bacher A, Fischer M. 2014.
 Inhibition of methyl-com reductase from methanobrevibacter ruminantium
 by 2-bromoethanesulfonate. Journal of agricultural and food chemistry.
 62(52):12487-12490.
- Guo C. 2009. Rapid densification of the oil sands mature fine tailings (mft) by microbial activity.
- Henderson G, Cook GM, Ronimus RS. 2016. Enzyme-and gene-based approaches for developing methanogen-specific compounds to control ruminant

methane emissions: A review. Animal Production Science. 58(6):1017-1026.

- Holowenko FM, MacKinnon MD, Fedorak PM. 2000. Methanogens and sulfatereducing bacteria in oil sands fine tailings waste. Canadian Journal of Microbiology. 46(10):927-937.
- Immig I, Demeyer D, Fiedler D, Van Nevel C, Mbanzamihigo L. 1996. Attempts to induce reductive acetogenesis into a sheep rumen. Archives of Animal Nutrition. 49(4):363-370.
- Kaminsky HAW, Etsell TH, Ivey DG, Omotoso O. 2009. Distribution of clay minerals in the process streams produced by the extraction of bitumen from athabasca oil sands. Can J Chem Eng. 87(1):85-93.
- Kim S, Woo S-G, Lee J, Lee D-H, Hwang S. 2019. Evaluation of feasibility of using the bacteriophage t4 lysozyme to improve the hydrolysis and biochemical methane potential of secondary sludge. Energies. 12(19):3644.
- Kuznetsova A, Kuznetsov P, Young RF, Semple KM, Li C, Foght JM, Siddique T. 2021. Microbial transformation of solid phase impacts quality of recovered water during consolidation of bioreactor-treated oil sands tailings. Journal of Environmental Chemical Engineering. 9(1):104715.
- Liang Y, Zheng Z, Hua R, Luo X. 2011. A preliminary study of simultaneous lime treatment and dry digestion of smooth cordgrass for biogas production. Chemical Engineering Journal. 174(1):175-181.

- Liggio J, Li S-M, Staebler RM, Hayden K, Darlington A, Mittermeier RL, O'Brien J, McLaren R, Wolde M, Worthy D. 2019. Measured canadian oil sands co2 emissions are higher than estimates made using internationally recommended methods. Nature communications. 10(1):1-9.
- Liu G, Wang K, Li X, Ma L, Ma X, Chen H. 2019. Enhancement of excess sludge hydrolysis and decomposition with different lysozyme dosage. Journal of Hazardous Materials. 366:395-401.
- Magdalena JA, Llamas M, Tomás-Pejó E, González-Fernández C. 2019. Semicontinuous anaerobic digestion of protease pretreated chlorella biomass for volatile fatty acids production. Journal of Chemical Technology & Biotechnology. 94(6):1861-1869.
- Magdalena JA, Tomás-Pejó E, Ballesteros M, González-Fernandez C. 2018. Volatile fatty acids production from protease pretreated chlorella biomass via anaerobic digestion. Biotechnology Progress. 34(6):1363-1369.
- Mahdy A, Mendez L, Ballesteros M, González-Fernández C. 2015. Protease pretreated chlorella vulgaris biomass bioconversion to methane via semicontinuous anaerobic digestion. Fuel. 158:35-41.
- McQueen AD, Kinley CM, Hendrikse M, Gaspari DP, Calomeni AJ, Iwinski KJ, Castle JW, Haakensen MC, Peru KM, Headley JV et al. 2017. A risk-based approach for identifying constituents of concern in oil sands processaffected water from the athabasca oil sands region. Chemosphere. 173:340-350.

- Navarro SS, Cimpoia R, Bruant G, Guiot SR. 2014. Specific inhibitors for identifying pathways for methane production from carbon monoxide by a nonadapted anaerobic mixed culture. Canadian Journal of Microbiology. 60(6):407-415.
- Park T, Mao H, Yu Z. 2019. Inhibition of rumen protozoa by specific inhibitors of lysozyme and peptidases in vitro. Frontiers in microbiology. 10:2822.
- Penner TJ, Foght JM. 2010. Mature fine tailings from oil sands processing harbour diverse methanogenic communities. Canadian journal of microbiology. 56(6):459-470.
- Ramos-Padrón E, Bordenave S, Lin S, Bhaskar IM, Dong X, Sensen CW, Fournier J, Voordouw G, Gieg LM. 2010. Carbon and sulfur cycling by microbial communities in a gypsum-treated oil sands tailings pond. Environmental science & technology. 45(2):439-446.
- Ramos-Padrón E, Bordenave S, Lin S, Bhaskar IM, Dong X, Sensen CW, Fournier J, Voordouw G, Gieg LM. 2011. Carbon and sulfur cycling by microbial communities in a gypsum-treated oil sands tailings pond. Environmental Science & Technology. 45(2):439-446.
- Romaniuk N, Mehranfar M, Selani H, Ozum B, Tate MJ, Scott JD. 2015. Residual bitumen recovery from oil sand tailings with lime. Proceedings tailings and mine waste 2015, vancouver, bc, october 26 to 28, 2015.

- Scott AC, Young RF, Fedorak PM. 2008. Comparison of gc-ms and ftir methods for quantifying naphthenic acids in water samples. Chemosphere. 73(8):1258-1264.
- Siddique T, Fedorak PM, MacKinnon MD, Foght JM. 2007. Metabolism of btex and naphtha compounds to methane in oil sands tailings. Environmental science & technology. 41(7):2350-2356.
- Siddique T, Kuznetsov P, Kuznetsova A, Arkell N, Young R, Li C, Guigard SE, Underwood E, Foght JM. 2014. Microbially-accelerated consolidation of oil sands tailings. Pathway i: Changes in porewater chemistry. Frontiers in microbiology. 5:106.
- Siddique T, Kuznetsova A. 2020. Linking hydrocarbon biodegradation to greenhouse gas emissions from oil sands tailings and its impact on tailings management. Canadian Journal of Soil Science.1-9.
- Siddique T, Semple K, Li C, Foght JM. 2020. Methanogenic biodegradation of isoalkanes and cycloalkanes during long-term incubation with oil sands tailings. Environmental Pollution. 258:113768.
- Simpson IJ, Blake NJ, Barletta B, Diskin GS, Fuelberg HE, Gorham K, Huey LG, Meinardi S, Rowland FS, Vay SA. 2010. Characterization of trace gases measured over alberta oil sands mining operations: 76 speciated c 2–c 10 volatile organic compounds (vocs), co 2, ch 4, co, no, no 2, no y, o 3 and so 2. Atmospheric Chemistry and Physics. 10(23):11931-11954.

- Small CC, Cho S, Hashisho Z, Ulrich AC. 2015. Emissions from oil sands tailings ponds: Review of tailings pond parameters and emission estimates. Journal of Petroleum Science and Engineering. 127:490-501.
- Sobolewski A. 1997. Microbial distributions in process-affected aquatic ecosystems. Consultants report submitted to Alberta Oil Sand Technology Research Authority Calgary, Alberta.
- Sutton NB, Maphosa F, Morillo JA, Abu Al-Soud W, Langenhoff AAM, Grotenhuis T, Rijnaarts HHM, Smidt H. 2013. Impact of long-term diesel contamination on soil microbial community structure. Applied and environmental microbiology. 79(2):619-630.
- Tate M, Leikam J, Fox J, Romaniuk N. 2017. Use of calcium hydroxide as a coagulant to improve oil sands tailings treatment. Proceedings tailings and mine waste 2017, banff, alberta, canada, november 2017.
- Tate M, Leikam J, Fox J, Romaniuk N. 2019. Treatment of oil sands tailings with lime at elevated ph levels. Google Patents.
- Toth CRA, Gieg LM. 2018. Time course-dependent methanogenic crude oil biodegradation: Dynamics of fumarate addition metabolites, biodegradative genes, and microbial community composition. Frontiers in Microbiology.
 8.
- Vedoy DRL, Soares JBP. 2015. Water-soluble polymers for oil sands tailing treatment: A review. The Canadian Journal of Chemical Engineering. 93(5):888-904.

- Wang W, Wang K, Zhao Q, Liu Y. 2022. Simultaneous degradation of anodic sludge and cathodic refractory pollutant in a mfc powered ef system enhanced by co-addition of lysozyme and 2-bromoethane sulfonate. Journal of Environmental Chemical Engineering. 10(3):108074.
- Wilkins D, Lu X-Y, Shen Z, Chen J, Lee PKH. 2015. Pyrosequencing of mcra and archaeal 16s rrna genes reveals diversity and substrate preferences of methanogen communities in anaerobic digesters. Applied and environmental microbiology. 81(2):604-613.
- Xin X-D, He J-G, Qiu W, Tang J, Liu T-T. 2015. Microbial community related to lysozyme digestion process for boosting waste activated sludge biodegradability. Bioresource technology. 175:112-119.
- Yamamoto K, Tamaki H, Cadillo-Quiroz H, Imachi H, Kyrpides N, Woyke T, Goodwin L, Zinder SH, Kamagata Y, Liu W-T. 2014. Complete genome sequence of methanoregula formicica smspt, a mesophilic hydrogenotrophic methanogen isolated from a methanogenic upflow anaerobic sludge blanket reactor. Genome announcements. 2(5):e00870-00814.
- You Y, Staebler RM, Moussa SG, Beck J, Mittermeier RL. 2021. Methane emissions from an oil sands tailings pond: A quantitative comparison of fluxes derived by different methods. Atmospheric Measurement Techniques. 14(3):1879-1892.

- Young RF, Li C, Semple KM, Baker MG, Kuznetsov P, Kuznetsova A, Siddique T, Foght JM. 2020. Accelerated consolidation of oil sands tailings using an anaerobic bioreactor. Bioresource Technology Reports. 11:100547.
- Yu X, Cao Y, Sampaga R, Rybiak S, Burns T, Ulrich AC. 2018. Accelerated dewatering and detoxification of oil sands tailings using a biological amendment. Journal of Environmental Engineering. 144(9):04018091.
- Zakaria BS, Dhar BR. 2019. Progress towards catalyzing electro-methanogenesis in anaerobic digestion process: Fundamentals, process optimization, design and scale-up considerations. Bioresource technology. 289:121738.
- Zakaria BS, Dhar BR. 2021a. Characterization and significance of extracellular polymeric substances, reactive oxygen species, and extracellular electron transfer in methanogenic biocathode. Scientific reports. 11(1):1-13.
- Zakaria BS, Dhar BR. 2021b. An intermittent power supply scheme to minimize electrical energy input in a microbial electrolysis cell assisted anaerobic digester. Bioresource Technology. 319:124109.
- Zhu S, Li MY, El -Din MG. 2017. Forward osmosis as an approach to manage oil sands tailings water and on-site basal depressurization water. Journal of Hazardous Materials. 327:18-27.
- Zinder SH, Anguish T, Cardwell SC. 1984. Selective inhibition by 2bromoethanesulfonate of methanogenesis from acetate in a thermophilic anaerobic digestor. Applied and environmental microbiology. 47(6):1343-1345.

6 CONCLUSIONS AND RECOMMENDATIONS

6.1 Conclusions and significance of results

The key findings and significance of results from each experimental chapter are summarized below:

Chapter 3. Impact of lime treatment on tailings dewatering and cap water quality under an oil sands end pit lake scenario. As the first study to examine the effectiveness of lime treatment under an EPL scenario, generally, the performance metrics and indicative parameters revealed the role of lime addition towards the reclamations benefits within EPLs. Higher water recovery % from the FFT layer was achieved after using a high lime dose of 3500 ppm (i.e., pH > 11.5) compared with low lime doses and control. Lower alkalinity for the cap water was maintained at high lime doses of 3500 ppm, which achieved low alkalinity environment for an aquatic ecosystem in EPLs. Furthermore, cation exchange and clay settling occurred at high lime doses (pH > 11.5). In terms of microbial cell numbers in the underlying FFT, when pH > 10 (i.e., high lime doses 1600, 3500, and 4000 ppm), the microbial cell numbers in the underlying FFT impacted negatively. At high lime doses, minimal reduction of organics (PHC) compared with moderate reduction at a low lime dose (650 ppm). Lime treatment demonstrated high toxicity effects on V. fischeri in the FFT layer. The atmospheric CO₂ reduced the pH of the cap water over time due to the regeneration of the carbonate buffer. Hence it reduced the toxicity of the cap water. The study introduced the proof of concept for the lime treatment in EPL in order to proceed with larger-scale applications of lime treatment. The current findings assisted in the optimization of the lime treatment with an understanding of the cation exchange occurrence during high lime dosage treatment. The study highlighted the associated risks for the effect of the movement of COPCs into the cap water and toward the aquatic life within an EPL scenario.

Chapter 4. Enzyme-assisted dewatering of oil sands tailings: Significance of water chemistry and biological activity. This was the first study to examine the effectiveness of enzymatic treatment in the oil sands sector to accelerate the dewatering of FFT. The dissolution of carbonate minerals during the release of methane gas increased ionic strength and decreased the diffuse double layer (DDL) of the FFT. A significant increase in water recovery from FFT using lysozyme was shown due to the increase in both ionic strength and zeta potential and the decrease in DDL thickness. The link between using enzymes (i.e., cellulase and lysozyme) and the ebullition of methane gas resulting from the methanogenic activity were illustrated, which created pathways for the trapped water release. Lysozyme was the most effective in charge neutralization of clay particles, which could be attributed to the positive charge carried by lysozyme under pH conditions of tailings (pH< 7.5). Lysozyme treatment reduced the NAFC concentrations in pore water. However, the released pore water from the lysozyme treatment had the highest toxicity compared with the other enzyme treatments. The study presented for the first time the concept of the enzymatic treatment in oil sands applications. The conclusions from the work endorsed the enzymatic treatment as a promising method for the dewatering of the oil sands during lysozyme treatment. The study highlighted the associated risks for the increase of toxicity in pore water released from the enzymatic treatment due to the release of organic fractions.

Chapter 5. Significance of enzyme and lime treatment in the mitigation of methane emissions from oil sands tailings. This study examined the mitigation of methane emissions from oil sands tailings using lime and enzymatic treatment. Increasing the enzyme dose (i.e., lysozyme and protease) to 3% suppressed the CH₄ production and reduced the degradation of hexane compared with lime treatment. Protease treatment caused a reduction in the pH below 5.0, which promotes the formation of VFA and reduce the CH₄ production. Another potential mechanism could be the elevated nitrogen levels (toxic compounds) after protease treatment, affecting the microbial Lime treatment demonstrated a low toxicity effect on V. community. fischeri compared with other enzymatic treatment. All lime doses illustrated the lowest degradation of hydrocarbons and highest reduction in 16S rRNA and archaeal gene copies compared with control and other treatment conditions. The PCA analysis identified a feeble association with the archaeal abundance for lysozyme 3%, protease 3%, and lime 5000 ppm. Therefore, an inhibition effect for the CH₄ production was embraced by the lysozyme 3%, protease 3%, and lime 5000 ppm. For the first time the concept of the mitigation of methane emissions from oil sands tailings using the enzymatic and lime treatment was presented. The study emphasized the associated benefits for the enzymatic treatment on the inhibition of CH4 production. The outcomes from this research work recommended applying lysozyme 3%, protease 3%, and lime 5000 ppm for the inhibition of CH₄ production.

6.2 Recommendations

Based on the results, some recommendations for future work are summarized as follows:

225

1. Impact of lime treatment on tailings dewatering and cap water quality under an oil sands end pit lake scenario:

- Future work is warranted to optimize the lime treatment and the occurrence of cation exchange in the EPL scenario.
- Monitoring EPL systems and the levels of contamination in cap water layer associated with the movement of COPCs will help in understanding the fate and transport of such contaminants and will support establishing an inclusive standard that reflect on the influenced environment characteristics, intended water usage, and reclaim substitutes.
- Oil Sands Industrial companies and researchers are advised to report the mine closure plan results, commissioned EPLs and EPL piloting's studies (Syncrude BML) of their monitoring programs for EPLs to verify the EPL cap layer water chemistry and the movement of COPCs mechanism to well derive a controlled condition understanding for EPL approaches.
 - 2. Enzyme-assisted dewatering of oil sands tailings: Significance of water chemistry and biological activity:
- Further investigation is still warranted to investigate the release of organic fractions from FFT and their associated potential risks.
- Future studies that consider different oil sands tailings with different characteristics and higher/lower solid content, organics load such as NAFCs and PHCs would evaluate the capability of the enzyme dosage and demonstrate and quantify the individual contribution of microbial community species with

regards to the enzyme treatment and biodegradation process to the overall dewatering.

3. Significance of enzyme and lime treatment in the mitigation of methane emissions from oil sands tailings:

- Future studies that include different tailings sources and age with higher loads of PHCs would evaluate and quantify the capability of enzyme treatment in association with the archaeal abundance, and the individual contribution of the various species to the inhibition of GHGs and to the overall treatment.
- In comparison to single enzyme, a combination of two or multi enzyme assisted process may offer a distinctive collaborative advantage to inhibit the GHGs which should be further examined.
- It could be beneficial to further investigate ways to further enrich the enzymes' role and the enzymes performance in an EPL system/scenario and evaluate the impact of enzyme treatment on improving the inhibition mechanism in EPL.

BIBLIOGRAPHY

- AER. (2019). State of Fluid Tailings Management for Mineable Oil Sands, 2018.Alberta Energy Regulator: Calgary, AB, Canada.
- Alberta Go. 2019. Electronic resource about oil sands. [www document], url https://www.Alberta.Ca/about-oil-sands.Aspx. Bioresource Technology Reports. 11:100547.
- Allam, N. E., Romaniuk, N., Tate, M., Meshref, M. N. A., Dhar, B. R., & Ulrich,
 A. C. (2021). Impact of lime treatment on tailings dewatering and cap water
 quality under an oil sands end pit lake scenario. Science of The Total
 Environment, 781, 146699.
 https://doi.org/https://doi.org/10.1016/j.scitotenv.2021.146699
- Allam NE, Anwar MN, Kuznetsov PV, Ulrich AC, Dhar BR. 2022. Enzymeassisted dewatering of oil sands tailings: Significance of water chemistry and biological activity. Chemical Engineering Journal.135162.
- Allen, E.W. 2008a. Process water treatment in Canada's oil sands industry: II. A review of emerging technologies. Journal of Environmental Engineering and Science 7(5), 499-524.
- Allen, E.W. 2008b. Process water treatment in Canada's oil sands industry: I. Target pollutants and treatment objectives. Journal of Environmental Engineering and Science 7(2), 123-138.
- Allen, J.P., Atekwana, E.A., Atekwana, E.A., Duris, J.W., Werkema, D.D. and Rossbach, S. 2007. The microbial community structure in petroleum-
contaminated sediments corresponds to geophysical signatures. Appl. Environ. Microbiol. 73(9), 2860-2870.

- Amoako KA. 2021. Geotechnical behaviour of two novel polymer treatments of oil sands fine tailings.
- Anderson J, Wiseman SB, Moustafa A, El-Din MG, Liber K, Giesy JP. 2012. Effects of exposure to oil sands process-affected water from experimental reclamation ponds on chironomus dilutus. water research. 46(6):1662-1672.
- Anderson, J.C., Wiseman, S.B., Wang, N., Moustafa, A., Perez-Estrada, L., Gamal El-Din, M., Martin, J.W., Liber, K. and Giesy, J.P. 2011. Effectiveness of Ozonation Treatment in Eliminating Toxicity of Oil Sands Process-Affected Water to Chironomus dilutus. Environ. Sci. Technol. 46(1), 486-493.
- Anderson, K., Sallis, P., & Uyanik, S. (2003). Anaerobic treatment processes. V:The Handbook of Water and Wastewater Microbiology. Mara D., Horan N.In: London, Academic Press.
- Ayol, A., & Dentel, S. K. (2005). Enzymatic treatment effects on dewaterability of anaerobically digested biosolids-II: laboratory characterizations of drainability and filterability. Process Biochemistry, 40(7), 2435-2442.
- Baillie, R.A. and Malmberg, E.W. 1969 Removal of clay from the water streams of the hot water process by flocculation, Google Patents.
- Baker LF, Ciborowski JJH, MacKinnon MD. 2012. Petroleum coke and soft tailings sediment in constructed wetlands may contribute to the uptake of

trace metals by algae and aquatic invertebrates. Science of the Total Environment. 414:177-186.

- Balaberda, A. L., & Ulrich, A. C. (2021). Persulfate Oxidation Coupled with Biodegradation by Pseudomonas fluorescens Enhances Naphthenic Acid Remediation and Toxicity Reduction. Microorganisms, 9(7). https://doi.org/10.3390/microorganisms9071502
- Barceló, D., Žonja, B., & Ginebreda, A. (2020). Toxicity tests in wastewater and drinking water treatment processes: A complementary assessment tool to be on your radar. Journal of Environmental Chemical Engineering, 8(5), 104262.
- Bartlett, A. J., Frank, R. A., Gillis, P. L., Parrott, J. L., Marentette, J. R., Brown, L.
 R., . . . Brunswick, P. (2017). Toxicity of naphthenic acids to invertebrates:
 Extracts from oil sands process-affected water versus commercial mixtures.
 Environmental pollution, 227, 271-279.
- Beaty, N.B. and Lane, M.D. 1983. The polymerization of acetyl-CoA carboxylase. Journal of Biological Chemistry 258(21), 13051-13055.
- Beier N, Sego D. 2008. The oil sands tailings research facility. Geotechnical news. 26(2):72-77.
- Beltrán-Heredia J, Sánchez-Martín J. 2009. Removal of sodium lauryl sulphate by coagulation/flocculation with moringa oleifera seed extract. Journal of hazardous materials. 164(2-3):713-719.

- Benitez EI, Genovese DB, Lozano JE. 2007. Effect of ph and ionic strength on apple juice turbidity: Application of the extended dlvo theory. Food Hydrocolloid. 21(1):100-109.
- Bennett, D.D., Higgins, S.E., Moore, R.W., Beltran, R., Caldwell, D.J., Byrd, J.A. and Hargis, B.M. 2003. Effects of lime on Salmonella enteritidis survival in vitro. Journal of applied poultry research 12(1), 65-68.
- Biswas AA, Lee SS, Mamuad LL, Kim S-H, Choi Y-J, Bae G-S, Lee K, Sung H-G, Lee S-S. 2016. Use of lysozyme as a feed additive on in vitro rumen fermentation and methane emission. Asian-Australasian journal of animal sciences. 29(11):1601-1607.
- Board, E.R.C. 2012 Energy Resources Conservation Board Tailings Management Assessment Report: Oil Sands Mining Industry, (posted date: June 12, 2013), from Alberta Energy Regulator, Accessed onlineApril2017:http://osipfiles.alberta.ca/datasets/158/TailingsManageme ntAssessmentReport2011-2012.pdf.
- Bolyen E, Rideout JR, Dillon MR, Bokulich NA, Abnet CC, Al-Ghalith GA, Alexander H, Alm EJ, Arumugam M, Asnicar F. 2019. Reproducible, interactive, scalable and extensible microbiome data science using qiime 2. Nature biotechnology. 37(8):852-857.
- Bonilla, S. (2017). Protein-Based Conditioners for Enhancing Biosludge Dewaterability.

- Bonilla, S., Tran, H., & Allen, D. G. (2015). Enhancing pulp and paper mill biosludge dewaterability using enzymes. Water research, 68, 692-700.
- Bordenave, S., Kostenko, V., Dutkoski, M., Grigoryan, A., Martinuzzi, R. J., & Voordouw, G. (2010). Relation between the activity of anaerobic microbial populations in oil sands tailings ponds and the sedimentation of tailings. Chemosphere, 81(5), 663-668.
- Bowman, D.T., Arriaga, D., Morris, P., Risacher, F., Warren, L.A., McCarry, B.E. and Slater, G. 2016. Chemical fingerprinting of naphthenic acids at an oil sands end pit lake by comprehensive two-dimensional gas chromatography/time-of-flight mass spectrometry (GC× GC/TOFMS). AGUFM 2016, H23C-1563.
- Bradford, L.M., Ziolkowski, L.A., Goad, C., Warren, L.A. and Slater, G.F. 2017. Elucidating carbon sources driving microbial metabolism during oil sands reclamation. Journal of environmental management 188, 246-254.
- Brickner, H. 2013. The mobility of petroleum hydrocarbons in Athabasca oil sands tailings.
- Brandon JT. 2016. Turbidity mitigation in an oil sands end pit lake through ph reduction and fresh water addition.
- Brown, L.D., Pérez-Estrada, L., Wang, N., El-Din, M.G., Martin, J.W., Fedorak, P.M. and Ulrich, A.C. 2013. Indigenous microbes survive in situ ozonation improving biodegradation of dissolved organic matter in aged oil sands process-affected waters. Chemosphere 93(11), 2748-2755.

- Brown LD, Ulrich AC. 2015. Oil sands naphthenic acids: A review of properties, measurement, and treatment. Chemosphere. 127:276-290.
- Burns, H. and Gremminger, L. 1994 Lime and fly ash stabilization of wastewater treatment sludge, Google Patents.
- Cárdenas-Manríquez M, Young RF, Semple KM, Li C, Coy D, Underwood E, Siddique T, Guigard S, Bressler DC, Gupta R et al. 2020. Microbiallymediated de-watering and consolidation ("biodensification") of oil sands mature fine tailings, amended with agri-business by-products. Nova Scientia. 12(24).
- Caughill, D.L., Morgenstern, N.R. and Scott, J.D. 1993. Geotechnics of nonsegregating oil sand tailings. Canadian Geotechnical Journal 30(5), 801-811.
- CCME 2008. CANADA-WIDE STANDARD FOR PETROLEUM HYDROCARBONS (PHC) IN SOIL.
- Cervini-Silva, J., Ruíz, G., Hernández, J. M., Palacios, E., Balmaseda, J., Orta, M., & Hernández, S. (2020). Coupled reduction of bicarbonate to methane and generation/sequestration of carbon dioxide in natural basaltic-and claymatrices at 25° C≤ T≤ 100° C. Applied Clay Science, 198, 105844.
- Chalaturnyk, R.J., Don Scott, J. and Özüm, B. 2002. MANAGEMENT OF OIL SANDS TAILINGS. Petroleum Science and Technology 20(9-10), 1025-1046.

- Chan ASK, Parkin TB. 2000. Evaluation of potential inhibitors of methanogenesis and methane oxidation in a landfill cover soil. Soil Biology and Biochemistry. 32(11):1581-1590.
- Chen L, Guangxia WU, Huanzhen MU, Zonghuan Y, Lianyi T, Xiangwei LIN. 2008. Synthesis and application of lignin-based copolymer lsaa on controlling non-point source pollution resulted from surface runoff. Journal of Environmental Sciences. 20(7):820-826.
- Chen M, Walshe G, Chi Fru E, Ciborowski JJH, Weisener CG. 2013a. Microcosm assessment of the biogeochemical development of sulfur and oxygen in oil sands fluid fine tailings. Applied Geochemistry. 37:1-11.
- Chen M, Walshe G, Fru EC, Ciborowski JJH, Weisener CG. 2013b. Microcosm assessment of the biogeochemical development of sulfur and oxygen in oil sands fluid fine tailings. Applied geochemistry. 37:1-11.
- Chen Z, Zhang W, Wang D, Ma T, Bai R. 2015. Enhancement of activated sludge dewatering performance by combined composite enzymatic lysis and chemical re-flocculation with inorganic coagulants: Kinetics of enzymatic reaction and re-flocculation morphology. Water Research. 83:367-376.
- Choi, J.-M., Han, S.-K., & Lee, C.-Y. (2018). Enhancement of methane production in anaerobic digestion of sewage sludge by thermal hydrolysis pretreatment. Bioresource technology, 259, 207-213.
- Choudhary M, Ray MB, Neogi S. 2019. Evaluation of the potential application of cactus (opuntia ficus-indica) as a bio-coagulant for pre-treatment of oil

sands process-affected water. Separation and Purification Technology. 209:714-724.

- Clark KA, Pasternack DS. 1932. Hot water seperation of bitumen from alberta bituminous sand. Industrial & Engineering Chemistry. 24(12):1410-1416.
- Clemente, J.S. and Fedorak, P.M. 2005. A review of the occurrence, analyses, toxicity, and biodegradation of naphthenic acids. Chemosphere 60(5), 585-600.
- COSIA 2012 Technical Guide for Fluid Fine Tailings Management pp. Report, 1-131., Canadian Oil Sands Innovation Alliance, Edmonton, AB. August2012, Edmonton, Alberta, Canada.
- Cossey, H. L., Anwar, M. N., Kuznetsov, P. V., & Ulrich, A. C. (2021). Biofilms for Turbidity Mitigation in Oil Sands End Pit Lakes. Microorganisms, 9(7), 1443. https://doi.org/10.3390/microorganisms9071443
- Cossey HL, Batycky AE, Kaminsky H, Ulrich AC. 2021. Geochemical stability of oil sands tailings in mine closure landforms. Minerals. 11(8).
- Crittenden JC, Trussell RR. 2005. Hand, dw; howe, kj: Tchobanoglous, g. Water treatment principles and design john wiley & sons. Inc. Design concepts for thin layered freeze-thaw dewatering systems. 1993.
- Danilova, O. V., & Dedysh, S. N. (2014). Abundance and diversity of methanotrophic Gammaproteobacteria in northern wetlands. Microbiology, 83(1), 67-76.

- Dawson RF, Sego DC, Pollock GW. 1999. Freeze-thaw dewatering of oil sands fine tails. Canadian geotechnical journal. 36(4):587-598.
- Dean, E.W. and Stark, D.D. 1920. A Convenient Method for the Determination of Water in Petroleum and Other Organic Emulsions. Industrial & Engineering Chemistry 12(5), 486-490.
- Derjaguin, B., Landau, L., 1993. Theory of the stability of strongly charged lyophobic sols and of the adhesion of strongly charged-particles in solutions of electro-lytes. Prog. Surf. Sci. 43 (1-4), 30-59.
- Demeter, M. A., Lemire, J., Yue, G., Ceri, H., & Turner, R. J. (2015). Culturing oil sands microbes as mixed species communities enhances ex situ model naphthenic acid degradation. Frontiers in microbiology, 6, 936.
- Devenny, D. W. (2010). A screening study of oil sand tailings technologies and practices. Prepared for Alberta Energy Research Institute.
- Dompierre, K.A. and Barbour, S.L. 2016. Characterization of physical mass transport through oil sands fluid fine tailings in an end pit lake: a multi-tracer study. Journal of Contaminant Hydrology 189, 12-26.
- Dompierre, K.A., Lindsay, M.B.J., Cruz-Hernández, P. and Halferdahl, G.M. 2016a. Initial geochemical characteristics of fluid fine tailings in an oil sands end pit lake. Science of The Total Environment 556(Supplement C), 196-206.

- Dompierre, K.A., Lindsay, M.B.J., Cruz-Hernández, P. and Halferdahl, G.M. 2016b. Initial geochemical characteristics of fluid fine tailings in an oil sands end pit lake. Science of The Total Environment 556, 196-206.
- Dompierre, K.A.H. 2016. Controls on mass and thermal loading to an oil sands end pit lake from underlying fluid fine tailings.
- Dompierre KA, Barbour SL. 2016b. Thermal properties of oil sands fluid fine tailings: Laboratory and in situ testing methods. Canadian Geotechnical Journal. 54(3):428-440.
- Dursun, D., Turkmen, M., Abu-Orf, M., & Dentel, S. K. (2006). Enhanced sludge conditioning by enzyme pre-treatment: comparison of laboratory and pilot scale dewatering results. Water science and technology, 54(5), 33-41.
- Ehimen, E. A., Holm-Nielsen, J. B., Poulsen, M., & Boelsmand, J. E. (2013). Influence of different pre-treatment routes on the anaerobic digestion of a filamentous algae. Renewable energy, 50, 476-480.
- El-Din MG, Fu H, Wang N, Chelme-Ayala P, Pérez-Estrada L, Drzewicz P, Martin JW, Zubot W, Smith DW. 2011. Naphthenic acids speciation and removal during petroleum-coke adsorption and ozonation of oil sands processaffected water. Science of the Total Environment. 409(23):5119-5125.
- ERCB. (2012). Alberta's Energy Reserves 2011 and Supply/Demand Outlook 2012-2021. Government of Alberta, C., AB, Canada, 2012 (ed).

- Ewin, G.J., Erno, B.P. and Hepler, L.G. 1981. Clay chemistry: Investigation of thermodynamics of ion exchange reactions by titration calorimetry. Canadian Journal of Chemistry 59(20), 2927-2933.
- Famakinwa, T.E., Su, Y., Wang, J. and Gates, I.D. 2018. An in-situ process to consolidate oil sands mine tailings. Journal of Environmental Chemical Engineering 6(2), 3295-3305.
- Farkish A. 2013. Sap based rapid dewatering of oil sands mature fine tailings. Ma sc thesis. Citeseer.
- Farkish A, Fall M. 2013. Rapid dewatering of oil sand mature fine tailings using super absorbent polymer (sap). Minerals Engineering. 50-51(Supplement C):38-47.
- Farkish A, Fall M. 2014. Consolidation and hydraulic conductivity of oil sand mature fine tailings dewatered by using super absorbent polymer. Journal of Geotechnical and Geoenvironmental Engineering. 140(7):06014006.
- Fedorak PM, Coy DL, Dudas MJ, Simpson MJ, Renneberg AJ, MacKinnon MD. 2003. Microbially-mediated fugitive gas production from oil sands tailings and increased tailings densification rates. Journal of Environmental Engineering and Science. 2(3):199-211.
- Ferris, A.P. and Jepson, W.B. 1975. The exchange capacities of kaolinite and the preparation of homoionic clays. Journal of Colloid and Interface Science 51(2), 245-259.

- Ficker M, Krastel K, Orlicky S, Edwards E. 1999. Molecular characterization of a toluene-degrading methanogenic consortium. Applied and environmental microbiology. 65(12):5576-5585.
- Fine, P., Graber, E.R. and Yaron, B. 1997. Soil interactions with petroleum hydrocarbons: abiotic processes. Soil Technology 10(2), 133-153.
- Foght JM, Fedorak PM, Westlake DWS, Boerger HJ. 1985. Microbial content and metabolic activities in the syncrude tailings pond. AOSTRA J Res. 1:139-146.
- Foght, J. M., Gieg, L. M., & Siddique, T. (2017). The microbiology of oil sands tailings: past, present, future [Review]. Fems Microbiology Ecology, 93(5), 22, Article fix034. https://doi.org/10.1093/femsec/fix034
- Foght, J. M., Siddique, T., & Gieg, L. M. (2014). Protocols for Handling, Storing, and Cultivating Oil Sands Tailings Ponds Materials for Microbial and Molecular Biological Study. In Hydrocarbon and Lipid Microbiology Protocols (pp. 175-197). Springer.
- Folwell, B. D., McGenity, T. J., & Whitby, C. (2020). Diamondoids are not forever: microbial biotransformation of diamondoid carboxylic acids. Microbial biotechnology, 13(2), 495-508.
- Fowler, S. J., Toth, C. R. A., & Gieg, L. M. (2016). Community structure in methanogenic enrichments provides insight into syntrophic interactions in hydrocarbon-impacted environments. Frontiers in microbiology, 7, 562.

- Frank RA, Kavanagh R, Burnison BK, Arsenault G, Headley JV, Peru KM, Van Der Kraak G, Solomon KR. 2008. Toxicity assessment of collected fractions from an extracted naphthenic acid mixture. Chemosphere. 72(9):1309-1314.
- Gold, N. D., & Martin, V. J. J. (2007). Global view of the Clostridium thermocellum cellulosome revealed by quantitative proteomic analysis. Journal of bacteriology, 189(19), 6787-6795.
- Georgiou D, Liliopoulos V, Aivasidis A. 2019. Investigation of an integrated treatment technique for anaerobically digested animal manure: Lime reaction and settling, ammonia stripping and neutralization by biogas scrubbing. Bioresource Technology Reports. 5:127-133.
- Gosselin P, Hrudey SE, Naeth MA, Plourde A, Therrien R, Van Der Kraak G, XuZ. 2010. Environmental and health impacts of canada's oil sands industry.Royal Society of Canada, Ottawa, ON. 10.
- Government of Alberta. 2015. Lower athabasca region : Tailings management framework for the mineable athabasca oil sands.
- Government of Canada. 2018. Technical backgrounder: Federal methane regulations for the upstream oil and gas sector - canada.Ca, https://www. Canada.Ca/en/environment-climate-change/news/2018/04/federalmethaneregulations-for-the-upstream-oil-and-gas-sector.Html, (accessed jan 28, 2022).

- Government of Alberta. 2019. Electronic resource about oil sands. [www document], url https://www.Alberta.Ca/about-oil-sands.Aspx. Bioresource Technology Reports. 11:100547.
- Grasso D, Subramaniam K, Butkus M, Strevett K, Bergendahl J. 2002. A review of non-dlvo interactions in environmental colloidal systems. Reviews in Environmental Science and Biotechnology. 1(1):17-38.
- Grawert T, Hohmann H-P, Kindermann M, Duval S, Bacher A, Fischer M. 2014.
 Inhibition of methyl-com reductase from methanobrevibacter ruminantium
 by 2-bromoethanesulfonate. Journal of agricultural and food chemistry.
 62(52):12487-12490.
- Grewer, D. M., Young, R. F., Whittal, R. M., & Fedorak, P. M. (2010). Naphthenic acids and other acid-extractables in water samples from Alberta: what is being measured? Science of the Total Environment, 408(23), 5997-6010.
- Gulley JR. 1992. Oil sands tailings sludge abandonment dry land reclamation study.
- Gumfekar, S. P., Rooney, T. R., Hutchinson, R. A., & Soares, J. B. P. (2017).
 Dewatering Oil Sands Tailings with Degradable Polymer Flocculants. ACS
 Applied Materials & Interfaces, 9(41), 36290-36300.
 https://doi.org/10.1021/acsami.7b10302
- Gumfekar SP, Vajihinejad V, Soares JBP. 2019. Advanced polymer flocculants for solid–liquid separation in oil sands tailings. Macromolecular Rapid Communications. 40(1):1800644.

- Guo C. 2009. Rapid densification of the oil sands mature fine tailings (mft) by microbial activity.
- Guo, J., & Xu, Y. (2011). Review of enzymatic sludge hydrolysis. Journal of Bioremediation and Biodegradation, 2(5).
- Hamza, H.A., Stanonik, D.J. and Kessick, M.A. 1996. Flocculation of lime-treated oil sands tailings. Fuel 75(3), 280-284.
- Hargesheimer, E. E., Coutts, R. T., & MacKinnon, M. D. (1984). Characterization of simple phenols in oil sands extraction-process water. Environmental Technology, 5(1-11), 433-440.
- Hardy WB. 1900. A preliminary investigation of the conditions which determine the stability of irreversible hydrosols. Proceedings of the Royal Society of London. 66(424-433):110-125.
- Haveroen ME, MacKinnon MD, Fedorak PM. 2005. Polyacrylamide added as a nitrogen source stimulates methanogenesis in consortia from various wastewaters. Water Research. 39(14):3333-3341.
- Headley, J. V., Barrow, M. P., Peru, K. M., Fahlman, B., Frank, R. A., Bickerton, G., . . . Hewitt, L. M. (2011). Preliminary fingerprinting of Athabasca oil sands polar organics in environmental samples using electrospray ionization Fourier transform ion cyclotron resonance mass spectrometry. Rapid Communications in Mass Spectrometry, 25(13), 1899-1909. https://doi.org/10.1002/rcm.5062

- Headley, J. V., & McMartin, D. W. (2004). A review of the occurrence and fate of naphthenic acids in aquatic environments. Journal of Environmental Science and Health, Part A, 39(8), 1989-2010.
- Headley, J. V., Peru, K. M., & Barrow, M. P. (2009). Mass spectrometric characterization of naphthenic acids in environmental samples: a review. Mass spectrometry reviews, 28(1), 121-134.
- Henderson G, Cook GM, Ronimus RS. 2016. Enzyme-and gene-based approaches for developing methanogen-specific compounds to control ruminant methane emissions: A review. Animal Production Science. 58(6):1017-1026.
- Herman DC, Fedorak PM, MacKinnon MD, Costerton JW. 1994. Biodegradation of naphthenic acids by microbial populations indigenous to oil sands tailings. Canadian journal of microbiology. 40(6):467-477.
- Holowenko, F. M., MacKinnon, M. D., & Fedorak, P. M. (2000). Methanogens and sulfate-reducing bacteria in oil sands fine tailings waste. Canadian Journal of Microbiology, 46(10), 927-937.
- Holowenko FM, MacKinnon MD, Fedorak PM. 2002. Characterization of naphthenic acids in oil sands wastewaters by gas chromatography-mass spectrometry. Water research. 36(11):2843-2855.
- Hrynyshyn, J. 2012. End pit lakes guidance document 2012. Cumulative Environmental Management Association, Fort McMurray, Alberta. CEMA Contract (2010-0016).

- Huc AY. 2011. Heavy crude oils: From geology to upgrading (introduction and synopsis. Editions Technip, April.27-34.
- Hughes, S. A., Mahaffey, A., Shore, B., Baker, J., Kilgour, B., Brown, C., . . .
 Bailey, H. C. (2017). Using ultrahigh-resolution mass spectrometry and toxicity identification techniques to characterize the toxicity of oil sands process-affected water: The case for classical naphthenic acids. Environmental toxicology and chemistry, 36(11), 3148-3157.
- Humphries, M., & Library of Congress. Congressional Research Service. (2008).North American Oil Sands history of development, prospects for the future (Updated January 17, 2008 ed.). Congressional Research Service.
- Hunter RJ. 2001. Foundations of colloid science. Oxford ; New York: Oxford University Press.
- Hyndman, A. and Sobkowicz, J. 2010, Oil sands tailings: reclamation goals and the state of technology, pp. 12-16.
- Immig I, Demeyer D, Fiedler D, Van Nevel C, Mbanzamihigo L. 1996. Attempts to induce reductive acetogenesis into a sheep rumen. Archives of Animal Nutrition. 49(4):363-370.
- Jaenicke, S., Ander, C., Bekel, T., Bisdorf, R., Dröge, M., Gartemann, K.-H., ... Tille, F. (2011). Comparative and joint analysis of two metagenomic datasets from a biogas fermenter obtained by 454-pyrosequencing. PloS one, 6(1), e14519.

- Jeeravipoolvarn S. 2010. Geotechnical behavior of in-line thickened oil sands tailings. University of Alberta Edmonton.
- Jeeravipoolvarn S, Scott JD, Chalaturnyk RJ. 2009. 10 m standpipe tests on oil sands tailings: Long-term experimental results and prediction. Canadian Geotechnical Journal. 46(8):875-888.
- Ji Y, Lu Q, Liu Q, Zeng H. 2013. Effect of solution salinity on settling of mineral tailings by polymer flocculants. Colloids and Surfaces A: Physicochemical and Engineering Aspects. 430:29-38.
- Johnson RL, Bork P, Allen EA, James WH, Koverny L. 1993. Oil sands sludge dewatering by freeze-thaw and evapotranspiration.
- Johnson, R. J., West, C. E., Swaih, A. M., Folwell, B. D., Smith, B. E., Rowland, S. J., & Whitby, C. (2012). Aerobic biotransformation of alkyl branched aromatic alkanoic naphthenic acids via two different pathways by a new isolate of Mycobacterium. Environmental microbiology, 14(4), 872-882.
- Kabwe, L.K., Scott, J.D., Beier, N.A., Wilson, G.W. and Jeeravipoolvarn, S. 2018.Environmental Implications of End Pit Lakes at Oil Sands Mine.Environmental Geotechnics, 1-25.
- Kaminsky, H. A. W., Etsell, T. H., Ivey, D. G., & Omotoso, O. (2009). Distribution of Clay Minerals in the Process Streams Produced by the Extraction of Bitumen from Athabasca Oil Sands. Canadian Journal of Chemical Engineering, 87(1), 85-93. https://doi.org/10.1002/cjce.20133

- Kannel PR, Gan TY. 2012. Naphthenic acids degradation and toxicity mitigation in tailings wastewater systems and aquatic environments: A review. Journal of Environmental Science and Health, Part A. 47(1):1-21.
- Kasperski, K. L., & Mikula, R. J. (2011). Waste streams of mined oil sands: characteristics and remediation. Elements, 7(6), 387-392.
- Kavanagh RJ, Frank RA, Oakes KD, Servos MR, Young RF, Fedorak PM, MacKinnon MD, Solomon KR, Dixon DG, Van Der Kraak G. 2011.
 Fathead minnow (pimephales promelas) reproduction is impaired in aged oil sands process-affected waters. Aquatic Toxicology. 101(1):214-220.
- Kim, S., Woo, S.-G., Lee, J., Lee, D.-H., & Hwang, S. (2019). Evaluation of feasibility of using the bacteriophage t4 lysozyme to improve the hydrolysis and biochemical methane potential of secondary sludge. Energies, 12(19), 3644.
- Kong JD, Wang H, Siddique T, Foght J, Semple K, Burkus Z, Lewis MA. 2019. Second-generation stoichiometric mathematical model to predict methane emissions from oil sands tailings. Science of The Total Environment. 694:133645.
- Kuznetsova A, Kuznetsov P, Young RF, Semple KM, Li C, Foght JM, Siddique T. 2021. Microbial transformation of solid phase impacts quality of recovered water during consolidation of bioreactor-treated oil sands tailings. Journal of Environmental Chemical Engineering. 9(1):104715.

- Lawrence, G.A., Tedford, E.W. and Pieters, R. 2015. Suspended solids in an end pit lake: potential mixing mechanisms. Canadian Journal of Civil Engineering 43(3), 211-217.
- Leahy JG, Colwell RR. 1990. Microbial degradation of hydrocarbons in the environment. Microbiological reviews. 54(3):305-315.
- Leishman C, Widdup EE, Quesnel DM, Chua G, Gieg LM, Samuel MA, Muench DG. 2013. The effect of oil sands process-affected water and naphthenic acids on the germination and development of arabidopsis. Chemosphere. 93(2):380-387.
- Li, C. (2010) Methanogenesis in oil sands tailings. [electronic resource] : an analysis of the microbial community involved and its effects on tailings densification, 2010.
- Li C, Fu L, Stafford J, Belosevic M, El-Din MG. 2017. The toxicity of oil sands process-affected water (ospw): A critical review. Science of the Total Environment. 601:1785-1802.
- Li, Q., Jonas, U., Zhao, X.S., Kapp, M., 2008. The forces at work in collodial self-assembly: a review on fundamental interactions between collodial particels.Asia Pac. J. Chem. Eng. 3 (3), 255-268.
- Liang, B., Wang, L.-Y., Mbadinga, S. M., Liu, J.-F., Yang, S.-Z., Gu, J.-D., & Mu, B.-Z. (2015). Anaerolineaceae and Methanosaeta turned to be the dominant microorganisms in alkanes-dependent methanogenic culture after long-term of incubation. Amb Express, 5(1), 1-13.

- Liang Y, Zheng Z, Hua R, Luo X. 2011. A preliminary study of simultaneous lime treatment and dry digestion of smooth cordgrass for biogas production. Chemical Engineering Journal. 174(1):175-181.
- Liggio J, Li S-M, Staebler RM, Hayden K, Darlington A, Mittermeier RL, O'Brien J, McLaren R, Wolde M, Worthy D. 2019. Measured canadian oil sands co2 emissions are higher than estimates made using internationally recommended methods. Nature communications. 10(1):1-9.
- Lin, F., Zhu, X., Luo, Y., Yu, P., & Liu, M. (2019). Improvement of activated sludge dewatering properties using green conditioners: chitosan hydrochloride and lysozyme. RSC advances, 9(12), 6936-6945.
- Lin, L., Chowdhury, B., Zakaria, B. S., & Dhar, B. R. (2020). Temperaturedependent (20–55° C) electrocatalytic characteristics during ethanol/propionate degradation by methanogenic communities grown on conductive carbon fibers. Chemical Engineering Journal, 391, 123566.
- Liu G, Wang K, Li X, Ma L, Ma X, Chen H. 2019. Enhancement of excess sludge hydrolysis and decomposition with different lysozyme dosage. Journal of Hazardous Materials. 366:395-401.
- Liu LY, Hu Y, Min FF, Zhang MX, Song SX. 2013. Characterizations and stability of colloidal coal-measure kaolinite in aqueous suspensions: A review. Surf Rev Lett. 20(1).
- Liu, W., Zhong, X., Cheng, L., Wang, J., Sun, Y., Deng, Y., & Zhang, Z. (2018). Cellular and compositional insight into the sludge dewatering process using

enzyme treatment. Environmental Science and Pollution Research, 25(29), 28942-28953.

- Long J, Li H, Xu Z, Masliyah JH. 2006. Role of colloidal interactions in oil sand tailings treatment. AIChE journal. 52(1):371-383.
- Lorentz, J., Bara, B., Mikula, R., Leneve, E. and Wang, N. 2014 Oil sands fluid fine tailings dewatering using rim ditching, Google Patents.
- Lu J, Rao S, Le T, Mora S, Banerjee S. 2011. Increasing cake solids of cellulosic sludge through enzyme-assisted dewatering. Process biochemistry. 46(1):353-357.
- Lu Q. 2016. Understanding two-step polymer flocculation of oil sands tailings. [2016].
- Lu QY, Yan B, Xie L, Huang J, Liu Y, Zeng HB. 2016. A two-step flocculation process on oil sands tailings treatment using oppositely charged polymer flocculants. Science of the Total Environment. 565:369-375.
- Ma, C. and Eggleton, R.A. 1999. Cation exchange capacity of kaolinite. Clays and Clay minerals 47(2), 174-180.
- MacKinnon MD, Boerger H. 1986. Description of two treatment methods for detoxifying oil sands tailings pond water. Water Quality Research Journal. 21(4):496-512.
- MacKinnon, M.D. 1989. Development of the tailings pond at Syncrude's oil sands plant: 1978–1987. AOSTRA J. Res 5(2), 109-133.

- MacKinnon MD, Matthews JG, Shaw WH, Cuddy RG. 2000. Water quality issues associated with implementation of composite tailings (ct) technology for managing oil sands tailings.
- MacKinnon, M.D., Matthews, J.G., Shaw, W.H. and Cuddy, R.G. 2001. Water quality issues associated with composite tailings (CT) technology for managing oil sands tailings. International Journal of Surface Mining, Reclamation and Environment 15(4), 235-256.
- MacKinnon M. 2004. Oil sands water quality issues: Properties, treatment and discharge options.
- Mackinnon M, Zubot W. 2013. Chapter 10: Water management. Kingsley Knowledge Publishing: Alberta.
- Madill REA, Orzechowski MT, Chen G, Brownlee BG, Bunce NJ. 2001.
 Preliminary risk assessment of the wet landscape option for reclamation of oil sands mine tailings: Bioassays with mature fine tailings pore water.
 Environmental Toxicology: An International Journal. 16(3):197-208.
- Magdalena JA, Llamas M, Tomás-Pejó E, González-Fernández C. 2019. Semicontinuous anaerobic digestion of protease pretreated chlorella biomass for volatile fatty acids production. Journal of Chemical Technology & Biotechnology. 94(6):1861-1869.
- Magdalena JA, Tomás-Pejó E, Ballesteros M, González-Fernandez C. 2018. Volatile fatty acids production from protease pretreated chlorella biomass via anaerobic digestion. Biotechnology Progress. 34(6):1363-1369.

- Mahdavi, H. (2014). The Application Of Algae For The Removal Of Metals And Naphthenic Acids From Oil Sands Tailings Pond Water.
- Mahdavi, H., Prasad, V., Liu, Y. and Ulrich, A.C. 2015. In situ biodegradation of naphthenic acids in oil sands tailings pond water using indigenous algae– bacteria consortium. Bioresource technology 187, 97-105.
- Mahdavi H, Ulrich AC, Liu Y. 2012. Metal removal from oil sands tailings pond water by indigenous micro-alga. Chemosphere. 89(3):350-354.
- Mahdy A, Mendez L, Ballesteros M, González-Fernández C. 2015. Protease pretreated chlorella vulgaris biomass bioconversion to methane via semicontinuous anaerobic digestion. Fuel. 158:35-41.
- Mamer M. 2010. Oil sands tailings technology: Understanding the impact to reclamation.
- Masliyah JH, Bhattacharjee S. 2006. Electrokinetic and colloid transport phenomena. Hoboken, N.J.: Wiley-Interscience.
- Masliyah, J.H., Xu, Z. and Czarnecki, J.A. (2011) Handbook on theory and practice of bitumen recovery from Athabasca oil sands, Kingsley Knowledge Pub.
- Matsuda, S., & Ohtsuki, T. (2016). Establishment and analysis of microbial communities capable of producing methane from grass waste at extremely high C/N ratio. Int J New Technol Res, 2, 81-86.
- Matthews, J.G., Shaw, W.H., MacKinnon, M.D. and Cuddy, R.G. 2002. Development of composite tailings technology at Syncrude. International Journal of Surface Mining, Reclamation and Environment 16(1), 24-39.

- McQueen, A.D., Kinley, C.M., Hendrikse, M., Gaspari, D.P., Calomeni, A.J., Iwinski, K.J., Castle, J.W., Haakensen, M.C., Peru, K.M. and Headley, J.V. 2017. A risk-based approach for identifying constituents of concern in oil sands process-affected water from the Athabasca Oil Sands region. Chemosphere 173, 340-350.
- Meshref, M. N. A., Chelme-Ayala, P., & Gamal El-Din, M. (2017). Fate and abundance of classical and heteroatomic naphthenic acid species after advanced oxidation processes: Insights and indicators of transformation and degradation. Water Research, 125, 62-71. https://doi.org/https://doi.org/10.1016/j.watres.2017.08.007
- Meshref, M. N. A., Klamerth, N., Islam, M. S., McPhedran, K. N., & Gamal El-Din, M. (2017). Understanding the similarities and differences between ozone and peroxone in the degradation of naphthenic acids: Comparative performance for potential treatment. Chemosphere, 180, 149-159. https://doi.org/10.1016/j.chemosphere.2017.03.113
- Miles, S. M., Asiedu, E., Balaberda, A.-l., & Ulrich, A. C. (2020). Oil sands process affected water sourced Trichoderma harzianum demonstrates capacity for mycoremediation of naphthenic acid fraction compounds. Chemosphere, 258, 127281.
- Miles, S.M., Hofstetter, S., Edwards, T., Dlusskaya, E., Cologgi, D.L., Gänzle, M. and Ulrich, A.C. 2019. Tolerance and cytotoxicity of naphthenic acids on microorganisms isolated from oil sands process-affected water. Science of The Total Environment 695, 133749.

- Mishra, P. N., Scheuermann, A., Bore, T., & Li, L. (2019). Salinity effects on soil shrinkage characteristic curves of fine-grained geomaterials. Journal of Rock Mechanics and Geotechnical Engineering, 11(1), 181-191.
- Misiti, T., Tezel, U., & Pavlostathis, S. G. (2013). Fate and effect of naphthenic acids on oil refinery activated sludge wastewater treatment systems. Water Research, 47(1), 449-460. https://doi.org/https://doi.org/10.1016/j.watres.2012.10.036
- Missana T, Adell A. 2000. On the applicability of dlvo theory to the prediction of clay colloids stability. J Colloid Interf Sci. 230(1):150-156.
- Mohamed MH, Wilson LD, Headley JV, Peru KM. 2011. Sequestration of naphthenic acids from aqueous solution using β-cyclodextrin-based polyurethanes. Physical Chemistry Chemical Physics. 13(3):1112-1122.
- Morandi GD, Wiseman SB, Guan M, Zhang XW, Martin JW, Giesy JP. 2017. Elucidating mechanisms of toxic action of dissolved organic chemicals in oil sands process-affected water (ospw). Chemosphere. 186:893-900.
- Morandi, G. D., Wiseman, S. B., Pereira, A., Mankidy, R., Gault, I. G. M., Martin, J. W., & Giesy, J. P. (2015). Effects-directed analysis of dissolved organic compounds in oil sands process-affected water. Environmental science & technology, 49(20), 12395-12404.
- Navarro SS, Cimpoia R, Bruant G, Guiot SR. 2014. Specific inhibitors for identifying pathways for methane production from carbon monoxide by a

nonadapted anaerobic mixed culture. Canadian Journal of Microbiology. 60(6):407-415.

- Nelson LR, MacKinnon M, Gulley JR. 1993. Application of toxicity testing in the evaluation of reclamation options for oil sands fine tails.
- Nix PG. 1983. Detoxification of tailings pond top water. Suncor Incorporated.
- Nix PG, Martin RW. 1992. Detoxification and reclamation of suncor's oil sand tailings ponds. Environmental Toxicology and Water Quality. 7(2):171-188.
- Nyberg, K.A., Vinnerås, B., Lewerin, S.S., Kjellberg, E. and Albihn, A. 2011. Treatment with Ca(OH)2 for inactivation of Salmonella Typhimurium and Enterococcus faecalis in soil contaminated with infected horse manure. Journal of Applied Microbiology 110(6), 1515-1523.
- Odnell, A., Recktenwald, M., Stensén, K., Jonsson, B.-H., & Karlsson, M. (2016). Activity, life time and effect of hydrolytic enzymes for enhanced biogas production from sludge anaerobic digestion. Water Research, 103, 462-471. https://doi.org/https://doi.org/10.1016/j.watres.2016.07.064
- Op den Camp, H. J. M., Islam, T., Stott, M. B., Harhangi, H. R., Hynes, A., Schouten, S., . . . Dunfield, P. F. (2009). Environmental, genomic and taxonomic perspectives on methanotrophic Verrucomicrobia. Environmental Microbiology Reports, 1(5), 293-306.
- Oren, A. (2014). The family Methanoregulaceae. The Prokaryotes: Other Major Lineages of Bacteria and the Archaea, 253-258.

- O'Shea J-P, Qiao GG, Franks GV. 2011. Temperature responsive flocculation and solid–liquid separations with charged random copolymers of poly(n-isopropyl acrylamide). Journal of Colloid and Interface Science. 360(1):61-70.
- Owolagba, J., & Azam, S. (2015). Effect of seasonal weather variations on the desiccation behavior of treated oil sand fine tailings. Environmental Earth Sciences, 74(2), 1711-1717.
- Pandey, J., Chauhan, A. and Jain, R.K. 2009. Integrative approaches for assessing the ecological sustainability of in situ bioremediation. FEMS Microbiology Reviews 33(2), 324-375.
- Parawira, W. (2012). Enzyme research and applications in biotechnological intensification of biogas production. Critical Reviews in Biotechnology, 32(2), 172-186. https://doi.org/10.3109/07388551.2011.595384
- Park T, Mao H, Yu Z. 2019. Inhibition of rumen protozoa by specific inhibitors of lysozyme and peptidases in vitro. Frontiers in microbiology. 10:2822.
- Parmar, N., Singh, A., & Ward, O. P. (2001). Enzyme treatment to reduce solids and improve settling of sewage sludge. Journal of industrial microbiology and biotechnology, 26(6), 383-386.
- Passos, F., Hom-Diaz, A., Blanquez, P., Vicent, T., & Ferrer, I. (2016). Improving biogas production from microalgae by enzymatic pretreatment. Bioresource Technology, 199, 347-351. https://doi.org/https://doi.org/10.1016/j.biortech.2015.08.084

- Peng, H., Sun, J., Alharbi, H. A., Jones, P. D., Giesy, J. P., & Wiseman, S. (2016). Peroxisome proliferator-activated receptor γ is a sensitive target for oil sands process-affected water: effects on adipogenesis and identification of ligands. Environmental science & technology, 50(14), 7816-7824.
- Penner, T. J., & Foght, J. M. (2010). Mature fine tailings from oil sands processing harbour diverse methanogenic communities. Canadian journal of microbiology, 56(6), 459-470.
- Pereira, A. S., Bhattacharjee, S., & Martin, J. W. (2013). Characterization of oil sands process-affected waters by liquid chromatography orbitrap mass spectrometry. Environmental science & technology, 47(10), 5504-5513.
- Pereira, A. S., Islam, M. D. S., Gamal El-Din, M., & Martin, J. W. (2013). Ozonation degrades all detectable organic compound classes in oil sands process-affected water; an application of high-performance liquid chromatography/obitrap mass spectrometry. Rapid Communications in Mass Spectrometry, 27(21), 2317-2326.
- Poon, H.Y. 2019 An Examination of the Effect of Diluent on Microbial Dynamics in Oil Sands Tailings and the Mechanistic Insight on Carbon Dioxidemediated Turbidity Reduction in Oil Sands Surface Water.
- Poon, H.Y., Brandon, J.T., Yu, X. and Ulrich, A.C. 2018. Turbidity Mitigation in an Oil Sands Pit Lake through pH Reduction and Fresh Water Addition. Journal of Environmental Engineering 144(12), 04018127.

- Pourrezaei P. 2013. Physico-chemical processes for oil sands process-affected water treatment.
- Pourrezaei P, Drzewicz P, Wang Y, Gamal El-Din M, Perez-Estrada LA, Martin JW, Anderson J, Wiseman S, Liber K, Giesy JP. 2011. The impact of metallic coagulants on the removal of organic compounds from oil sands process-affected water. Environmental Science & Technology. 45(19):8452-8459.
- Pourrezaei, P., Alpatova, A., Chelme-Ayala, P., Perez-Estrada, L. A., Jensen-Fontaine, M., Le, X. C., & El-Din, M. G. (2014). Impact of petroleum coke characteristics on the adsorption of the organic fractions from oil sands process-affected water. International Journal of Environmental Science and Technology, 11(7), 2037-2050.
- Preez ACd, S Preston J. 1992. The solvent extraction of rare-earth metals by carboxylic acids. Solvent Extraction and Ion Exchange. 10(2):207-230.
- Proskin, S., Sego, D. and Alostaz, M. 2012. Oil sands MFT properties and freezethaw effects. Journal of Cold Regions Engineering 26(2), 29-54.
- Proskin S, Sego D, Alostaz M. 2010. Freeze-thaw and consolidation tests on suncor mature fine tailings (mft). Cold Regions Science and Technology. 63(3):110-120.
- Ramos-Padrón, E., Bordenave, S., Lin, S., Bhaskar, I. M., Dong, X., Sensen, C. W.,
 . . . Gieg, L. M. (2011). Carbon and Sulfur Cycling by Microbial
 Communities in a Gypsum-Treated Oil Sands Tailings Pond.

Environmental Science & Technology, 45(2), 439-446. https://doi.org/10.1021/es1028487

- Ramos-Padrón E, Bordenave S, Lin S, Bhaskar IM, Dong X, Sensen CW, Fournier J, Voordouw G, Gieg LM. 2010. Carbon and sulfur cycling by microbial communities in a gypsum-treated oil sands tailings pond. Environmental science & technology. 45(2):439-446.
- Ramos-Padrón E, Bordenave S, Lin S, Bhaskar IM, Dong X, Sensen CW, Fournier J, Voordouw G, Gieg LM. 2011. Carbon and sulfur cycling by microbial communities in a gypsum-treated oil sands tailings pond. Environmental Science & Technology. 45(2):439-446.
- Redfield E, Croser C, Zwiazek JJ, MacKinnon MD, Qualizza C. 2003. Responses of red-osier dogwood to oil sands tailings treated with gypsum or alum. Journal of environmental quality. 32(3):1008-1014.
- Reid ML, Warren LA. 2016. S reactivity of an oil sands composite tailings deposit undergoing reclamation wetland construction. Journal of environmental management. 166:321-329.
- Richardson, E., Bass, D., Smirnova, A., Paoli, L., Dunfield, P. and Dacks, J.B.
 2020. Phylogenetic estimation of community composition and novel eukaryotic lineages in Base Mine Lake: an oil sands tailings reclamation site in Northern Alberta. Journal of Eukaryotic Microbiology 67(1), 86-99.
- Risacher, F.F., Morris, P.K., Arriaga, D., Goad, C., Nelson, T.C., Slater, G.F. and Warren, L.A. 2018. The interplay of methane and ammonia as key oxygen

consuming constituents in early stage development of Base Mine Lake, the first demonstration oil sands pit lake. Applied Geochemistry 93, 49-59.

- Robert, C., Chassard, C., Lawson, P. A., & Bernalier-Donadille, A. (2007). Bacteroides cellulosilyticus sp. nov., a cellulolytic bacterium from the human gut microbial community. International journal of systematic and evolutionary microbiology, 57(7), 1516-1520.
- Rogers VV, Liber K, MacKinnon MD. 2002. Isolation and characterization of naphthenic acids from athabasca oil sands tailings pond water. Chemosphere. 48(5):519-527.
- Roman, H. J., Burgess, J. E., & Pletschke, B. I. (2006). Enzyme treatment to decrease solids and improve digestion of primary sewage sludge. African Journal of Biotechnology, 5(10).
- Romaniuk N, Mehranfar M, Selani H, Ozum B, Tate MJ, Scott JD. 2015. Residual bitumen recovery from oil sand tailings with lime. Proceedings tailings and mine waste 2015, vancouver, bc, october 26 to 28, 2015.
- Ross, M. S., Pereira, A. d. S., Fennell, J., Davies, M., Johnson, J., Sliva, L., & Martin, J. W. (2012). Quantitative and Qualitative Analysis of Naphthenic Acids in Natural Waters Surrounding the Canadian Oil Sands Industry. Environmental Science & Technology, 46(23), 12796-12805. https://doi.org/10.1021/es303432u

- Sabyasachi M, Venkatesh M, Ajay K D, Dena W M, John V H, Kerry M P. 2010. Photocatalysis of naphthenic acids in water. Journal of Water Resource and Protection. 2010.
- Saidi-Mehrabad, A., He, Z., Tamas, I., Sharp, C. E., Brady, A. L., Rochman, F. F.,... Dong, X. (2013). Methanotrophic bacteria in oilsands tailings ponds of northern Alberta. The ISME journal, 7(5), 908-921.
- Samadi, N. 2019. Partitioning of inorganic contaminants between fluid fine tailings and cap water under end pit lake scenario: Biological, Chemical and Mineralogical processes.
- Sawyer CN, McCarty PL, Parkin GF, Sawyer CN. 1994. Chemistry for environmental engineering. New York: McGraw-Hill.
- Schofield RK, Samson HR. 1954. Flocculation of kaolinite due to the attraction of oppositely charged crystal faces. Discussions of the Faraday Society. 18:135-145.
- Schulze H. 1882. Schwefelarsen in wässriger lösung. Journal für praktische Chemie. 25(1):431-452.
- Schoof, A. (2015). Enzymatic Treatment of Phenolic Industrial Wastewater With Nitrogen Management.
- Scott, A. C., Young, R. F., & Fedorak, P. M. (2008). Comparison of GC-MS and FTIR methods for quantifying naphthenic acids in water samples. Chemosphere, 73(8), 1258-1264. https://doi.org/10.1016/j.chemosphere.2008.07.024

- Scott AC, Mackinnon MD, Fedorak PM. 2005. Naphthenic acids in athabasca oil sands tailings waters are less biodegradable than commercial naphthenic acids. Environmental science & technology. 39(21):8388-8394.
- Scott, A.C., Zubot, W., MacKinnon, M.D., Smith, D.W. and Fedorak, P.M. 2008.Ozonation of oil sands process water removes naphthenic acids and toxicity.Chemosphere 71(1), 156-160.
- Seyedy Niasar H. 2017. Treatment of oil sands process-affected water using activated and surface modified petroleum coke for organic compounds recovery.
- Shahimin MFM, Foght JM, Siddique T. 2016. Preferential methanogenic biodegradation of short-chain n-alkanes by microbial communities from two different oil sands tailings ponds. Science of the Total Environment. 553:250-257.
- Shahimin MFM, Siddique T. 2017. Methanogenic biodegradation of paraffinic solvent hydrocarbons in two different oil sands tailings. Science of the Total Environment. 583:115-122.
- Shen W, Li Z, Liu Y. 2008. Surface chemical functional groups modification of porous carbon. Recent Patents on Chemical Engineering. 1(1):27-40.
- Siddique T, Fedorak PM, Foght JM. 2006. Biodegradation of short-chain n-alkanes in oil sands tailings under methanogenic conditions. Environmental science & technology. 40(17):5459-5464.

- Siddique T, Fedorak PM, MacKinnon MD, Foght JM. 2007. Metabolism of btex and naphtha compounds to methane in oil sands tailings. Environmental science & technology. 41(7):2350-2356.
- Siddique T, Gupta R, Fedorak PM, MacKinnon MD, Foght JM. 2008. A first approximation kinetic model to predict methane generation from an oil sands tailings settling basin. Chemosphere. 72(10):1573-1580.
- Siddique, T., Penner, T., Semple, K., & Foght, J. M. (2011). Anaerobic
 Biodegradation of Longer-Chain n-Alkanes Coupled to Methane
 Production in Oil Sands Tailings. Environmental Science & Technology, 45(13), 5892-5899. https://doi.org/10.1021/es200649t
- Siddique T, Penner T, Klassen J, Nesbø C, Foght JM. 2012. Microbial communities involved in methane production from hydrocarbons in oil sands tailings. Environmental science & technology. 46(17):9802-9810.
- Siddique, T., Kuznetsov, P., Kuznetsova, A., Arkell, N., Young, R., Li, C., Guigard, S.E., Underwood, E. and Foght, J.M. 2014. Microbially-accelerated consolidation of oil sands tailings. Pathway I: changes in porewater chemistry. Frontiers in microbiology 5, 106.
- Siddique T, Kuznetsov P, Kuznetsova A, Li C, Young R, Arocena JM, Foght J.2014c. Microbially-accelerated consolidation of oil sands tailings. Pathwayii: Solid phase biogeochemistry. Frontiers in Microbiology. 5:107.
- Siddique T, Stasik S, Mohamad Shahimin MF, Wendt-Potthoff K. 2018. Microbial communities in oil sands tailings: Their implications in biogeochemical

processes and tailings management. Microbial communities utilizing hydrocarbons and lipids: members, metagenomics and ecophysiology Handbook of hydrocarbon and lipid microbiology Springer Nature, Basel, Switzerland.1-33.

- Siddique T, Kuznetsova A. 2020. Linking hydrocarbon biodegradation to greenhouse gas emissions from oil sands tailings and its impact on tailings management. Canadian Journal of Soil Science.1-9.
- Siddique T, Semple K, Li C, Foght JM. 2020. Methanogenic biodegradation of isoalkanes and cycloalkanes during long-term incubation with oil sands tailings. Environmental Pollution. 258:113768.
- Siddique T, Mohamad Shahimin MF, Zamir S, Semple K, Li C, Foght JM. 2015. Long-term incubation reveals methanogenic biodegradation of c5 and c6 iso-alkanes in oil sands tailings. Environmental science & technology. 49(24):14732-14739.
- Simpson IJ, Blake NJ, Barletta B, Diskin GS, Fuelberg HE, Gorham K, Huey LG, Meinardi S, Rowland FS, Vay SA. 2010. Characterization of trace gases measured over alberta oil sands mining operations: 76 speciated c 2–c 10 volatile organic compounds (vocs), co 2, ch 4, co, no, no 2, no y, o 3 and so 2. Atmospheric Chemistry and Physics. 10(23):11931-11954.
- Small CC, Cho S, Hashisho Z, Ulrich AC. 2015. Emissions from oil sands tailings ponds: Review of tailings pond parameters and emission estimates. Journal of Petroleum Science and Engineering. 127:490-501.

- Small CC, Ulrich AC, Hashisho Z. 2012. Adsorption of acid extractable oil sands tailings organics onto raw and activated oil sands coke. Journal of Environmental Engineering. 138(8):833-840.
- Smith, K. S., & Ingram-Smith, C. (2007). Methanosaeta, the forgotten methanogen? Trends in microbiology, 15(4), 150-155.
- Sobolewski A. 1997. Microbial distributions in process-affected aquatic ecosystems. Consultants report submitted to Alberta Oil Sand Technology Research Authority Calgary, Alberta.
- Staples TL, Chatterjee PK. 2002. Synthetic superabsorbents. Textile science and technology. Elsevier. p. 283-322.
- Stasik S, Loick N, Knöller K, Weisener C, Wendt-Potthoff K. 2014. Understanding biogeochemical gradients of sulfur, iron and carbon in an oil sands tailings pond. Chemical Geology. 382:44-53.
- Stasik S, Wendt-Potthoff K. 2014. Interaction of microbial sulphate reduction and methanogenesis in oil sands tailings ponds. Chemosphere. 103:59-66.
- Steffen F, Janzon R, Saake B. 2018. Enzymatic treatment of deinking sludge–effect on fibre and drainage properties. Environmental technology. 39(21):2810-2821.
- Strosher MT, Peake E. 1978. Characterization of organic constituents in waters and wastewaters of the athabasca oil sands mining area.
- Sutton NB, Maphosa F, Morillo JA, Abu Al-Soud W, Langenhoff AAM, Grotenhuis T, Rijnaarts HHM, Smidt H. 2013. Impact of long-term diesel
contamination on soil microbial community structure. Applied and environmental microbiology. 79(2):619-630.

- Sworska A, Laskowski JS, Cymerman G. 2000. Flocculation of the syncrude fine tailings: Part i. Effect of ph, polymer dosage and mg2+ and ca2+ cations. International journal of mineral processing. 60(2):143-152.
- Tate, M., Leikam, J., Fox, J. and Romaniuk, N. 2017 Use of calcium hydroxide as a coagulant to improve oil sands tailings treatment. Proceedings Tailings and Mine Waste 2017, Banff, Alberta, Canada, November 2017.
- Tate, M., Leikam, J., ScotT, J.D., Mehranfar, M., Romaniuk, N. and Ozum, B. 2016 Impacts of Calcium Compounds on Oil Sands Water Chemistry. Conference presentation at the Mining and Tailings Waste Conference, keystone, Colorado, USA.
- Tate M, Leikam J, Fox J, Romaniuk N. 2019. Treatment of oil sands tailings with lime at elevated ph levels. Google Patents.
- Thomas L, Jungschaffer G, Sprössler B. 1993. Improved sludge dewatering by enzymatic treatment. Water Science and Technology. 28(1):189-192.
- Thomas, L., Joseph, A., & Gottumukkala, L. D. (2014). Xylanase and cellulase systems of Clostridium sp.: an insight on molecular approaches for strain improvement. Bioresource technology, 158, 343-350.
- Toth CRA, Gieg LM. 2018. Time course-dependent methanogenic crude oil biodegradation: Dynamics of fumarate addition metabolites, biodegradative

genes, and microbial community composition. Frontiers in Microbiology. 8.

- Tukanghan, W., Hupfauf, S., Gómez-Brandón, M., Insam, H., Salvenmoser, W., Prasertsan, P., . . . Sompong, O. (2021). Symbiotic Bacteroides and Clostridium-rich methanogenic consortium enhanced biogas production of high-solid anaerobic digestion systems. Bioresource Technology Reports, 14, 100685.
- Vajihinejad, V., Guillermo, R., & Soares, J. B. P. (2017). Dewatering Oil Sands Mature Fine Tailings (MFTs) with Poly(acrylamide-codiallyldimethylammonium chloride): Effect of Average Molecular Weight and Copolymer Composition. Industrial & Engineering Chemistry Research, 56(5), 1256-1266. https://doi.org/10.1021/acs.iecr.6b04348
- Vajihinejad, V. and Soares, J.B.P. 2018. Monitoring polymer flocculation in oil sands tailings: A population balance model approach. Chemical Engineering Journal 346, 447-457.
- Vajihinejad V, Gumfekar SP, Dixon DV, Silva MA, Soares JBP. 2021. Enhanced dewatering of oil sands tailings by a novel water-soluble cationic polymer. Separation and Purification Technology. 260:118183.
- Vanwonterghem, I., Jensen, P. D., Dennis, P. G., Hugenholtz, P., Rabaey, K., & Tyson, G. W. (2014). Deterministic processes guide long-term synchronised population dynamics in replicate anaerobic digesters. The ISME journal, 8(10), 2015-2028.

- Vedoy DRL, Soares JBP. 2015. Water-soluble polymers for oil sands tailing treatment: A review. The Canadian Journal of Chemical Engineering. 93(5):888-904.
- Wagner CD, Passoja DE, Hillery HF, Kinisky TG, Six HA, Jansen WT, Taylor JA. 1982. Auger and photoelectron line energy relationships in aluminum– oxygen and silicon–oxygen compounds. Journal of Vacuum Science and Technology. 21(4):933-944.
- Wang, C., Alpatova, A., McPhedran, K.N. and Gamal El-Din, M. 2015. Coagulation/flocculation process with polyaluminum chloride for the remediation of oil sands process-affected water: Performance and mechanism study. Journal of Environmental Management 160(Supplement C), 254-262.
- Wang, C., Harbottle, D., Liu, Q. and Xu, Z. 2014. Current state of fine mineral tailings treatment: A critical review on theory and practice. Minerals Engineering 58, 113-131.
- Wang, S., Hou, X., & Su, H. (2017). Exploration of the relationship between biogas production and microbial community under high salinity conditions. Scientific reports, 7(1), 1-10.
- Wang W, Wang K, Zhao Q, Liu Y. 2022. Simultaneous degradation of anodic sludge and cathodic refractory pollutant in a mfc powered ef system enhanced by co-addition of lysozyme and 2-bromoethane sulfonate. Journal of Environmental Chemical Engineering. 10(3):108074.

- Wells, P.S. and Riley, D.A. 2007, MFT drying—case study in the use of rheological modification and dewatering of fine tailings through thin lift deposition in the oil sands of Alberta, pp. 13-15.
- Westcott F, Watson L. 2005. End pit lakes technical guidance document. Clearwater environmental consultants for CEMA end Pit Lakes subgroup, Project. 61.
- White J, Stewart CA, Wareham DG, Wilson TM. 2011. Treatment of volcanic ashcontaminated surface waters through the optimisation of physical and chemical processes. GNS Science.
- White, K.B. 2017. Characterizing annual changes in the chemistry and toxicity of surface water from Base Mine Lake, an Alberta oil sands end pit lake.
- White, K.B. and Liber, K. 2018. Early chemical and toxicological risk characterization of inorganic constituents in surface water from the Canadian oil sands first large-scale end pit lake. Chemosphere 211, 745-757.
- White, K.B. and Liber, K. 2020. Chronic Toxicity of Surface Water from a Canadian Oil Sands End Pit Lake to the Freshwater Invertebrates Chironomus dilutus and Ceriodaphnia dubia. Archives of Environmental Contamination and Toxicology 78(3), 439-450.
- Whitman, W. B., Bowen, T. L., & Boone, D. R. (2006). The methanogenic bacteria. Prokaryotes, 3(Chapter 9), 165-207.

- Wilkins, D., Lu, X.-Y., Shen, Z., Chen, J., & Lee, P. K. H. (2015). Pyrosequencing of mcrA and archaeal 16S rRNA genes reveals diversity and substrate preferences of methanogen communities in anaerobic digesters. Applied and environmental microbiology, 81(2), 604-613.
- Wilson, S. L., Li, C., Ramos-Padrón, E., Nesbø, C., Soh, J., Sensen, C. W., . . . Gieg, L. M. (2016). Oil sands tailings ponds harbour a small core prokaryotic microbiome and diverse accessory communities. Journal of Biotechnology, 235, 187-196.
- Wong, J.W.C., Fang, M. and Jiang, R. 2001. Persistency of bacterial indicators in biosolids stabilization with coal fly ash and lime. Water environment research 73(5), 607-611.
- Wong, J.W.C. and Selvam, A. 2009. Reduction of indicator and pathogenic microorganisms in pig manure through fly ash and lime addition during alkaline stabilization. Journal of Hazardous Materials 169(1-3), 882-889.
- Wu B, Chai X, Zhao Y. 2016. Enhanced dewatering of waste-activated sludge by composite hydrolysis enzymes. Bioprocess and biosystems engineering. 39(4):627-639.
- Wu, C., De Visscher, A., & Gates, I. D. (2019). On naphthenic acids removal from crude oil and oil sands process-affected water. Fuel, 253, 1229-1246.
- Wu, C., Zhang, K., Chen, J., & Li, S. (2014). Effect of Cellulase and Protease Pretreatment on Dewaterability of Waste Activated Sludge from Paper Mill. BioResources, 9(3), 5198-5207.

- Wu S, Wang B, Sego D, Naeth A. 2011. Responses of wheatgrass species to composite/consolidated oil sands tailings.
- Xin X-D, He J-G, Qiu W, Tang J, Liu T-T. 2015. Microbial community related to lysozyme digestion process for boosting waste activated sludge biodegradability. Bioresource technology. 175:112-119.
- Yamamoto K, Tamaki H, Cadillo-Quiroz H, Imachi H, Kyrpides N, Woyke T, Goodwin L, Zinder SH, Kamagata Y, Liu W-T. 2014. Complete genome sequence of methanoregula formicica smspt, a mesophilic hydrogenotrophic methanogen isolated from a methanogenic upflow anaerobic sludge blanket reactor. Genome announcements. 2(5):e00870-00814.
- Yang, G.-C., Zhou, L., Mbadinga, S. M., Liu, J.-F., Yang, S.-Z., Gu, J.-D., & Mu,
 B.-Z. (2016). Formate-dependent microbial conversion of CO2 and the dominant pathways of methanogenesis in production water of high-temperature oil reservoirs amended with bicarbonate. Frontiers in microbiology, 7, 365.
- Yang H, Fan M, Liu A, Dong L. 2015. General formulas for drag coefficient and settling velocity of sphere based on theoretical law. International Journal of Mining Science and Technology. 25(2):219-223.
- You Y, Staebler RM, Moussa SG, Beck J, Mittermeier RL. 2021. Methane emissions from an oil sands tailings pond: A quantitative comparison of

fluxes derived by different methods. Atmospheric Measurement Techniques. 14(3):1879-1892.

- Young RF, Li C, Semple KM, Baker MG, Kuznetsov P, Kuznetsova A, Siddique T, Foght JM. 2020. Accelerated consolidation of oil sands tailings using an anaerobic bioreactor. Bioresource Technology Reports. 11:100547.
- Yu, X., Cao, Y., Sampaga, R., Rybiak, S., Burns, T. and Ulrich, A.C. 2018a. Accelerated dewatering and detoxification of oil sands tailings using a biological amendment. Journal of Environmental Engineering 144(9), 04018091.
- Yu, X., Lee, K., Ma, B., Asiedu, E. and Ulrich, A.C. 2018b. Indigenous microorganisms residing in oil sands tailings biodegrade residual bitumen. Chemosphere 209, 551-559.
- Yu, X., Lee, K. and Ulrich, A.C. 2019. Model naphthenic acids removal by microalgae and Base Mine Lake cap water microbial inoculum. Chemosphere 234, 796-805.
- Yuan X. 2009. Thickened tailings (paste) technology and its applicability in oilsand tailings management. Tailings and Mine Waste 2009, Nov 1-4, Banf, Alberta.
- Yuan, X. S., & Shaw, W. (2007). Novel Processes for Treatment of Syncrude Fine Transition and Marine Ore Tailings. Canadian Metallurgical Quarterly, 46(3), 265-272. https://doi.org/10.1179/cmq.2007.46.3.265

- Zakaria BS, Dhar BR. 2019. Progress towards catalyzing electro-methanogenesis in anaerobic digestion process: Fundamentals, process optimization, design and scale-up considerations. Bioresource technology. 289:121738.
- Zakaria, B. S., & Dhar, B. R. (2021a). An intermittent power supply scheme to minimize electrical energy input in a microbial electrolysis cell assisted anaerobic digester. Bioresource Technology, 319, 124109.
- Zakaria, B. S., & Dhar, B. R. (2021b). Characterization and significance of extracellular polymeric substances, reactive oxygen species, and extracellular electron transfer in methanogenic biocathode. Scientific reports, 11(1), 1-13.
- Zhang, L., Zhang, Y., Patterson, J., Arslan, M., Zhang, Y., & El-Din, M. G. (2020). Biofiltration of oil sands process water in fixed-bed biofilm reactors shapes microbial community structure for enhanced degradation of naphthenic acids. Science of the Total Environment, 718, 137028.
- Zhu R, Liu QX, Xu ZH, Masliyah JH, Khan A. 2011. Role of dissolving carbon dioxide in densification of oil sands tailings. Energy & Fuels. 25(5):2049-2057.
- Zhu, S., Li, M. Y., & El -Din, M. G. (2017). Forward osmosis as an approach to manage oil sands tailings water and on-site basal depressurization water [Article]. Journal of Hazardous Materials, 327, 18-27. https://doi.org/10.1016/j.jhazmat.2016.12.025

- Zinder SH, Anguish T, Cardwell SC. 1984. Selective inhibition by 2bromoethanesulfonate of methanogenesis from acetate in a thermophilic anaerobic digestor. Applied and environmental microbiology. 47(6):1343-1345.
- Zubot, W. 2010. Water use and aquatic reclamation at Syncrude Canada Ltd. Integrated oil sands mining operation. The Association of Professional Engineers, Geologists, and Geophysicists of Alberta (APEGA).
- Zubot W, An Z, Benally C, El-Din MG. 2021. Treatment of oil sands process water using petroleum coke: Field pilot. Journal of Environmental Management. 289:112407.
- Zubot, W., MacKinnon, M. D., Chelme-Ayala, P., Smith, D. W., & El-Din, M. G. (2012). Petroleum coke adsorption as a water management option for oil sands process-affected water. Sci. Total Environ., 427, 364-372. https://doi.org/10.1016/j.scitotenv.2012.04.024
- Zverlov, V. V., Kellermann, J., & Schwarz, W. H. (2005). Functional subgenomics of Clostridium thermocellum cellulosomal genes: identification of the major catalytic components in the extracellular complex and detection of three new enzymes. Proteomics, 5(14), 3646-3653.

APPENDIX A. STANDARD CURVES



Figure A1. Methane determination calibration curve using standards.



Figure A2. Toluene determination calibration curve using standards.



Figure A3. Hexane determination calibration curve using standards.

APPENDIX B. SUPPLEMENTARY INFORMATION

WEEK		0	1	3	5	7	9	11	13	15	17	19	21	23
PROTEASE 1%	P1	0.8	1.1	1.1	1.1	1.2	1.1	1.1	1.0	1.0	1.1	1.1	1.1	1.4
	P2	0.8	1.5	1.5	1.5	1.4	1.5	1.3	1.3	1.3	1.5	1.2	0.0	1.5
	P3	0.9	1.2	1.4	1.4	1.3	1.2	1.2	1.2	1.1	1.4	1.0	0.0	1.0
PROTEASE 1.5%	P4	1.2	1.9	2.1	2.1	2.0	2.0	1.8	1.7	1.8	2.0	1.8	1.7	1.5
	Р5	1.2	1.7	2.0	2.1	2.0	2.0	1.8	1.6	1.6	2.0	1.6	0.0	1.3
	P6	1.2	1.3	1.4	1.6	1.5	1.6	1.4	1.4	1.3	1.4	1.2	0.0	1.3
PROTEASE 3%	P7	1.9	2.2	1.8	2.1	2.0	1.8	1.5	1.4	1.4	1.6	1.3	1.5	0.9
	P8	2.2	1.6	1.8	2.0	1.8	1.8	1.6	0.0	1.5	1.6	1.5	0.0	1.0

Table B1. Methane production/ concentration in (mmol) throughout the experiment for all treatment conditions, control/untreated FFT

 and negative control 2-Bromoethanesulfonate (BES).

WEEK		0	1	3	5	7	9	11	13	15	17	19	21	23
	Р9	2.5	2.4	2.2	2.5	1.9	2.2	1.4	0.1	1.3	1.9	1.7	0.0	2.0
LYSOZYME 1.5%	LZ1	1.6	2.1	2.1	2.1	2.3	2.6	2.5	3.0	2.4	3.6	2.7	2.9	2.2
	LZ2	1.6	1.8	1.8	2.2	2.1	2.1	2.1	2.6	2.5	4.1	3.6	0.0	3.3
	LZ3	1.5	1.7	1.9	2.0	1.7	2.4	1.9	2.7	2.1	3.8	4.0	0.0	2.0
LYSOZYME 3%	LZ4	1.5	1.5	1.5	1.4	1.3	1.4	1.4	1.3	1.4	1.4	0.1	1.2	0.9
	LZ5	1.5	1.6	1.3	1.0	1.3	1.5	1.5	1.4	1.3	1.5	1.3	0.0	1.1
	LZ6	1.6	1.1	1.8	1.0	1.4	1.6	1.5	1.4	1.4	1.7	1.4	1.4	1.2
LIME 1600 PPM	LM1	1.2	1.3	1.6	1.5	1.3	1.4	1.3	1.3	1.1	1.4	1.1	1.1	1.2
	LM2	1.2	1.2	1.3	1.3	1.1	1.3	1.1	11.1	1.2	1.4	1.1	1.1	1.0

WEEK			0	1	3	5	7	9	11	13	15	17	19	21	23
		LM3	1.2	0.9	0.8	0.8	0.8	0.9	0.8	0.9	0.8	0.3	0.2	0.3	0.8
LIME 3500 PPM	LM4	1.5	1.4	1.6	1.6	1.5	1.5	1.3	1.4	1.2	1.5	1.4	1.3	1.3	
	LM5	1.3	1.3	1.2	1.2	1.1	1.1	0.9	1.0	0.1	1.0	0.8	0.8	0.7	
	LM6	1.1	1.1	1.3	1.0	1.0	0.9	0.8	1.0	0.1	1.1	1.2	0.8	0.7	
LIME 5000 PPM	LM7	1.7	1.6	2.2	1.9	1.9	1.7	1.5	1.5	0.1	1.8	0.1	0.0	0.1	
	LM8	2.0	2.0	2.1	2.2	2.1	1.7	1.5	1.8	0.1	1.9	1.6	0.0	1.4	
	LM9	1.8	1.8	1.5	1.8	1.8	1.5	1.7	1.5	0.1	1.8	1.5	1.3	1.3	
BES 50 UMOL/L	BES 1	0.6	0.7	0.0	0.0	0.0	0.0	0.1	0.0	0.0	0.1	0.1	0.0	0.7	
	BES 2	0.5	0.6	1.1	1.0	1.0	0.9	0.9	0.9	0.9	1.1	0.9	0.9	0.8	

WEEK		0	1	3	5	7	9	11	13	15	17	19	21	23
	BES 3	0.7	0.6	0.8	0.8	0.8	0.7	0.7	0.7	0.6	0.8	0.6	0.0	0.6
CONTROL	Control 1	1.4	2.0	1.9	1.9	2.4	2.5	2.6	2.6	2.7	2.7	2.8	2.7	2.1
	Control 2	1.5	1.4	2.0	2.1	2.1	2.6	2.6	2.7	2.7	2.6	2.7	2.7	1.8
	Control 3	1.5	2.1	1.8	2.0	2.4	2.5	2.7	2.7	2.7	2.8	2.7	2.6	1.8



Figure B1. Pressure filter (OFITE; Bench-Mount Filter Press with Hose and Regulator; #140-31) used at Chapter 3 to increase the solids content of raw FFT to 50% (wt./wt.) at 100 psi.



Figure B2. The experimental set-up for Chapter 3 consists of 500 ml lime treated or untreated FFT (50% solids) covered with 500 ml Raw synthetic OSPW. All columns were open to the atmosphere.



Figure B3. The experimental set-up for Chapter 4, (a) 1L column consists of 800 ml treated or untreated FFT and sealed with parafilm and aluminum foil. (b) 158 mL microcosms consist of 75 mL of FFT and 25 mL of anaerobic media.



Figure B4. The experimental set-up for Chapter 5, (a) 158 mL microcosms consists of 65 mL FFT and 35 mL of anaerobic media control, treated with enzyme, lime or BES, (b) All the microcosms incubated at 30 °C incubator in dark conditions.